Ficus deltoidea and its associated fig wasps in

Peninsular Malaysia

Siti Khairiyah Binti Mohd Hatta

Submitted in accordance with the requirements for the degree of Doctor of Philosophy

> The University of Leeds School of Biology

> > May 2019

The candidate confirms that the work submitted is her own and that appropriate credit has been given where reference has been made to the work of others.

This copy has been supplied on the understanding that it is copyright material and that no quotation from the thesis may be published without proper acknowledgement.

© 2019 The University of Leeds and Siti Khairiyah Binti Mohd Hatta

ACKNOWLEDGEMENTS

بِسْ إِلَا وَالْبَحْمِ الْحَيْمُ

First and foremost, I would like to express my sincere gratitude to my supervisor, Dr Stephen Compton for his continuous support of my PhD study, his endless patience, motivation, earnest guidance and immense knowledge. I could not have imagined having a better advisor and mentor for my PhD study. I am also deeply indebted to my co-supervisor, Dr Rupert Quinnell for his knowledgeable help with statistical analysis. My sincere thanks also goes to my two other co-supervisors from Malaysia, Prof Idris Abd Ghani and Prof Nashriyah Mat who provided me with the opportunity to do this research in their laboratories, provided material for some experiments and provided other research facilities. Without they precious support it would not have been possible to conduct this research.

My special thanks to Noor Nasuha Abd Aziz for assisting me throughout my PhD journey especially by accompanying me to the fields to collect the data. To all plantations owners, Puan Siti, Encik Ismail, Encik Ng, thank you for giving me permission to conduct the research in your plantations. Furthermore, I would like to take this opportunity to say my thanks to technical staff in Universiti Kebangsaan Malaysia and Universiti Sultan Zainal Abidin for their kindness and technical support. A very special gratitude goes out to all at the Ministry of Education Malaysia and also Universiti Teknologi MARA for helping and providing the funding for the work.

The completion of this study would have been not possible and was dependent on the steadfast support and encouragement of my husband, Mohd Fadli Nazri and daughters (Khaireen Amni, Nisreen Alya and Naureen Ayra). They made an equal contribution to the study for which I will always feel profound gratitude in my heart. I am grateful to my mother, mother in law and father in law who helped me taking care of my two babies who were born along my PhD journey. I am also grateful to my other family members and friends who have supported me along the way. Thank you very, very much.

ABSTRACT

The interaction of fig trees (Ficus) and their fig wasp pollinators (Agaonidae) is an obligate mutualism where the fig trees need the pollinators in order to reproduce while the pollinators require the figs as their brood sites. Ficus deltoidea is a common dioecious fig tree in Malaysian oil palm plantations and may contribute figs and other resources for birds and other animals that would otherwise not be able to survive in the plantations. *F. deltoidea* is a true epiphyte – a growth form that is very rare in *Ficus*. It has numerous named varieties and the taxonomic and biological status of these varieties is unclear, as is the extent to which different varieties support different Blastophaga pollinators. A period of 18 months practical study was carried out in Malaysian oil palm plantations and the *F. deltoidea* germplasm collection in Universiti Sultan Zainal Abidin. F. deltoidea was found to be one of ten fig tree species in oil palm plantations in Peninsular Malaysia, but the only true epiphyte. The figs of different varieties of *F. deltoidea* showed a high degree of variation in terms of flower numbers, tepal numbers and size of the figs. Male and female F. deltoidea var. angustifolia produced new leaves and figs all year around. There was no existence of nonpollinating fig wasps in this species. A *Blastophaga* sp. foundress can enter several figs to lay their eggs and pollinate, but many figs on female plants were not entered and many figs aborted. The pollinators liked to enter figs on their natal male plants if the figs were available. Experiments found pollinators preferred to enter figs of their own variety. The high specificity of the pollinators suggests that many varieties might be distinct species. The limited dispersal of *Blastophaga* sp. helps them to reproduce but results in the pollinator shortage that is prominent in the female figs. The pollinator can be said to be ahead of the trees. Greater understanding of how they interact may help explain how the mutualism can persist. This is important because the mutualism between F. deltoidea and its pollinators can increase biodiversity in oil palm plantations. This study can also increase our understanding of coevolution between plants and insects.

TABLE OF CONTENTS

ABSTR	ACT.		. iii
TABLE OF CONTENTS iv			. iv
LIST O	F TAB	LES	. . x
LIST O	F FIGl	JRES	kiv
LIST O	F ABB	REVIATION x	xii
Chapte	r 1		1
1.1	Introc	duction	1
1	.1.1	Plant and insect interactions and pollination in general	1
1	.1.2	Nursery pollination	4
1	.1.3	Ficus trees	5
1	.1.4	Ficus pollination	7
1	.1.5	Fig wasps	11
1	.1.6	Host specificity in <i>Ficus</i>	13
1	.1.7	Ficus deltoidea	15
1	.1.8	Epiphytes and Hemiepiphytes (stranglers)	18
1	.1.9	Oil palm	19
1.2	Rese	arch aims and objectives	22
1.3	Thesi	is outline	22
Chapte	r 2		24
2.1	Study	/ sites	24
2	.1.1	Banting oil palm plantation	24
2	.1.2	Dengkil oil palm plantation	27
2	.1.3	Batu Pahat oil palm plantation	27
2	.1.4	Bagan Serai oil palm plantation	27
2	.1.5	Tembila oil palm plantation	29
2	.1.6	Terengganu germplasm collection	29
2.2	Fig d	evelopment phases	31
Chapte	r 3		35
3.1	Introc	duction	35
3	.1.1	Epiphytes	35
3	.1.2	Ficus as epiphytes	37

3	.1.3	Epiphytes in oil palm plantations	38
3	.1.4	Frugivores	39
3.2	Objec	ctives	40
3.3	Metho	ods	40
3	.3.1	Study sites	40
3	.3.2	Epiphyte assessments	41
3	.3.3	Ficus deltoidea assessment	41
3	.3.4	Frugivore observations	42
3	.3.5	Statistical methods	42
3.4	Resu	lts	43
3	.4.1	<i>Ficus deltoidea</i> var. <i>angustifolia</i> in plantations near Kua 43	la Lumpur
3	.4.2	Epiphyte loads	45
3	.4.3	Distribution of <i>Ficus</i> species	47
3	.4.4	Differences in establishment heights of Ficus on oil palms	54
3	.4.5	Ficus deltoidea var. angustifolia and Ficus deltoidea var. 60	deltoidea
3	.4.6	Ficus deltoidea var. angustifolia in Banting	62
3	.4.7	Frugivore observations	62
3.5	Discu	ission	64
3	.5.1	Epiphytes	64
3	.5.2	Ficus species	65
3	.5.3	Ficus deltoidea as an epiphyte	66
3	.5.4	Frugivores	68
3.6	Conc	lusions	68
Chapte	r 4		69
4.1	Introc	duction	69
4	.1.1	Ficus	69
4	.1.2	Ficus pollination	70
4	.1.3	Species concepts in <i>Ficus</i>	72
4	.1.4	Ficus deltoidea	75
4	.1.5	F. deltoidea classification	76
4	.1.6	F. deltoidea varieties	76
4.2	Objec	ctives	79
4.3	Metho	ods	79

4	.3.1	Study varieties	79		
4	.3.2	Study sites and sampling methods			
4	.3.3	Fig characterisations			
4	.3.4	Fig wasp characterisations			
4	.3.5	Statistical methods			
4.4	Resu	ts			
4	.4.1	Variation among populations of var. angustifoli	<i>a</i>		
	4.4.	.1 Variation in female figs of var. angustifolia	a		
	4.4.	.2 Between-population variation in male fig	s of var. <i>angustifolia</i> 88		
	4.4.	.3 Variation in Blastophaga sp. pollinato angustifolia	rs reared from var. 90		
4	.4.2	Variation between varieties of <i>F. deltoidea</i>			
	4.4.	.1 Characterisation of female figs			
	4.4.	.2 Male Figs Characterisation	101		
	4.4.	.3 Fig wasp Characterisation	110		
4	.4.3	Fig and fig wasp inter-relationships	113		
	4.4.	.1 Style length and ovipositor length relation	ships 113		
4	.4.4	Flower utilisation and pollination	119		
4	.4.5	Inflorescence design in closely related fig trees different numbers of flowers in the figs	: The effects of having 124		
	4.4.	.1 Relationship between fig flower numbers	and other characters		
4.5	Discu	ssion			
4	.5.1	Variation between populations of <i>F. deltoidea</i>	var. angustifolia138		
4	.5.2	Variation between varieties of <i>F. deltoidea</i>			
4	.5.3	Variation between fig wasps from three variet	ies of Ficus deltoidea		
4	.5.4	Variation in reproductive output in different va 141	rieties of <i>F. deltoidea</i>		
4.6	Conc	usions	142		
Chapte	er 5		143		
5.1	Intro	uction			
5	.1.1	Specificity of interactions between plants and i	nsects 143		
5	.1.2	Ficus pollination			
5	.1.3	The breakdown of host specificity			

	5	.1.4	Spe	cificity of pollination in Ficus deltoidea varieties	148
	5.2	Obje	ctives	5	150
	5.3	Meth	ods		150
	5	.3.1	Stuc	ly species	150
	5	.3.2	Stuc	ly sites	151
	5	.3.3	Hos	t choice by <i>Blastophaga</i> sp. (var. angustifolia)	152
		5.3.3	3.1	Transplant experiment one	152
		5.3.3	3.2	Transplant experiment two	155
		5.3.3	3.3	Transplant experiment three	156
		5.3.3	3.4	Transplant experiment four	159
	5	.3.4	Hos	t choice by <i>Blastophaga</i> sp. (var. <i>deltoidea</i>)	159
	5	.3.5	Stat	istical methods	159
	5.4	Resu	ılts		160
	5	.4.1	Hos	t choice by <i>Blastophaga</i> sp. (var. angustifolia)	160
		5.4.	1.1	Transplant experiment 1	160
		5.4.	1.2	Transplant experiment 2	166
		5.4.	1.3	Transplant experiment 3	174
		5.4.	1.4	Transplant experiment 4	177
	F	lost ch	noice	by Blastophaga sp. (var. deltoidea)	183
	5.5	Discu	ussio	٦	184
	5	.5.1	Hos	t choice by <i>Blastophaga</i> sp. (var. <i>angustifolia</i>)	184
		5.5.	1.1	Host preference between varieties	184
		5.5.	1.2	Preferences between male and female plants	186
	5	.5.2	Hos	t choice by <i>Blastophaga</i> sp. (var. <i>deltoidea</i>)	187
	5.6	Conc	lusio	ns	188
Ch	apte	er 6			189
	6.1	Introd	ductio	on	189
	6	5.1.1	Phe	nology	189
	6	5.1.2	Ficu	s phenology	190
	6	5.1.3	Ficu	s deltoidea	192
	6.2	Obje	ctives	3	193
	6.3	Meth	ods		193
	6	5.3.1	Stuc	ly sites and sampling methods	193
	6	.3.2	Leaf	phenology	194
	6	.3.3	Fruit	ting phenology	194

6	6.3.4	Fig development cycle195		
6	6.3.5	Statistical methods 195		
6.4	Resu	lts		
6	6.4.1	Leaf phenology197		
6	6.4.2	Fruiting phenology		
	6.4.2	2.1 Male plants		
	6.4.2	2.2 Female plants		
6	6.4.3	Fig development		
6	6.4.4	Pollinator limitation		
6.5	Discu	ussion		
6	6.5.1	Leaf phenology230		
6	6.5.2	Fig phenology230		
6	6.5.3	Pollination limitation		
6.6	Conc	lusions		
Chapte	er 7			
7.1	Introd	duction		
7	7.1.1	Fig tree pollination		
7	.1.2	Re-emergence of foundresses		
7	7.1.3	Pollinator sex ratio variation		
7	'.1.4	Ficus deltoidea reproduction		
7.2	Obje	ctives		
7.3	Meth	ods		
7	7.3.1	Study varieties		
7	.3.2	Study sites and experimental methods		
7	7.3.3	Statistical methods		
7.4	Resu	lts		
7	.4.1	The number of fig wasps entering a fig		
7	.4.2	Emergence rates from experimental figs		
7	.4.3	Brood size in re-emergence and non-emergence foundresses250)	
7	.4.4	Brood sizes in the first and subsequent figs entered		
7	.4.5	Pollination by foundresses that entered one or more female f 254	igs	
7	.4.6	Variation in offspring sex ratios in figs entered by a single foundre 256	ess	
7.5	Discu	ussion		

7	.5.1	Multiple foundresses	263
7	.5.2	Walking rates	263
7	.5.3	Brood size comparisons	264
7	.5.4	Seed production	265
7	.5.5	Sex ratios	265
7.6	Conc	lusions	266
Chapte	r 8		267
8.1	Epiphytes in oil palm plantations		
8.2	Ficus	s deltoidea	268
8.3	Is Ficus deltoidea a complex of different biological species? 274		274
8.4	Conclusions		275
8.5	Future work276		
References			

LIST OF TABLES

- Table 3.2. Frequency of bare trunks and epiphyte groups in five different plantations.

 46

- Table 4.4. The numbers of female flowers and tepals in female figs of var. angustifolia from four sites. Values with different superscripts in a column differed significantly in Mann-Whitney tests (P< 0.05)..... 85

- Table 4.7 The diameter of receptive male var. angustifolia figs and their maximum diameters of male figs at four sites.

 89

- Table 4.13 The number of female flowers and the lengths of styles in female figs of six varieties of *F. deltoidea*. Values with different superscripts in a column differed significantly (Tukey tests, P < 0.001). * indicates that values differed significantly between trees of the same variety... 94

Table 4.23 The femur lengths of female fig wasps reared from three varieties of F. deltoidea. Values with different superscripts in a column differed

- Table 4.24 The number of seeds in three figs of three varieties of *F. deltoidea*. Values with different superscripts differed significantly (Tukey Tests P< 0.05). * indicates that values differed significantly between trees of the same variety. 121
- Table 4.25 The contents of male figs (Mean ± SD) of three varieties of F. deltoidea. Values with different superscripts in a row differed significantly (tests vary, see above, P< 0.001). * indicates values differed
- Table 4.26 Within-variety relationships between the maximum diameter of female figs and the numbers of female flowers they contain using
- Table 4.27 Within-variety relationships between the maximum diameter of male figs and the numbers of female flowers they contain using Pearson
- Table 4.28 Within-variety relationships between the number of male flowers and the number of female flowers in male figs of Ficus deltoidea using
- Table 4.29. The ratio of anthers to ovules in male figs of six varieties of F.
- Table 4.30 The relationships between numbers of female flowers and the number of female pollinator offspring in varieties of Ficus deltoidea.
- Table 4.31 Correlations between the total number of female flowers and the number of galls (successful and empty combined) in male figs of Ficus
- Table 4.32 Correlations between the number of female flowers and the number of seeds produced in the female figs of Ficus deltoidea.135
- Table 4.33 Correlations between the number of female pollinator offspring and the number of male flowers in male figs using Pearson correlation.
- Table 4.34 The ratios of male flowers to female pollinator offspring in three varieties of Ficus deltoidea......137
- Table 5.1 Ficus deltoidea individuals obtained from TGC and positioned in pots around potential pollinator donor male individuals of var. angustifolia growing as epiphytes in an oil palm plantation in Banting.
- Table 5.2. The entry of pollinators into figs of *Ficus deltoidea* varieties positioned around epiphytic male var. angustifolia in Banting plantation.

- Table 5.4. The entry into figs of Ficus deltoidea varieties in the TerengganuGermplasm Collection (TGC) after sources of var. angustifolia fig waspswere added.167
- Table 5.6. The contents of aborting figs collected from potted trees placed around male var. *angustifolia* that were releasing pollinators.... 176

- Table 6.2. Types of leaf on the 39 accessible var. angustifolia fig trees.200
- Table 6.3 Juvenile leaves on var. angustifolia in Banting plantation as a
proportion of total leaves based on 34 observations at approximately 14
day intervals.200

- Table 7.1 The numbers of naturally-visited male figs with foundress bodies either in the fig cavity or with the body stuck head-outwards in the ostiole. Means do not include figs with no foundress corpses. 245
- Table 7.2 The numbers of naturally-visited female figs with foundress bodies either in the fig cavity or with the body stuck head-outwards in the ostiole. Means do not include figs with no foundress corpses. 246
- Table 7.4 The offspring sex ratio ratios (number of male offspring/total offspring) in the first fig entered by a *Blastophaga* sp. foundress.257
- Table 7.5 The offspring sex ratio (number of male offspring/total offspring) in the subsequent figs entered by a *Blastophaga* sp. foundress... 258
- Table 7.6 Offspring sex ratios in first and subsequently entered figs 260

LIST OF FIGURES

Figure 1.1 Adult female and male fig wasps (<i>Blastophaga</i> sp. from <i>Ficus deltoidea</i> var. <i>angustifolia</i>)12
Figure 2.1 The six study sites in Peninsular Malaysia
 Figure 2.2 Climate data for Banting oil palm plantation, Selangor. Monthly percentage of relative humidity (■) and monthly average temperatures (•) for June 2016-August 2017 as measured by KLIA Sepang (19 km from the research site).
 Figure 2.3 Climate data for Banting oil palm plantation, Selangor. Monthly percentage of relative humidity (■) and monthly average temperatures (•) for June 2016-August 2017 as measured by KLIA Sepang (19 km from the research site).
Figure 2.4 Climate data for Batu Pahat oil palm plantation, Johor. Monthly mean rainfall and average temperature (■) for June 2016-August 2017 as measured by Batu Pahat station
Figure 2.5 Climate data for Bagan Serai oil palm plantation, Johor. Monthly mean rainfall and average temperature (■) for June 2016-August 2017 as measured by Chersonese Est. Kuala Kurau station
Figure 2.6 Climate data for Tembila oil palm plantation, Johor. Monthly mean rainfall and average temperature (■) for June 2016-August 2017 as measured by Institut Pertanian Besut station
 Figure 2.7 Fig developmental stages in male <i>F. deltoidea</i> var. <i>angustifolia</i>. (A) Young fig. (B) The fig is receptive to pollinators. (C) Development of fig wasp offspring. (D) The next generation of wasps are released. (E) After the wasps have emerged. (F) An aborting, un-entered fig
 Figure 2.8 Fig developmental stages on female <i>F. deltoidea</i> var. angustifolia fig trees. (A) Young fig. (B) The fig is receptive to pollinators. (C) Development of the seed. (D) Late Development of the seed. (E) Mature seeds ready to be dispersed (F) Aborting fig
Figure 3.1 The cover of oil palm trunks by different epiphyte groups. Open bars = bare trunks, solid bars = <i>Ficus deltoidea</i> , grey bars = other <i>Ficus</i> , spotted bars = bryophytes, hashed bars = pteridophytes, horizontal lines = other Angiosperms
Figure 3.2 Frequencies of <i>Ficus</i> species in Banting plantation. <i>F. deltoidea</i> (a) = var. angustifolia, <i>F. deltoidea</i> (d) = var. deltoidea, <i>F. deltoidea</i> (t) = var. trengganuensis
Figure 3.3 Frequencies of <i>Ficus</i> species in Dengkil plantation. <i>F. deltoidea</i> (a) = var. angustifolia, <i>F. deltoidea</i> (d) = var. deltoidea, <i>F. deltoidea</i> (t) = var. trengganuensis

- Figure 3.8 The heights of apparent establishment sites of *Ficus* species in Dengkil plantation. *F. deltoidea* (*a*) = *Ficus deltoidea* var. *angustifolia*. The lines in boxes indicate the median, the boxes indicate upper and lower quartiles with extreme values at the end. Values with different superscripts differed significantly in Mann-Whitney tests (P< 0.05).56
- Figure 3.9 The heights of apparent establishment sites of *Ficus* species in Batu Pahat plantation. *F. deltoidea* (a) = *Ficus deltoidea* var. *angustifolia*. *F. deltoidea* (d) = *Ficus deltoidea* var. *deltoidea*. The lines in boxes indicate the median, the boxes indicate upper and lower quartiles with extreme values at the end.
- Figure 3.11 The heights of apparent establishment sites of *Ficus* species in Tembila plantation. *F. deltoidea*(*t*) = *Ficus deltoidea* var. *trengganuensis*. The lines in boxes indicate the median, the boxes indicate upper and lower quartiles with extreme values at the end. Values with different superscripts differed significantly in Mann-Whitney tests (P< 0.05).59

- Figure 4.4 Diameter at receptivity and the number of female flowers in female figs of *F. deltoidea* varieties. (■) = var. angustifolia, (●) = var. deltoidea, (▲) = var. trengganuensis, (★) var. bilobata, (●) var. kunstleri and (♦) var. motleyana).

- Figure 4.11 The relationship between the ovipositor length of *Blastophaga* sp. (solid bars) and style lengths in male figs of *F. deltoidea* (open bars). A = var. *angustifolia*, B = var. *deltoidea*, C = var. *trengganuensis*. 118

Figure 4.15 The offspring sex ratios in relation to total offspring in three varieties of *Ficus deltoidea*. (Mann-Whitney test, P < 0.05)....... 123

- Figure 4.18 Relationships between the number of male flowers and the number of female flowers in male figs (■) = var. angustifolia, (●) = var. deltoidea, (▲) = var. trengganuensis, (★) var. bilobata, (●) var. kunstleri and (♥) var. intermedia. Each data point represents the individual fig. 129

- Figure 4.23 Relationships between the mean number of seeds per fig and the number of female flowers in female figs. (■) = var. angustifolia, (▲) = var. trengganuensis, (●) = var. deltoidea). Each data point represents one fig. 135
- Figure 5.1 Varieties of *Ficus deltoidea* encircling large epiphytic male *Ficus deltoidea* var. *angustifolia*. Left donor tree D7, right donor tree D22.
- Figure 5.2. The arrangement of pots containing varieties of *Ficus deltoidea* in the Terengganu Germplasm Collection (TGC). The var. *angustifolia* that were releasing pollinators were FDA 1, FDA 2, FDA 3, FDA 4 and FDA 5.

- Figure 5.6 Pollination rates in four transplanted *Ficus deltoidea* var. *angustifolia*. Solid bars = male trees, open bars = female trees. 164

- Figure 5.9. Initial and final entry rates on the male trees of var. *angustifolia* in experiment two. Solid bars = entry rates at the start of the experiment, open bars = entry rates at the end of the experiment. FDA 1, FDA 2, FDA 3, FDA 4 and FDA 5 had previously-entered figs at the start of the experiment. FDA 9 was the only control tree with figs that were entered.

- Figure 5.16. The numbers of entered male and female figs around different donor trees (Solid bars = male recipient trees, open bars = female recipient trees). No female figs were entered around tree FDA1.182
- Figure 5.17 The relationship between the proportion of entered figs and different donor trees. (Solid bars = male trees, open bars = female trees).

- Figure 6.7 Changes over time in crop sizes on 21 male trees...... 205

- Figure 6.9 The presence of young AB phase figs on male trees of *F. deltoidea* var. *angustifolia* in Banting oil palm plantation in each month between May 2016 and August 2017. (Grey bars = Figs were present, white bars = figs were absent and black bars = when the trees were cut off.. 207
- Figure 6.10 The presence of DE phase figs on male trees of *F. deltoidea* var. *angustifolia* in Banting oil palm plantation in each month between May 2016 and August 2017. (Grey bars = Figs were present, white bars = figs were absent and black bars = when the trees were cut off)....... 208
- Figure 6.12 Frequency of the fig developmental phases on 18 female fig trees.

- Figure 6.14 The presence of mature phase figs on female trees of *F. deltoidea* var. *angustifolia* in Banting oil palm plantation in each month between May 2016 and August 2017. (Grey bars = figs were present, white bars = figs were absent and black bars = when the trees were cut off). 213
- Figure 6.15 The length of developmental phases of figs on male trees. 124 unpollinated A phase figs were marked in week zero. Grey bars = AB phase, white bars = C phase, black bars = DE phase, hashed bars = aborting figs and spotted bars = absent figs......Error! Bookmark not defined.
- Figure 6.16 The length of developmental phases in figs on female trees. 185 small A phase figs were marked in week zero. Grey bars = AB phase, white bars = C phase, black bars = DE phase, hashed bars = aborting and spotted bars = absent figs......Error! Bookmark not defined.

- Figure 6.20 The relationship between the percentage of entered figs in 18 female trees and the distance to the nearest adjacent mature male tree.

- Figure 7.11. The developing male figs pollinated by a single foundress.255
- Figure 7.13 *Blastophaga* sp. offspring sex ratios (number of male offspring/total offspring) in relation to oviposition sequence. ... 259
- Figure 7.14 The relationship between the sex ratio and brood size in experimental figs entered by a single foundress. Each point indicates the contents of one fig. One foundress generated 1- 4 data points.. 261

LIST OF ABBREVIATION

cm	Centimetre
com.	Communication
Fd (a)	Ficus deltoidea var. angustifolia
Fd (d)	Ficus deltoidea var. deltoidea
Fd (t)	Ficus deltoidea var. trengganuensis
GLM	Generalized Linear Model
KESEDAR	Lembaga Kemajuan Kelantan Selatan
KLIA	Kuala Lumpur International Airport
km	Kilometre
Lab.	Laboratory
mm	Millimetre
NPK	Nitrogen, phosphorus, kalium
Pers.	Personal
SD	Standard deviation
SE	Standard error
sqrt	Square root
TGC	Terengganu Germplasm Collection
UKM	Universiti Kebangsaan Malaysia
UniSZA	Universiti Sultan Zainal Abidin
var.	Varieties

xxii

Chapter 1

General Introduction

1.1 Introduction

1.1.1 Plant and insect interactions and pollination in general

Plants routinely interact with insects. These interactions be either beneficial or detrimental to one or both parties. Interactions between plants and the herbivores that eat them are often highly species-specific. Many plant- feeding insects are also specific in the parts and developmental stages of the plants that they feed on (Bernays and Chapman, 1994). Plant-feeding insects can be divided into three categories, those which are monophagous (feeds on a single host plant), oligophagous (feeds on a number of related plants in the same family) or polyphagous (feeds on a wide range of host plants from different families) (Bernays and Chapman, 1994). The specificity of this relationship varies between feeding guilds - groups of insects that feed on the same parts of a plant in similar ways (Novotny et al., 2010). Leaf miners are often more specific than leaf chewing species while seed feeders and gall-formers are typically highly specific. Insect herbivores can feed by chewing, sap sucking, epidermis scraping, flower feeding, root or shoot mining, boring, gall induction and seed predation (Gossner et al., 2014). The interactions involving plant-feeders are usually antagonistic and harmful to the plants. However, in some cases the plants benefit from being eaten. An example of this is pollination. Insects are the animal group most widely utilised by plants for pollination.

Pollination is an unusual type of herbivory where the plants provide rewards for pollinators and the pollinator behaviour and structures evolve to help collect the rewards. As such, it is a mutualistic interaction where both participants can benefit. The rewards provided to pollinators by plants include pollen, nectar, oils, wax, resin and edible trichomes (Pansarin and Maciel, 2017). Pollinators, like other herbivores, can be either generalist or specialist. Generalists take rewards from many plants while specialists get their rewards for pollination services from one or a few species of plants.

Generalists are often more abundant than specialists because the pollinators can switch to other hosts if one host is absent (Bosch et al., 2009). However, pollination by generalists has some disadvantages for the plants because it can lead to interspecific pollen transfer and pollen deposition (Morales and Traveset, 2008). Heterospecific pollen deposition causes low fertilization and seed production. On the other hand, specialists visit a limited range of flowers and the risk of deposition of heterospecific pollen and loss of pollen through transfer to other plant species is low (Maldonado et al., 2013).

The ability to remove and deposit pollen on the stigmas of conspecifics varies between pollinators and between flowers. Host plant specialists tend to be more effective at removing the pollen from host anthers and depositing them onto host stigmas (Parker et al., 2016). For example, pollinator effectiveness in removing pollen from the herb *Knautia arvensis* (Dipsacaceae) that is visited by both generalist and specialist pollinators was examined by (Larsson, 2005). He showed that a more specialist pollinator, a solitary bee *Andrena hattorfiana* (Andrenidae) moved more pollen in a single visit compared with generalist visitors (bumblebees and flies).

Pollinators provide a service to plants in order to get a reward and the plants need to make sure that when visitors collect their rewards that pollination is taking place. In order to achieve this, plants have evolved morphological specializations to attract their most effective pollinators (Sakai, 2002). Similarly, the morphological adaptations of specialist pollinators mean they are well adapted to exploit efficiently the flowers they visit (Kahnt et al., 2017). Two fly families in Africa, Nemestrinidae and Tabanidae have a proboscis up to three time longer than their body length in order to gain the rewards provided by their long tubed host flowers (Goldblatt and Manning, 2000). The plants develop long tubes to prevent generalist pollinators getting access to the rewards and thereby enhance the chance of successful pollination when visited by the flies adapted to pollinate them.

Another example where specialist pollinators have evolved their morphology in order to maximize getting the rewards from their host plants can be seen in South African *Rediviva* bees (Melittidae) where females have evolved extremely elongated forelegs in order to get to the oil rewards from the long paired floral spurs of *Diascia* species (Scrophulariaceae). They also have highly modified structures to damage oil-

producing cells and to collect the oil. Phylogenetic studies have shown that their ancestors had short legs and *Diascia* was their host plant (Kahnt et al., 2017).

In order to achieve pollination, plants need to attract the pollinators by giving appropriate cues to help them be recognised and located. Host-plant selection by pollinators is usually linked with floral traits like colour, size, volatiles (Burger et al., 2010), flower surface texture, symmetry and shape (Glover, 2011). Floral traits can lead to specific interactions between a plant and pollinator and this can cause reproductive isolation and facilitate speciation (Klahre et al., 2011). A study by (Bradshaw and Schemske, 2003) discovered that a mutation in a single gene can have a major influence on pollinator preferences between closely-related species that leads to host shifts. Klahre et al. (2011) compared the significance of two floral traits (colour and scent) in attracting a pollinator hawkmoth (Sphingidae) and showed they were equally weighted. On the other hand, a study by (Hirota et al., 2012) suggested that colour traits had a more significant role than scent in shifting pollinators from visiting daylily *Hemerocallis fulva* L. var. *aurantiaca* (Baker) M. Hotta (Asphodelaceae) to nightlily *Hemerocallis citrina* var. *vespertina* (H. Hara) M. Hotta (Asphodelaceae).

To find flowers, floral colour (visual cues) together with olfactory cues (floral scents) appear to be the most critical for many pollinators (Burger et al., 2010). As with most other insects associated with plants, pollinators must be able to recognize suitable hosts from a distance away and upon arriving on a potential host they must make sure that they are choosing one with the correct good quality rewards (Bernays and Chapman, 1994). Angiosperms associated with specialized pollinators usually emit highly specific volatiles to allow their pollinators to discriminate and recognize them (Sakai, 2002; Milet-Pinheiro et al., 2013).

Of all the cues, floral volatiles emitted from receptive flowers (those at the developmental stage when their pollen is mature and ready to be collected and their stigmas can utilise pollen) are a key factor in pollination mutualisms (Hossaert-McKey et al., 2016). Volatiles can also allow the pollinators to distinguish between rewarding and non-rewarding plants of the same species via the nectar's odours (Howell and Alarcon, 2007). Some pollinators depend on the wind to bring floral scents to them and the insects then perform positive anemotaxis (orientate up-wind) to locate the host, and zigzagging flight behaviour helps sustain contact with the scent (Borges, 2016). Some insects can orient into the wind from the ground. When they perceive the

odour, they will fly for a short distance and land on the ground again. They will take off again when they continue to respond to the odours until they reached the target (Bernays and Chapman, 1994). Some small insects depend on the wind and this gives them the opportunity to fly for greater distances, but at the cost of not being able to control their direction of flight (Compton, 2002). Once landed on a suitable host, pollinators depend on short-range cues to decide whether the flower will provide the rewards they want or not. Short range cues include morphological features and the surface texture of the flowers (van Noort, 2003).

1.1.2 Nursery pollination

The rewards that are most usually associated with pollination are nectar and pollen. These rewards are provided to insects such as most bees, moths and flies and are often associated with quite generalist plant-pollinator interactions. A less common, and often highly specific pollinator reward is involved with nursery pollination where the reward for pollination services is mating sites and sites where insects offspring can develop (Waser and Ollerton, 2006). This interaction, is also known as brood-place pollination (Hossaert-McKey et al., 2016). In this type of pollination, the insects need to pollinate the flowers to secure a place for their offspring to live and feed, and it is the resources provided to their offspring that represents the reward (Duthie and Nason, 2016). The larvae of the pollinators typically feed on the seeds, ovules or other female tissues within the flowers of their host plants (Bronstein, 2001; Hembry and Althoff, 2016). This kind of interaction is usually obligate and there is a high degree of specificity between each host plant and its pollinators (Borges, 2016).

Pollination in a nursery pollination is started when (1) the plants release a specific volatile compound when they are receptive and need to attract pollinators (Hossaert-McKey et al., 2016) (2) the volatile diffuse and are transported mostly by the wind to where pollinators occur (3) the pollinators detect the chemical cues with their antennae (van Noort, 2003) (4) the sensory systems of the pollinators interpret the cue (Kjellberg et al., 2005b) and (5) lastly, the pollinators move to locate their reward's location (Borges, 2016). Most insects associated with nursery pollination are short-lived as adults and thus a learning process is not important in locating their hosts (Borges, 2016). The scent emitted by the host must be very effective and specific to ensure the success of finding the host by their obligate partner (Hossaert-McKey et al., 2010). The high specificity in this form of mutualism requires synchronization between the development times of the fruit and the development of their pollinators (Sakai, 2002).

The availability of receptive flowers at the same time as the release of pollinators helps maintain the populations of the pollinators (Peng et al., 2010).

Sakai (2002), divided nursery pollination into three major groups based on how the pollinators breed in the flowers they pollinate; (1) the brood are ovule/seed parasites (2) the brood are pollen parasites and (3) the brood feed on the decomposing flowers after pollination has taken place. The receptive flowers may also act as the sites for mating for the pollinators as well as the brood sites for the pollinator's progeny (Borges, 2016). Nursery pollinators mostly come from the insect Orders Hymenoptera, Lepidoptera, Coleoptera, Diptera and Thysanoptera (Sakai, 2002).

Nursery pollination provides a model for studying the evolution of cooperation and diversification among populations and species (Hembry and Althoff, 2016). The speciation of one party caused another party to diversify. Examples of nursery pollination are provided by yuccas (*Yucca*) (Asparagaceae) and yucca moths (*Tegeticula*) (Prodoxidae), leafflowers (*Glochidion*) (Phyllanthaceae) and leafflower moths (*Epicephala*) (Gracillariidae) (Borges, 2016), saxifrages (Saxifragaceae) and *Greya* moth (Prodoxidae), *Silene* (Caryophyllaceae) species and *Hadena* moths (Noctuidae), senita cacti (Cactaceae) and senita moths (Crambidae) and fig trees *Ficus* (Moraceae) and fig-wasps (Agaonidae) (Hembry and Althoff, 2016). Of all the organisms involved in nursery pollination mutualisms, the fig tree-fig wasp mutualism is the most widely studied and involves the largest number of plant and insect species.

1.1.3 Ficus trees

Fig trees belong to the genus *Ficus* and family Moraceae. Their distribution is worldwide in warmer areas, with almost 900 described species making them one of the largest genera of land plants (Weiblen, 2002) and the only woody genus found in almost all lowland tropical forests, though often at low densities (Ghazoul and Sheil, 2010). Fig trees occupy a very wide-ranging spectrum of habitats, from wet lands to deserts and have coevolved with their highly specific fig wasp pollen carriers (Kjellberg et al., 2001). *Ficus* is likely to have originated from western Asia Minor but can now be found globally in tropical, sub -tropical and some warm temperate areas (Mat et al., 2012). The genus is possibly the most important source of food to frugivores in lowland tropical rainforests (Harrison, 2005) and more generally (Shanahan et al., 2001).

Trees of the genus *Ficus* produce large quantities of white latex in their inflorescences, branches and leaves, have stipules covering the new buds at the twig tips, ring scars on their twigs left by the stipules that have fallen off, and urn-shaped inflorescences with numerous florets (Burrows and Burrows, 2003). Fig trees exhibit a wide range of growth forms from epiphytes, hemiepiphytes, climbers, shrubs and as trees (Herre et al., 2008). They also can be bushy with leaves organised in spiral and ascending twigs (Mat et al., 2012). All fig trees bear an enclosed urn-shaped inflorescence that is called a fig that is lined with many tiny florets on its inner surface (Cook and Rasplus, 2003). This inflorescence is closed by apical bracts around an ostiole (Kjellberg et al., 2005b). The ostiole is the path that allows the entry of the pollinators. Ecologically, a fig is a compound fruit but as it is both the site of pollination and pollen production and the unit of dispersal, it acts as an inflorescence and a fruit (Gonzaga, 2012). The sizes of figs are different depending on the species. They can be from pea to peach sized or larger and the colour of ripe figs also differs from red, yellow, and green or purple (Borges et al., 2008). Different fig trees appear to produce different volatile compounds that their associated fig wasps recognize (Herre et al., 2008). Later, the figs ripen to form 'pseudo-fruits' containing the seeds, each of which is technically a fruit (Ghazoul and Sheil, 2010).

Fig development can be divided into five phases (1) pre-female flowers – young figs and before the ostiole is open, (2) female flowers – receptive for pollination and female wasp start entering their ostioles, (3) interfloral – right after the ostiole is closed the ovaries occupied by wasps larvae will develop into galls, (4) male flower - the next generation of wasps are released from the galled ovules where they developed and (5) a postfloral phase – after the wasps have emerged and the figs ripen (Galil and Eisikowitch, 1968). These developmental phases are widely used in studies of fig development, though not all the stages are necessarily present in all the figs of each species.

Figs also support a diverse community of non-pollinating fig wasps, either parasitoids, or ovule gallers. Similar to any mutual relationship, the presence of a third party always creates constraints and weakens the strength of rewards in the mutualism. Seed gallers will compete with the figs' pollinators for food resources, and parasitoids attack pollinator larvae in figs (Weiblen and Bush, 2002) and compete with the pollinating wasps for oviposition sites (Kerdelhue et al., 2000). Typical seed-eating fig wasps are rare (Wang et al., 2014). The development of non-pollinating fig wasps does not benefit the plant and they have been referred to as 'parasites' of the mutualism

(Compton and van Noort, 1992). The specificity of the non-pollinating fig wasp was less specific (van noort, 2003).

Fig trees are believed to act as keystone species in tropical forests (Yang et al., 2002) and specially in Malaysia (Lambert, 1991), India (Kannan and James, 1999), South Africa (Bleher et al., 2003), Panama (Korine et al., 2000). However, due to relatively low densities of figs in the Afrotopical region (continental Africa and the Malagasy region), Gabon (Gautier-Hion and Michaloud, 1989) and Uganda (Chapman et al., 2005), some authors did not consider them as keystone species there (Goodman and Ganzhorn, 1997). Their keystone status elsewhere is because they sustain populations of vertebrate frugivores, which often comprise a high proportion of the total vertebrate biomass in tropical rainforests (Thornton et al., 1996).

The year-round production of figs and their abundance attracts large numbers of frugivores like hornbills, bulbuls, monkeys, squirrels, fruit pigeons, bats and non-volant mammals to consume them (Shanahan et al., 2001). The number of frugivores locally usually peaks at the time the figs ripen (Mackay et al., 2018). A study done by (Lambert, 1991) revealed that at least 60 species of bird and 17 species of mammals consumed figs in Peninsular Malaysia. Fig trees also support numerous other species that live inside the figs, including microorganisms, mites (Compton et al., 1994), nematodes (Jauharlina et al., 2012) and flies (Lanchaise et al., 1988; Macgowan and Compton, 2018). In addition to animals, fig trees provide habitats that can host saprophytes, epiphytes, parasitic plants and also shade-enduring flowering plants (Yang et al., 2002).

1.1.4 *Ficus* pollination

The pollination between fig trees and fig wasps is one of the classic examples of obligate mutualism (Deng et al., 2016). For obligate mutualists, the loss of one species will have critical effects on the other species, possibly leading to cascading effects across trophic levels because of the high degree of dependency and coevolution between the species (Cook and Rasplus, 2003). The mutualism between fig trees and fig wasps involves the pollination by the fig wasps and as a reward, the fig trees provide sites for mating, development and reproduction sites of the progeny together with the nutrition for the brood to complete their life cycle (Wachi et al., 2016). Fig trees depend on the fig wasps for transmission of their pollen and therefore for viable seed production (Herre et al., 2008). With a few rare exceptions, the fig wasps only

reproduce within the figs of one fig tree species and in return, the fig trees are often pollinated exclusively by one species of wasp (Yang et al., 2002; van Noort, 2003). Examples where a fig tree is pollinated by more than one species of fig wasp are nonetheless increasing (Yang et al., 2015).

The pollination in *Ficus* starts when specific volatile compounds are released from receptive figs to attract the specific fig wasps associated with each species of fig tree (Hossaert-McKey et al., 2016). The female fig wasp (known as a foundress when she enters a receptive fig), detects the volatiles and flies towards the receptive figs (Molbo et al., 2003). The foundresses can be seen hovering around receptive figs (Ware and Compton, 1992) before penetrating through a narrow (usually) bract-lined tunnel ostiole (Liu et al., 2013), which closes soon afterwards, sealing the foundress wasps in a 'tomb blossom' (Weiblen, 2002). This ostiole become loose during the receptivity period (Yang et al., 2002; Liu et al., 2013). On entry into a receptive fig, they lose their wings and parts of their antennae, so they cannot fly away to subsequent figs (Suleman et al., 2013b).

Once inside they pollinate the flowers (Nefdt and Compton, 1996) using the pollen from their natal fig (Herre et al., 2008) and at the same time lay eggs in some of the flowers (Kjellberg et al., 2005b). Only one egg is laid per flower (Ghana et al., 2012). Once the flowers have been probed, she and other fig wasp foundresses will avoid ovipositing their eggs in those flowers (Jousselin et al., 2001). A single ovule then provides the site for larval development and turns into a seed-sized gall (Kjellberg et al., 2005b). The larva feed on the endosperm (Deng et al., 2016) that developed from either double fertilisation or parthenogenesis (Borges and Kjellberg, 2014).

The wingless adult male fig wasps hatch first from their galls and search for galls containing females (Yang et al., 2002). The males then bite holes into the female galls and insert their genitalia in order to mate (Kjellberg et al., 2005b). Individual females may mate once or several times. Because sex ratios are skewed towards females, each male can mate with numerous females. Adult female fig wasps then emerge into the fig cavity and at this time the male fig flowers are mature and liberating pollen. The time from oviposition to adult emergence can take as little as one month (Harrison, 2005). The adult fig wasps have a short life span that rarely exceeds 48 hours (Harrison and Yamamura, 2003; Dunn et al., 2008). Within this limited time, they do not feed but need to find a receptive inflorescence of the correct fig tree species

(Harrison and Yamamura, 2003). Ahmed et al. (2009) found that a pollinator of an African fig tree *Ficus sycomorous* L. is able to fly up to 160 kilometres with the assistance of the wind.

Ficus species exhibit two breeding systems namely dioecy and monoecy. Roughly half of all fig species are monoecious, with individual inflorescences providing both female (seed production and dispersal) and male (pollen production and dispersal) reproductive functions (Herre et al., 2008). The remaining *Ficus* species are functionally dioecious. In these species, there are two types of trees; female trees that produce only seed-bearing figs; and male trees with figs that produce only pollen and pollinator wasp progeny to transport the pollen (Patel and Hossaert-McKey, 2000).

In a monoecious fig, both wasp larvae and seeds develop within the same figs. Pollinated ovules that do not receive an egg will develop into a seed (Bain et al., 2014). Monoecious fig trees contain many female flowers compared to male flowers with variable style length and each mature fig on these trees contains a mixture of pollen, wasps and seeds (Galil and Eisikowitch, 1968). Usually, in this type of breeding system, only 40-50% of the flowers will develop into viable seeds that signify a large portion of the fig tree's investment in female function (Kerdelhue and Rasplus, 1996) while the flowers that support the development of the pollen carrier female wasps represent the fig tree's investment in male reproductive function. In monoecious species of *Ficus*, the separation in time between male and female developmental phases (protogyny), both at the level of inflorescences and often the individual tree (because of within-tree fruiting synchrony), are much longer than in many other dichogamous plants (Hossaert-McKey and Bronstein, 2001). In monoecious species, normal pattern is that shorter styled flowers tend to support the development of offspring, longer styled flowers are more likely to be seeds (Nefdt and Compton, 1996). Male eggs are often laid first, and the shortest styled flowers tend to be the ones where eggs are laid first (Li et al., 2016).

In a dioecious fig tree, the female fig wasps enter the female figs and pollinate flowers which produce only seeds (Patel, 1996). The long styles of the flowers in the female figs and the structure of their stigmas prevent the wasps from laying their eggs but the insects continue to pollinate (Berg, 1989). In *Ficus*, the structure of the stigma varies significantly between sexes and taxonomic groups (Berg and Corner, 2005). The foundress can attempt to probe the styles but even if she succeeds to insert her

ovipositor the long styles of the flowers usually prevents the tip of the fig wasp ovipositor from reaching the fig ovule (Kjellberg et al., 2005b; Ghana et al., 2017). Entering the female fig has a fatal effect as no offspring are produced and the loss of her wings means that she cannot fly to another tree (Hossaert-McKey et al., 2016).

On male trees, the female flowers are modified for receiving a pollinator egg and only wasp larvae develop there. One to several months after pollination, the male fig wasp adult offspring emerge and mate with the gall-enclosed females (Yang et al., 2002). Due to protogyny, the female flowers become receptive simultaneously, several weeks before the male flowers mature (Patel, 1996). The short styled flowers in male figs are consumed by pollinators larvae (Ghazoul and Sheil, 2010). The persistence of dioecy in many species of figs is enigmatic because it represents an intense conflict between the interest of the pollinating wasps and those of the plant (Weiblen and Bush, 2002).

Most plants are pollinated passively, but the pollination in *Ficus* can be done actively or passively (Ahmed et al., 2009). The frequency of active pollination differs between lineages and consequently between regions (Kjellberg et al., 2001). Actively pollinated fig trees have pollinators with special behaviour and morphological adaptations (Kjellberg et al., 2014). Actively pollinated fig wasps are equipped with special sternal structure called pollen pockets that function in storing the pollen (Kjellberg et al., 2005b). The fig wasps actively collect pollen into their thoracic pollen pockets and later deposit it deliberately onto flower styles (Jousselin et al., 2003). They also engage in complex behaviour before leaving their natal figs, when they will search for the anthers and collect the pollen using their forelegs lined with fined coxal combs to aid the collection (Deng et al., 2016).

In the receptive figs, they remove the pollen from the pockets and transfer them to the stigmas while depositing their eggs (Kjellberg et al., 2014). The presence of these morphological structures and its associated behaviour results in efficient pollen transfer and reduces the amount of pollen needed, so actively-pollinated figs have a reduced number of anthers (Kjellberg et al., 2001). Passively pollinating fig wasps do not have either special morphological structures or complex pollination behaviour. Dehiscent anthers on their natal figs release pollen that becomes trapped on their bodies before they leave their natal figs (Waser and Ollerton, 2006). Subsequently, the pollen is brushed off onto the stigmas in the receptive figs, thus allowing pollination to occur.

1.1.5 Fig wasps

Figs act as brood sites for their pollinators - chalcid wasps from the family Agaonidae. They are the only routine pollinators of *Ficus* species and fig wasps are only able to reproduce within the figs they pollinate (Hossaert-McKey and Bronstein, 2001). The intriguing association between *Ficus* and these agaonid wasps has been known for a long time (Hill, 1967) but many species are still un-described (Quicke, 2012). Even at the start of this century when around 900 species of figs were known, only 350 pollinators had been described (Weiblen, 2002).

The body length of an adult fig wasp ranges from less than one millimetre up to two millimetres (van Noort, 2003). The fig wasp exhibit sexual dimorphism (Weiblen, 2002). Males differ from female fig wasps in that the wings are absent, they have reduced eyes, reduced middle legs and shortened antennae (Kjellberg et al., 2005b) (Figure 1.1). The male fig wasp has well developed mandibles and a telescopic gaster (Yang et al., 2002). Their morphology is well adapted to conditions in the lumen of the figs where they may spend their entire life (Compton and McLaren, 1989).

The female fig wasp has wings and their morphology is adapted to locate figs and to penetrate the ostiole. They bear a flattened head to help them enter the ostiole, mandibular appendages and short and strong spiny legs to push them through the ostiole (Liu et al., 2013). They also bear very sensitive antennae to locate receptive figs by smell, and the organs of smell (sensilla) are often elongate. The antennae break off during their entry into through the fig ostiole (Kjellberg et al., 2005b). Together with the loss of wings this helps the plant because it prevents the pollinators from flying to other trees. It may also make the wasps be less likely to get trapped in the ostiole.

The morphology of female fig wasps, especially their heads are correlated with the morphology of the ostiole length and fig wall thickness, further ensuring that pollination is very species specific (van Noort and Compton, 1996). These morphological traits act as the filter and help maintain the specificity of the relationship between fig wasp and its host species (Souto-Vilaros et al., 2018). The length of the ovipositor is also strongly correlated with the mean style length of the host flowers (van Noort, 2003).



Figure 1.1 Adult female and male fig wasps (*Blastophaga* sp. from *Ficus deltoidea* var. *angustifolia*).

1.1.6 Host specificity in Ficus

Among plants in general the specificity of the relationship is maintained before and after pollination. Pre-pollination barriers include the production and detection of cues produced by the host plant (Hossaert-McKey et al., 2010), morphological compatibilities and the behaviour of the pollinators (Sedeek et al., 2014). The post-pollination filters potentially include pollen competition, gametic mismatches, negative fitness and hybrid sterility (Coyne and Orr, 2004). In nursery pollination mutualisms, the specificity between plants and pollinators are often very high, with one plant species regularly pollinated by one or a few insect species (Rodriguez et al., 2017). The obligate mutualism between fig and fig wasps can be used as a model of evolution and speciation (Wei et al., 2014). Entering atypical hosts can lead to hybrids (Ghana et al., 2017). This hybridization may contribute to speciation and the diversification that has occurred in the long history of the association between fig trees and fig wasps (Kusumi et al., 2012).

In the fig and fig wasp mutualism, the most obvious filters that maintain the specificity between the obligate partners are the volatile organic compounds emitted from receptive figs (Proffit et al., 2009). In addition, surface features of the ostiole together with its size and shape (Nefdt and Compton, 1996), the size of the fig wasp (Liu et al., 2013), the compatibility of fig wasp ovipositor and the style length in the female flowers (Ghana et al., 2017) and the ability of the pollinators to induce galls (Ghana et al., 2015a) all contribute to the specificity of their interaction.

It used to be believed that only a single fig wasp species pollinated each fig tree species and vice versa (Deng et al., 2016) and many *Ficus* species still have only one recorded pollinator species (Herre et al., 1996; Machado et al., 2001). However, more sampling and molecular approaches have showed increasing numbers of fig trees with two or more pollinators (Kerdelhue et al., 2000; Machado et al., 2005; Su et al., 2008; Cornille et al., 2012; Rodriguez et al., 2017). In some cases fig wasps may also be shared across closely related fig species (Cook and Segar, 2010) but this seems rare.

A breakdown of specificity in the fig and fig wasp mutualism occurs when a fig wasp responds to the volatile chemicals emitted by the non-usual host receptive figs and successfully gains access in the fig cavity via the ostiole. It may pollinate and may be able to reproduce inside (van Noort, 2003). Generally, fig trees emit volatile compounds that attract only their pollinators but sometimes fig wasps do enter figs
produced by species other than its regular host (Cook and Segar, 2010; Souto-Vilaros et al., 2018). There are a few cases where closely related *Ficus* species share the same pollinators. This might occur due to host shifts between sister species (Cook and Segar, 2010). Closely related *Ficus* may produce rather similar volatile compounds when receptive and some fig wasps may not be able to differentiate between the volatiles (Cornille et al., 2012) thus making gene flow between taxa possible. A study by (Wang et al., 2016) found that sharing of pollinators within the same section of *Sycomourus* was five times greater compared to section Hemicardia. A study by (Ware and Compton, 1992) also showed the successful reproduction of pollinators in figs of *Ficus lutea* Vahl. (subgenus *Urostigma*) that normally reproduce in *Ficus thonningii* Blume (also subgenus *Urostigma*) but that pollinators of *Ficus sur* Forssk. (subgenus Sycomourus) that entered the figs did not reproduce. Even so, gene flow between distantly related fig tree species is possible. Distantly related fig species (*Ficus auriculata* Lour. and *Ficus tsiangil* Merr. ex Corner) in subgenus *Sycidium* occasionally share pollinators (Wang et al., 2016).

Exceptions to the strict species-specific relationship in the fig tree and fig wasp mutualism occurs among fig trees with both breeding systems (Moe, 2011), though there may be with more cases where breakdown occurs in monoecious figs (Cook and Segar, 2010; Moe et al., 2011). Yang et al. (2015) examined the cause of pollinator sharing in monoecious species and concluded that it is due to host shifting and duplication (where the two pollinators of a fig tree are sister species) whereas the presence co-pollinators in dioecious figs was due to only duplication. Yang et al. (2015) also proposed that the higher breakdown of specificity in monoecious figs was due to the cost for entering the wrong host was more serious in dioecious figs. Since the life span of the pollinator fig wasp is really short (Compton, 1993), they have such a short time to live that they cannot afford to be too choosy and rushing to enter the first receptive fig they found (Ashman, 2009). The dispersal ability of the pollinators might also cause the monoecious figs tree to have more breakdowns. A study by Borges (2016) discovered that the pollinators of dioecious fig trees tend to have shorter ranges of dispersal compared to the long range dispersal by pollinators in monoecious fig trees. Being able to fly further may increase the chance of a pollinator being attracted to another host apart from their normal host. All of these reasons make pollinator sharing in dioecious figs less likely (Kobmoo et al., 2010).

It is clear that a single *Ficus* species with more than one pollinator is not very rare. Multiple pollinators usually do not overlap in distribution (are parapatric). Surveys of pollinators therefore need to extend across the distribution of the host trees to establish how many pollinator species are present on that tree. Examples where two or more *Ficus* species routinely share the same pollinator appear to be much rarer, but confirmation of pollinator specificity is difficult because it requires surveys across a range of fig tree species, with each fig tree sampled in multiple locations. Two Ficus taxa with two different pollinators throughout their ranges are likely to represent two distinct biological species, because pollinator behaviour determines plant gene flow. Plant species can be subdivided into varieties and sub-varieties on the basis of morphological or other differences (Hahnke et al., 2016). If these varieties consistently have different pollinators, then pollen flow between them will be absent, and they represent distinct biological entities. Conversely, varieties that share the same pollinators can inter-breed, so long as other barriers such as their geographical locations do not prevent this. One *Ficus* species that has a lot of described varieties is Ficus deltoidea. Their ecology is also poorly understood, and the specificity or otherwise of the relationship between different varieties and their fig wasp pollinators is unknown.

1.1.7 Ficus deltoidea

Ficus deltoidea Jack also known as the mistletoe fig, is one of the more frequently found fig trees in South East Asia. Its natural distribution includes Thailand, Indonesia and Malaysia (Lansky and Paavilainen, 2011). Some varieties are also commonly grown as ornamentals, usually as pot plants. They are locally known as "Mas Cotek" by Malaysians because of having gold (in Malay gold means emas) coloured fine spots on the surface of the leaves (Mat et al., 2012). They also are known as sempit-sempit in Sabah, Sarawak and Kalimantan and tabat barito in Indonesia (Desaku, 2005).

Traditionally, *F. deltoidea* is widely used to cure a variety of diseases such as rheumatism, diabetes, toothache, headache, cold, and to cure sore throats using the powdered roots and leaves (Bunawan et al., 2014). More recently it has been shown that *F. deltoidea* has anti-hyperglycaemic (Adam et al., 2012), anti-diabetic (Sirisha et al., 2010), antioxidant (Hakiman and Maziah, 2009) and wound-healing properties (Abdulla et al., 2010). Adam et al. (2007) stated that all parts of *F. deltoidea* have potential as medicinal properties that include the roots, barks, leaves and fruit. *F. deltoidea* is a well-known house plant, and is also significant as a source of traditional pharmaceutical products (Bunawan et al., 2014). Their leaf also has been used commercially to make a herbal tea (personal observation).

F. deltoidea commonly occurs as a true epiphyte or terrestrial bush in coastal areas, heath land and mountainous sites below 3200 m altitude. Plants usually range between 0.3 m to 7.0 m in height and are often found as a shrub rather than a tree (Starr et al., 2003). *F. deltoidea* produces picturesque aerial roots under warm and humid conditions (Riffle, 1998). Species in the *F. deltoidea* group can be epiphytic, epilithic (germinating on rocks as 'rock-splitters') or terrestrial shrubs and treelets. As terrestrial plants they are often found on nutrient-poor soils (Berg and Corner, 2005). These authors linked the slow growth of *F. deltoidea* and its production of figs sequentially and asynchronously to their occurrence in nutrient poor habitats. Traits developed in response to poor soils, including the plants exceptionally large seeds may represent a pre-adaptation that has allowed them to survive as epiphytes, rather than as hemiepiphytes (stranglers).

F. deltoidea is dioecious, with separate male and female plants that either support fig wasp pollinators or produce seeds, respectively. *F. deltoidea* exhibits several characteristic that differ from many other *Ficus* such as its large seeds, small number of flowers and different juvenile and adult leaf forms (Corner, 1969). Female figs of this group are exceptional insofar as they only contain small numbers of flowers. This allows individual seeds to be far larger than is normal for fig trees (Kraft et al., 2015a) and is a presumed adaptation for an epiphytic life style. Birds and Orang Utans are the only confirmed feeders on the figs (Shanahan et al., 2001).

Whether the varieties of *F. deltoidea* are actually distinct species is unclear. Morphological studies of *F. deltoidea* show that all characters portray a high variability among the varieties (Awang et al., 2013). This variation between and within species may be due to cross-pollination and sexual recombination between the species (Corner, 1997). Genetic studies showed the variation between the varieties in *F. deltoidea* reflected morphological variation rather than the geographical origin of the plants (Zimisuhara et al., 2015).

Leaf morphology and anatomy are often used for identification of *F. deltoidea* varities because they are particularly discriminative (Nur Fatihah et al., 2014). However, F *deltoidea* often exhibit different states of leaf characters between the young and mature plants that has often led to misleading identification of the varieties. The leaf forms of *F. deltoidea* often exhibit great variation, from lanceolate, elliptic, obavate, obdeltoid, spathulate or show a combination of two or three of such forms (Awang et

al., 2013). The leaf lengths are between 4 cm and 8 cm. They are bright-green coloured on the upper surface and rust-red to olive-brown on the bottom surface of the leaf (Mat et al., 2012). The figs vary from 6 mm to 18 mm long and 5 mm to 22 mm wide with yellow, orange, red or purple in colour when ripe (Awang et al., 2013). The shape of the figs can be either globose, oblong, fusiform, conical or obconical (Laman and Weiblen, 1998).

Corner (1969) classified 13 varieties in *F. deltoidea*, with seven of them found in Peninsular Malaysia. Subsequent treatments have been based on this work, and the taxonomy of the group has remained essentially unchanged since 1969, except that many of his varieties were merged by Berg and Corner (2005). This study was concordant with a study by Mat et al. (2012) and Nur Fatihah et al. (2014) which found those seven varieties were native varieties in peninsular Malaysia. The varieties are var. *angustifolia* (Miq.), var. *deltoidea* Corner, var. *trengganuensis* Corner, var. *kunstleri* King, var. *bilobata* Corner, var. *motleyana* (Miq.) and var. *intermedia* Corner.

Information regarding the pollinators of *F. deltoidea* is very scarce. The only pollinator recorded for *F. deltoidea* is *Blastophaga quadrupes* Mayr, which probably pollinates var. *lutescens* (Corner, 1969) as it was collected in Java and Sumatra (Wiebes, 1993). The common characteristics in the genus *Blastophaga* is the absence of coxal and sternal corbiculae (Ramirez and Malavasi, 1997). Female fig wasp from genus *Blastophaga* does not have pollen pocket that makes them passively pollinate the host fig tree (Wiebes, 1994). Wiebes (1993) found that there was some variation in the females of *Blastophaga quadrupes* among different varieties of *F. deltoidea*, but the male seems uniform. According to Weibes, (1994) *Blastophaga quadrupes* pollinate several varieties of *Ficus deltoidea*. However, Corner (1969), based on floral and inflorescence differences, suggested that different pollinators were likely, and different varieties of *F. deltoidea* in Brunei have been found to support distinct species of pollinator (F. Kjellberg, personal comm. to S. G. Compton). If this is the case elsewhere, then *F. deltoidea* may represent a complex of closely-related but biologically distinct species.

A study by Zimisuhara et al. (2015) showed the gene flow between varieties is occurring in 30 accessions of *F. deltoidea* from five varieties (var. *angustifolia*, var. *bilobata*, var. *deltoidea*, var *kunstleri* and var. *trengganuensis*) collected in Peninsular Malaysia but also that there was a significance difference in eight morphological

characteristics between the varieties. This result supports (Corner, 1965) who could not differentiate all the varieties and placed them in a single species. Tsukaya (2005) concluded that many plants have different type of leaves within the same species (and could be regarded as varieties) due to having adapted to different environmental conditions such as light intensity. However, Tnah et al. (2016) found four sympatric varieties of *Ficus deltoidea* living in the same environmental conditions, and the molecular results detected great inter-varietal variation. The four varieties of *Ficus deltoidea* were divided into two groups, namely a small leaves group that consisted of var. *angustifolia* and var. *deltoidea* and a large leaf group that consisted of var. *kunstleri* and var. *lutescens*. A similar result was found by Zimisuhara et al. (2015) who separated var. *angustifolia*, var. *deltoidea* and var. *bilobata* as one group while var. *kunstleri* and var. *trengganuensis* were placed as another group. Based on their morphological characters, all the varieties in the first group have small leaves while the second group have big leaves. Varieties in both groups can grow as facultative epiphytes.

1.1.8 Epiphytes and Hemiepiphytes (stranglers)

Epiphytes are non-parasitic plants that live on the external surface of living plants where they take nothing from the host except a safe anchorage (Thomas and Packham, 2007). They can be vascular plants (pteridophytes, gymnosperms and angiosperms) or non-vascular plants like algae, lichens or bryophytes (Nadkarni et al., 2001). Epiphytes exploit sunlight by using other plants without giving any harm to the host tree (Zotz, 2013). Epiphytes obtain nutrients and water from the air, fog, water present in the humus and compost that lies on tree branches (Petruzzello, 2017). Despite not taking nutrients from hosts, the epiphytes may compete for sunlight with the host tree. Epiphytes have the ability to maintain positive carbon balance and retain their leaves especially during dry season (Putz et al., 1995). Epiphytes are true epiphytes that spend their entire life without contact with the ground (Benzing, 1990).

A hemiepiphyte is a plants that spend at least a part of their life cycle as epiphytes on their host and the other part rooted to the ground (Williams-Linera and Lawton, 1995). Hemiepiphytes can be divided into two categories which are primary hemiepiphytes or secondary hemiepiphytes. Primary hemiepiphytes start their life as epiphytes before spreading their roots to the ground while secondary hemiepiphytes germinate in the soil and grow up the sides of large trees utilizing adhesive roots and finally cut their

connection with ground (Nadkarni et al., 2001). In primary hemiepiphytes, their earlier epiphytic stage traits resemble the true epiphytes (Zotz and Winter, 1994).

Hemiepiphytes can be found abundantly in tropical forests (Hao et al., 2016). There are several factors that contribute to the distribution of hemi-epiphytes in the canopy. These incude their seed dispersers, the growth of hosts after formation of hemi-epiphytes, and host microsite quality (Harrison and Yamamura, 2003). Hemiepiphytes are very abundant in lowland tropical rainforests, and in lower montane and mid-montane cloud forests (Putz and Holbrook, 1986; Williams-Linera and Lawton, 1995). They play an important role in canopy dynamics by competing with their host tree species, and stabilize mats of epiphytic organic soil that affect rainfall interception and ecosystem hydrology (Veneklaas et al., 1990; Williams-Linera and Lawton, 1995). The flexibility of their physiology is important as the drastic changes between the two phases in hemiepihytes life cycle include radical changes in their rooting environments (Hao et al., 2010). When hemiepiphytes reach the ground they encounter an intense change in rooting volume and characteristics of the rooting zone (Holbrook and Putz, 1996b).

Two important hemiepiphytic genera are *Ficus* (Moraceae) and *Clusia* (Clusiaceae) (Hao et al., 2016). The physiology and ecology of these plants, is poorly known despite approximately 500 *Ficus* species recorded as being hemi-epiphytes (Putz and Holbrook, 1986). Epiphytic and hemiephytic *Ficus* species often colonise palm trees. The morphology of the erect trunk with a surface protected by persistent leaf bases makes them as an ideal habitat (Nadarajah and Nawawi, 1993). This includes planted palms and palms in plantations, such as oil palms. Some varieties of *F. deltoidea* are among the epiphytes that routinely take advantage of the trees growing in oil palm plantations.

1.1.9 Oil palm

Elaeis guineensis (Arecaceae), commonly known as oil palm, is planted mainly for their oil. This species originated from Africa where they live naturally in Sierra Leone, Liberia, Ivory Coast, Ghana, Cameroon and Republic of Congo and Zaire (Prokurat, 2013). In Soth East Asia, it was first planted in Sumatra in 1911 before spreading to Malaysia in 1917 (Kushairi et al., 2017). Today, Malaysia has become the second largest producer of oil palm after Indonesia (Dayang Norwana et al., 2011). Oil palm is planted frequently in Malaysia due to its climate suitability which received adequate

rainfall, sufficient light and its good soil conditions that ideal for the growth of the oil palm (Kushairi et al., 2017). The main contributor to Malaysian gross national profit comes from oil palm production with 16.8 billion USD and this industry provides more than 600 000 working opportunities (Oettli et al., 2018).

Oil palm cultivation is expanding due to rising global demand for palm oil and this trend is forecast to increase until 2050 where approximately 109 hectares of natural habitats will be transformed into agriculture (Tilman et al., 2001). Animal taxa are known to decrease in species diversity and species richness due to conversion of natural forest to oil palm (Foster et al., 2011). The fast-growing conversion of natural forest into agricultural land has urged Malaysian government to diversify the mitigation and maintains a biodiversity programme through the establishment of Roundtable on Sustainable Palm Oil (RSPO) that includes big-range cultivators to small-range farmers (Kushairi et al., 2017). A statistic from (Lee, 2011) showed 60% of overall oil palm plantations in Malaysia are managed by big companies, 28% from smallholders and 12% from independent smallholders.

Until now, various methods were proposed by RSPO to implement environmentalfriendly practices and thus mitigate the effect of conversion of natural forest to oil palm plantation. These include planting beneficial plants alongside the plantations to serve as food for beneficial insects, maintaining above-ground weeds that provide nectar for pollinators and herbivores (Siti Khairiyah et al., 2012). Maintaining epiphyte communities on the palm trunks is another option as all taxa showed a significant decrease due to conversion of natural habitat to oil palm plantation (Foster et al., 2011). Most published studies are on the impact of conversion of the forest into oil palm plantations on mammals, birds and reptiles and insects, but the role of epiphytes is rarely considered. The ecology of oil palm epiphytes together with their contribution to the insect populations in oil palm little understood (Suzanti et al., 2016).

An oil palm plantation that is 'well maintained' is a virtual monoculture crop. Palms nonetheless offer opportunities for colonisation by epiphytes, even if they are not like a natural forest and epiphytes can be abundant if they are not cleared by plantation workers (Suzanti et al., 2016). Epiphyte communities in oil palm plantations include lichens, algae, mosses, ferns and angiosperms (Nadarajah and Nawawi, 1993). The epiphyte community serves as buffer to species loss caused by the conversion of natural habitat to the plantation (Fayle et al., 2008). In Malaysia, epiphytes can be

found more abundantly in small plantations managed by independent smallholders and farmers (personal observations). These smallholders have more autonomy to decide their own practices and how to manage them. They are not linked to any bodies and receive limited funds, have low technical expertise and lack knowledge on best practices and new technologies (Nagiah and Azmi, 2012). The 'poor' management by these small holders often lets the epiphytes remain on the trunks rather than cutting them or spraying them. This is in contrast to a big company that runs big plantations, where the management tend to clear out everything as they think the epiphytes might give negative effects to yields, or they have staff at certain times of year with nothing else to do. The intense management by big companies aims to remove competing plant species that compete with crops for water, light and nutrients. These competing vegetation includes herbaceous understory and epiphytes that live on the trunks of the oil palm (Prescott et al., 2015).

In contrast to currently common assumptions by oil palm companies, maintaining the epiphytes does not have any significant effects on profits and at the same time reduces the biodiversity losses (Zotz, 2016). A study conducted by Prescott et. al. (2015) in three oil palm estates in Sabah, Malaysia found 58 species of epiphytes present comprised of ferns and angiosperms and also mosses. These epiphytes serve as microhabitats for smaller organisms such as ants and other insects (Fayle et al., 2008). A study by Suzanti et al., (2016) showed that plantations that maintain their epiphytes recorded a greater number and diversity of insects compared to plantations that had removed the epiphytes completely or reduced them by half. Another study by (Koh, 2008) found that an increase of species richness of 1.5 birds was correlated with the epiphytes in Malaysian Borneo oil palm plantations. These epiphytes and their associated insects act as a source of food for the birds. A study has concluded that oil palm plantations can support more bird species than other non-forest land-uses such as soy bean plantations (Lees et al., 2015). These birds not only eat the fruits of the angiosperm epiphytes but also act as seed dispersers for the angiosperms. Apart from birds, animals found in oil palm plantations might also act as seed dispersers include squirrels and monkeys.

Oil palm matters for *F. deltoidea* as it can be suitable habitat for this significant true epiphyte that does not give any harm to the oil palm trees. This fig tree species later can provide a continuous food source for frugivores. *F. deltoidea* also has an economic value, so having more of this species is valuable, and oil palms can support high densities of this fig tree.

1.2 Research aims and objectives

The research described in this thesis aims to explore the biology of epiphytic *Ficus deltoidea* and its pollinator in selected oil palm plantations in Peninsular Malaysia. The general *Ficus* community present as epiphytes were also studied. The specificity of pollination of varieties of *F. deltoidea* is unknown. This is the first study to the determine the biology and the host specificity of the pollinators for several varieties in *F. deltoidea* in peninsular Malaysia. The phenology of epiphytic *F. deltoidea* in oil palm plantations was also studied. The objectives of this research were to:

- i. Assess the epiphyte communities in certain oil palm plantations in Peninsular Malaysia and to determine which *Ficus* species are most abundant and whether they occupy different heights on the oil palm trunks.
- ii. Characterise the morphological characteristic of fig from seven native varieties of *F. deltoidea* and their *Blastophaga* sp. fig wasps from three different varieties.
- iii. Determine the host specificity of the Blastophaga sp. Pollinator from var. angustifolia
- iv. Describe the leafing and fruiting phenology of epiphytic *F. deltoidea* var. *angustifolia* in oil palm and the pollination limitation.
- v. Describe the pollination biology of *Blastophaga* sp. that pollinates *F. deltoidea* var. *angustifolia*.

1.3 Thesis outline

The contents of this thesis are organised as follows: Chapter 1 provides the background of this research and describes the research aims and objectives. Chapter 2 describes general methods and equipment and where the studies were conducted. This includes climate data, the age of plantations, what type of soil they were planted on and the management of each plantation. The general methods chapter also describes the fig developmental phases used in several chapters in this study. Further details of methods are elaborated in each chapter where appropriate (Chapters 3-7). Chapter 3 discusses the epiphytes present in five oil palm plantations in Peninsular Malaysia and describes differences in their height preferences. Host preferences of two different varieties of *Ficus deltoidea* that live sympatrically are also described. Chapter 4 provides characterisation of male and female figs of all seven native varieties of *F. deltoidea* in Peninsular Malaysia., The morphologies of three

Blastophaga sp. pollinators from three varieties are also compared. Chapter 5 describes experiments aiming to determine the host specificity of the *Blastophaga* sp. that pollinates *F. deltoidea* var. *angustifolia*. An apparent preference for male over female figs is also described. Chapter 6 describes the phenology of epiphytic *F. deltoidea* var. *angustifolia* in an oil palm plantation near Kuala Lumpur. Chapter 7 describes the pollination biology of this variety of *F. deltoidea*. Chapter 8 provides a general discussion and includes suggestions for future work.

Chapter 2

General methodology

2.1 Study sites

The study was conducted from June 2016 until August 2017 at five oil palm plantations in Peninsular Malaysia (Figure 2.1). The main study site was located in an oil palm plantation in Banting. The other plantations were at Dengkil, Batu Pahat, Bagan Serai and Tembila. Four of the plantations are managed by small-sale farmers, while Tembila is owned by a private company. Further studies were carried out in a *Ficus deltoidea* germplasm collection owned by Universiti Sultan Zainal Abidin (UnisZA) and inside the university area of UniSZA located in Besut, Terengganu.

The study sites were chosen because they had *F. deltoidea* growing as epiphytes on oil palm trunks. Before the selection of study sites, a number of other sites were visited to check for the presence of this species. Other criteria included personal safety, permission from the owner to conduct the study, the numbers of *F. deltoidea* at the site and travel distances (for the main study site).

2.1.1 Banting oil palm plantation

This 2.10 hectares plantation was the main study site. It is located in Selangor, about 50 km from Kuala Lumpur city centre. The coordinates of this plantation are 2°50.094" N and 101°35.074" E. It is managed by small-scale farmer who has lived on the site since the palms were planted in 2001. The plantation has a peat soil. There are about 285 oil palm trees in the plantation and 86 of the trunks serve as hosts for 113 epiphytic individuals of *F. deltoidea* var. *angustifolia* (Chapter 3). These include some big male trees of var. *angustifolia* that are convenient for experimental purposes (Chapter 5). The phenology of the fig trees is described in Chapter 6.

Climate data for Banting and the other sites was obtained from the Malaysia Meteorological Department. The closest station to Banting is the KLIA Sepang station located about 19 kilometres away. Banting has a tropical climate with an average annual temperature of 27.8 degrees celsius that varies little throughout the year. Along the sampling duration (June 2016 until August 2017), the average annual precipitation throughout the year was 144.3 mm. The precipitation ranged between 30.2 mm and 345.8 mm. The average daily minimum and maximum temperatures were relatively stable (Figure 2.2). The average daily minimum temperatures ranged from 24.0 to 25.2 °C, while average daily maximum temperatures were between 30.9 and 32.5 °C. Rainfall was lowest during August 2017 at only 26.5 mm the rainfall was the second lowest in the same month a year before with 64.8 mm. The rainfall recoded In April 2017 was the highest at 236.0 mm.



Figure 2.1 The six study sites in Peninsular Malaysia.



Figure 2.2 Climate data for Banting oil palm plantation, Selangor. Monthly percentage of rainfall (open bars), relative humidity (\blacksquare) and monthly average temperatures (•) for June 2016-August 2017 as measured by KLIA Sepang (19 km from the research site).

2.1.2 Dengkil oil palm plantation

The Dengkil, plantation is also in Selangor, about 10 km away from Banting, but has a silt soil. The coordinates for this plantation are 2°51.125" N and 101°39.424" E. This 40.53 -acre plantation is managed by small farmers and the oil palms were planted in 1995. There are about 427 oil palms in this plantation, which host numerous epiphytes (Chapter 3). KLIA Sepang was again the nearest meteorological station to Dengkil.

2.1.3 Batu Pahat oil palm plantation

Batu Pahat is located in Johor state, the southern part of Peninsular Malaysia, about 239 km from Kuala Lumpur. Its coordinates are 5°07.070" N and 100°39.445" E. This plantation is about 2.50 hectares and the palm trees were planted in 1991. This peat soil plantation is a private residence of a F. deltoidea trader, Mr Ismail. He grows examples from the F. deltoidea complex including var. angustifolia, var. deltoidea, var. kunstleri and var. trengganuensis and sells their leaves for research and pharmaceutical purposes. Apart from the planted trees, two varieties of F. deltoidea namely var. angustifolia and var. deltoidea grow as epiphytes in the old oil palm plantation within the residential area. Both varieties show evidence of being pollinated and can be found in abundance on the trunks. The planted examples of var. angustifolia and var. deltoidea were made from cuttings taken from the wild epiphytes. Plants from the other two varieties were brought from KESEDAR in Kelantan in 2006 and planted at Batu Pahat. The climate data for this plantation were taken from Batu Pahat station, 24.7 km from the plantation. The range of temperatures during the period of this study (June 2016 until August 2017) were stable and ranged from 27.0°C to 28.9°C while the rainfall ranged from 82.2 mm to 292.2 mm (Figure 2.3).

2.1.4 Bagan Serai oil palm plantation

Bagan Serai, is located on peat soil in Perak state about 305 km north of Kuala Lumpur. The coordinates for this plantation are 5°07.070" N and 100°39.445" E. This old plantation is about 0.9 hectares in area and the palms were planted in 1989. The trees are hosts for epiphytic *F. deltoidea* var. *angustifolia*. Climate data were obtained from Chersonese Est. Kuala Kurau, about 36.7 km away. The mean temperatures were stable within the range of 26.4 to 28.2°C (Figure 2.4). The highest rainfall during the study period was in September 2016 with 428.4 mm, while the driest period was July 2017 when only 20.1 mm of rainfall recorded.



Figure 2.3 Climate data for Batu Pahat oil palm plantation, Johor. Monthly mean rainfall (open bars) and average temperature (■) for June 2016-August 2017 as measured by Batu Pahat station.



Figure 2.4 Climate data for Bagan Serai oil palm plantation, Johor. Monthly mean rainfall (open bars) and average temperature (■) for June 2016-August 2017 as measured by Chersonese Est. Kuala Kurau station.

2.1.5 Tembila oil palm plantation

Tembila oil palm plantation is located on sandy soil close to the coast in Tembila, Terengganu, a south eastern coastal state. The coordinates for the plantation are 5°43.402" N and 102°39.445" E. It is managed by a company not small farmers. The overall plantation is about 129.5 hectares and the oil palms were planted in 2007. The climate data for rainfall and temperatures were taken from Institut Pertanian Besut station about 15 km away. Temperatures between June 2016 until August 2017 ranged from 25.0 to 28.8 °C (Figure 2.5). The driest month was April 2017 when only 10.9 mm rainfall was recorded while the wettest month was December 2016 which received a total of 632.5 mm rainfall. December is known locally as the month when floods are likely in Terengganu.

2.1.6 Terengganu germplasm collection

The germplasm collection is owned by Universiti Sultan Zainal Abidin (UniSZA) located in Terengganu. The collection includes varieties of *F. deltoidea* from Peninsular Malaysia together with others from Borneo. This collection was established in 2006 and contains 165 medium-large individuals and many smaller plants. It contains eight varieties: var. *angustifolia*, var. *deltoidea*, var. *trengganuensis*, var. *kunstleri*, var. *bilobata*, var. *motleyana*, var. *intermedia* and var. *borneensis*. Of these varieties, only var. *borneensis* is not native in Peninsular Malaysia namely Johor, Kelantan, Melaka and Terengganu. The var. *borneensis* were received from the Sarawak Biodiversity Centre. The cuttings vary from one to 10 years old. The plants are grown in clay pots with either a 39 cm or 49 cm diameter. The media was a combination of top soil, pome, rice husk and peat soil. All of the collection were given NPK Green organic fertilizer and were watered regularly.

The university area of UniSZA is about one kilometre from the germplasm collection. The nearest available weather data is from the Institut Pertanian Besut station, as given for the Tembila plantation.



Figure 2.5 Climate data for Tembila oil palm plantation, Johor. Monthly mean rainfall (open bars) and average temperature (■) for June 2016-August 2017 as measured by Institut Pertanian Besut station.

2.2 Fig development phases

An assessment was made to look for possible ways in which the developmental phases of *F. deltoidea* var. *angustifolia* could be assessed without removing the figs. This was necessary because relatively small crop sizes and asynchronous fruiting made extensive destructive sampling problematic. Initially, about 30 figs from male and female figs were collected from the Banting plantation. The figs covered the whole range of sizes. After that, the figs were examined individually and differences in their size, thickness, colours and contents were recorded. Based on these results a visual assessment of fig developmental stages was possible and used during phenological studies and to obtain suitable figs for experiments.

The assignments to fig developmental phases were based on the scheme of Galil and Eisikowitch (1968) where there are five phases in the fig developmental cycle. During the A phase in male figs, the ostiole is still closed and the size is about 0.2 cm (Figure 2.6). During this phase, the red tepals are about the same size as the female flowers. The styles of the female flowers are white in colour. Up to B phase, the female flowers, tepals and male flowers are growing. In terms of diameter, both phases are soft and green in colour and around 0.4 cm. Both phases were squeezable.

C phase male figs usually look yellowish green from the outside and the texture is hard Figure 2.6). During this phase, squeezing the figs is not possible. This is because fig wasp offspring development has started and the galled ovules have expanded. During this phase, the tepals change colour from red to brown and stop growing. At D phase, the next generation of fig wasp offspring emerge from their galls. Usually this phase has a softish feel when squeezed, though no clearly defined central cavity develops. The colour at D phase is usually yellow green. The anthers of the male flowers start to dehisce. The size of D phase figs varies but is usually between 0. 45cm to 0.55 cm.

At E phase the figs are attractive to frugivores that can become seed dispersers. The texture of this phase is also softish with a yellowish green to mild red colour. In this phase, the pollinator exit hole can be seen through the ostiole and the anthers are dehiscing. The tepals are brown colour and shrunken. The fig diameter is usually more than 0.56 cm.

An additional 'phase' was recognised, to record figs that had failed to develop because they had not been entered by pollinators. This condition is not included in the Galil and Eisikowitch (1968) scheme. In aborting figs the tepals are brown and the size is variable but usually does not exceed 0.6 cm. Their colour is often yellow to orange and if squeezed, a hollow texture can be felt.

The structure of female figs is simpler than that of male figs because they only contain female flowers and their associated tepals. The female figs also display fewer phases as compared to male phase, because D phase is absent (Figure 2.7). The size of figs on female trees is often bigger than for male figs at the same developmental stage. As in male figs, phase A is immature figs where the ostiole is still closed. Usually in this phase, the size is between 0.1 cm - 0.3 cm and the texture is soft with a green colour. The tepals appear more rounded in female than male figs. The mean style length is also much longer in female figs.

In B phase, the tepals and seed are growing and usually the tepals are bigger than the ovules. The texture is still soft, with green colour and the size reaches about 0.4 cm. Meanwhile, in C phase, there is not so much difference in colour, texture and size between female and male figs. The texture is hard, yellow green and within a range of 0.5 to 0.6 cm. In this phase, the seeds grow and the colour of the seed change from green to yellow, while the size of the tepals remains the same.

At E phase, the texture of the seed is soft with a slight red colour and the figs are ready to be eaten by frugivorous. The colour of the tepal has faded from bright red to mild red. The female figs lack a D phase because no pollinator offspring develop in the figs. The ripe are often more than 0.7 cm in diameter.

Aborting female figs develop when no pollinator enters the figs. The ovules and tepals stop growing and turn yellow. The style colour becomes brown. The colour is always yellow and their hollow structure can be felt when squeezed. As in male figs, they remain for extended periods on the trees before falling to the ground.



Figure 2.6 Fig developmental stages in male *F. deltoidea* var. *angustifolia*. (A) Young fig. (B) The fig is receptive to pollinators. (C) Development of fig wasp offspring. (D) The next generation of wasps are released. (E) After the wasps have emerged. (F) An aborting, un-entered fig.



Figure 2.7 Fig developmental stages on female *F. deltoidea* var. *angustifolia* fig trees. (A) Young fig. (B) The fig is receptive to pollinators. (C) Development of the seed. (D) Late Development of the seed. (E) Mature seeds ready to be dispersed (F) Aborting fig.

Chapter 3

Epiphytes in Malaysian oil palm plantations

3.1 Introduction

3.1.1 Epiphytes

Epiphytes are a group of autotrophic organisms that live on the external surfaces of other plants without causing any harm to the host plants. They are widely distributed, especially is widely tropical and subtropical areas (Thomas and Pachams, 2007; Zotz, 2016). Epiphytes most commonly inhabit humid climates where they receive less nutrients, more light, less water and larger daily temperature amplitudes than terrestrial plants (Benzing, 1990; Holbrook and Putz, 1996b). Epiphytes are significant for primary production and help retain water and nutrients (Veneklaas et al., 1990). Communities of epiphytes increase the canopy complexity and provide important resources for other organisms especially insects (Suzanti et al., 2016). Individual trees can have more than 80% epiphyte cover (Kreft et al., 2004) so they are therefore important for maintaining biodiversity in tropical areas (Ellwood and Foster, 2004).

Epiphytes include vascular and non-vascular plants. Vascular groups consist of ferns (pteridophytes) and higher plants (gymnosperms and angiosperms). A recent survey by Zotz (2013) recorded 27 614 species of vascular epiphytes from 913 genera and 73 families. Vascular epiphytes are usually associated with lower latitudes (Nadkarni et al., 2001) and are favoured by a continuous water supply from rainfall and deep organic layers on their hosts (Hemp, 2006). Non-vascular epiphytes like algae, lichens and bryophytes are also found abundantly in the tropics, where they prefer to inhabit the upper canopy (Nadkarni et al., 2001). Compared to vascular epiphytes, this group of epiphytes remains understudied due to their small size (Affeld et al., 2008).

About 2 700 species of ferns and their allies, in 121 genera, are known to develop as epiphytes (Zotz, 2013). Assessments in five oil palm plantations in North Sumatra found three fern species *Nephrolepis biserrata* (Sw.) Schott (Nephrolepidaceae), *Goniophlebium verrucosum* (Wall. ex Hook.) (Polypodiaceae), *Vittaria elongate* Sw. (Pteridaceae) and one angiosperm species *Asystasia gangetica* L. (Acanthaceae) were the dominant epiphytes (Suzanti et al., 2016). Similarly, in Malaysian oil palm plantations, Nadarajah and Nawawi (1993) recorded that ferns were the dominant group of epiphytes.

General characterisations of epiphytes often include having small leaves, slow growth and large seeds (Hao et al., 2016). Being detached from the soil causes the epiphytes to experience a microclimate different from ground-living rooted plants (Zotz, 2016). The morphology of epiphytes is adapted to extreme conditions, especially water and nutrient scarcity (Nadkarni et al., 2001). Living in a water limited environment is the biggest challenge faced by epiphytes and this problem is more critical in seasonally dry climates (Benzing, 1990). Having small leaves with a higher leaf mass per unit area, and a thicker epidermis helps them to reduce water loss (Hao et al., 2010). They also have the ability to retain carbon in the leaves during times of water scarcity (Putz et al., 1995).

Seed size reflects a compromise between the number of seeds produced and the investment in each seed (Westoby et al., 1992). The seeds of some epiphytes are large compared to related taxa. For example Rockwood (1985) found that Neotropical epiphytes had seeds of similar weight or averaged slightly larger than their relatives with other growth forms. Larger seeds can provide more nutrients in the early stage of epiphyte establishment and may be important due to the slow growth typically shown by epiphytes (Kraft et al., 2015b). Some epiphytic plants like orchids nonetheless have tiny wind-blown seeds while others have larger seeds structurally adapted for wind dispersion or depend on animals for dispersal (Zotz, 2016). This suggests that different taxonomic groups adapt differently to the selection generated by an epiphytic life style (Arditti and Ghani, 2013).

Host plant characteristics influence the epiphyte communities growing on them. Bark traits like thickness, rugosity, shedding rate, density and water holding capacity are all significant (Wyse and Burns, 2011; Wagner et al., 2015). Bark shedding prevents colonisation of tree trunks (Steward and Beveridge, 2010). According to Mondragon

and Calvo-Irabien (2006), the germination of epiphyte seeds does not depend on tree height and the age of the host tree. However, Zotz and Vollrath (2003) note that the age of a tree does affect the abundance of epiphytes it supports, because older trees have a larger surface area to support epiphytes and as the age of the tree increases so does the chance of repeated colonisation.

Epiphytes can be hemiepiphytes or non-hemiepiphytes (true epiphytes). Hemiepiphytes are plants that spend parts of their life as epiphytes whereas non-hemiepiphytes spend their whole life as an epiphyte (Quaresma et al., 2017). Hemiepiphytes can be primary hemiepiphytes where they start their life by being epiphytes on a tree before they develop aerial roots reaching the ground (Putz and Holbrook 1986; Zotz and Volrath. 2003). Secondary hemiepiphytes start their life by germinating on the ground and seedlings search for a host to ascend (Balcázar-Vargas et al., 2015). Once they reach the tree crown, as with true epiphytes, their longevity depends on the survival of the host plant (Suzanti et al., 2016).

3.1.2 *Ficus* as epiphytes

There are about 800 described *Ficus* (Moraceae) species. About 500 of them are hemiepiphytes while 300 are non-hemiepiphytes that include shrubs, free-standing trees and climbers (Berg and Corner, 2005). Most hemiepiphytic *Ficus* species are strangler figs that start their life after germination in the canopy or trunk of a host tree and after a period as epiphytes they spread their roots to the ground. They are primary hemiepiphytes in the sense of (Nadkarni et al., 2001). There are also creepers and scandent species that germinate on the ground and climb up and over other plants they come into contact.

Within the genus *Ficus* the habit of hemiepiphytism has evolved perhaps four times (Harrison, 2005). Hemipepiphytic fig trees get their mechanical support from the host while competing with it for light, nutrients and space (Male and Roberts, 2005). Hemiepiphytes can have the ability to strangle the host before becoming independent fig trees (Hao et al., 2016). Stranglers can benefit by having low risks of fire, floods, terrestrial herbivores and damage caused by falling debris (Hao et al., 2016). Conservative water use and better drought tolerance with strong control on stomatal conductance are important adaptations of hemiepiphytes (Hoolbrook and Putz, 1996a; Hoolbrook and Putz, 1996b). Strangler figs germinate from tiny epiphytic seedlings into terrestrial rooted trees that exhibit root fusion and larger species can surround and

eventually kill their hosts (Holbrook and Putz, 1996a). Some host trees have chemical and morphological features that make colonisation by hemiepiphytic figs less likely. These include a lack of deep branch crotches and hollows and a smooth bark (Compton and Musgrave, 1993; Wyse and Burns, 2011). These features are all absent in many palms, which is why they are frequently colonised by hemiepiphytic and other fig trees.

3.1.3 Epiphytes in oil palm plantations

Epiphyte establishment is associated with host traits such as plant morphology (Male and Roberts, 2005), bark stability (Wagner et al., 2015), water retention capacity (Wagner et al., 2015) and the presence of allelopathic components (Benzing, 1990). Palm trees have a prominent crown and a trunk that favours high detritus accumulation that allows epiphyte seeds to take hold and germinate (Kramer, 2011). The rough bark and leaf bases of oil palms provides a strong foothold that lets epiphytes avoid being washed away (Wagner et al., 2015) and allows a thicker layer of 'canopy soil' to develop (Williams and Sillett, 2007). The bases of the large leaf petioles store the humus and other organic materials that form canopy soil and helps keep them damp (Sofiyanti, 2013). Once the epiphytes successfully germinate they can remain growing until the surfaces they are attached to are shed, perhaps due to their weight (Zotz, 1998).

Thanks to their dramatic increase in popularity as a crop, oil palm plantations are now a major host for tropical epiphytes. The biotic and abiotic factors that determine the abundance of epiphytes in oil palm plantations have been identified as climate, especially water availability (from the rain), host plant size and the type of soil (Ding et al., 2016). Luskin and Potts (2011) concluded that the age of oil palm trees affects the abundance of their epiphytes (Teuscher et al., 2015). As plantation age increases, the canopy closes and increases the structural complexity if the epiphyte communities remain undisturbed (Luskin and Potts, 2011).

Each oil palm plantation has a different management system (Teuscher et al., 2015). They differ in terms of the intensity and frequency of use of fertilizers and herbicides to remove ground flora weeds and epiphytes on the trees (Roll et al., 2015). With smallholders, the clearing of oil palm trunks does not occur systematically or routinely as they are self-financed, self-managed, and free to decide their own practices (Dislich et al., 2015). 'Poor' management in smallholdings allows epiphyte

communities to become well established on the oil palm trunks and maintain their complexity (Foster et al., 2011). This contrasts with the actions typical of bigger companies (Azhar et al., 2015). In big and managed plantations, removing the epiphytes is conducted routinely as they are seen as obstructing the collection of mature palm fruits (Prescott et al., 2015).

A few decades ago, epiphytes were regularly removed from nearly 40% of mature plantations in West Malaysia to reduce the competition between the epiphytes and the oil palm trees (Piggott, 1980). However, removing the epiphytes leads to loss of many species that depend on epiphytes as their habitat. The growth of epiphytes creates a soil, litter and foliage environment on oil palms that can act as resource for herbivorous insects (Schmidt and Zotz, 2001). Among the dominant insects on epiphytes in oil palm plantations are weevils and cockroaches which themselves attract insect predators and parasites such as ants and braconid wasps (Suzanti et al., 2016). Epiphytes can also support pollinator populations (Wittman, 2000). Bryophytes, pteridophytes and angiosperms all commonly inhabit oil palm plantations. Fig trees are among the common angiosperms and their figs may provide food for birds in oil palm plantations.

3.1.4 Frugivores

Frugivores have shaped the evolution of fruit traits in *Ficus* (Lomascolo et al., 2010) and influenced the diversity of fig sizes and positions seen today (Compton et al., 2010). Smaller fruits, containing smaller seeds, have a broader spectrum of seed dispersal agents than larger-seeded fruits (Corlett, 2017). Ficus species usually produce smaller seeds than many angiosperms and often produce large quantities of seeds (Mumford, 2008). Many species also produce fruits at times of year when other trees are not fruiting (Mackay et al., 2018). More species of animals are recorded as feeding on figs than any other fruits (Shanahan et al., 2001). Among the known seed dispersers for Ficus are birds (Kerdkaew, 2018), bats (Lambert, 1991), apes (Mumford, 2008), monkeys (Chaves et al., 2018), squirrels and tree shrews (Peh and Fong, 2003) and others. Some hemiepiphytes undergo two-phased seed dispersal first by vertebrates followed by secondary dispersal by ants (Kaufmann et al., 1991). Primary hemiepiphytes and true epiphytes require their seeds to be deposited on host trees, whereas this is not important for secondary hemiepiphytes such as fig trees that grow as creepers. Terrestrial mammals that feed close to the ground can disperse these plants to suitable germination sites, but not the others.

Conversion of natural forest and fragmentation of habitats reduces fruit availability for birds and mammals (Chaves et al., 2018). Oil palm plantations support much less biodiversity than forests, but epiphytes growing on the palms can partially compensate (Turner and Foster, 2009; Vijay et al., 2016). Fig trees are often seen as being difficult to identify, and consequently they are not as well studied as some other groups of epiphytes. If they are allowed to mature, their figs can nonetheless support frugivores. In Peninsular Malaysia, some varieties of *Ficus deltoidea* are true epiphyte and are a widespread in oil palm plantations.

3.2 Objectives

We addressed the following questions about fig trees growing as epiphytes and hemiepiphytes on oil palms in Peninsular Malaysia (1) How common is *F. deltoidea* as an epiphyte in oil palm plantations around Kuala Lumpur? (2) Which groups of epiphytes are present in oil palm plantations and do they grow on the same individual palms as the fig trees? (3) Which *Ficus* species are present on the oil palms as epiphytes, what are their growth forms and do they ever manage to produce fruit? (4) What proportion of palms support the epiphytes and what densities? (5) Do different *Ficus* species become established on different parts and at different heights on the palms? And (6) which species of frugivores visit the *F. deltoidea* epiphytes.

3.3 Methods

3.3.1 Study sites

A survey was carried out to assess the presence of *F. deltoidea* var. *angustifolia* in oil palm plantations in an area near Kuala Lumpur. All plantations along the roadsides between the Banting and Dengkil study sites (which are approximately 10 kms apart) were assessed. The numbers of var. *angustifolia* were determined by naked eye counts of how many plants could be seen growing as epiphytes on oil palms within 20 m of the road. Accurate assessments could not be made for distances further into the plantations. No other varieties were seen.

Five plantations with high densities of *F. deltoidea* were used to assess their epiphyte communities. The plantations were at Banting and Dengkil in the state of Selangor, near Kuala Lumpur; Batu Pahat, to the South in the state of Johor, Bagan Serai in Perak state to the north of Kuala Lumpur and Tembila plantation on the east coast of Malaysia, in the state of Terengganu. Details of their locations, soils and climates were described in Chapter 2.

3.3.2 Epiphyte assessments

All the oil palms in the available sections of each plantation were assessed for the abundance and diversity of their epiphytes. Their trunks were scored based on their percentage cover of bryophytes, pteridophytes, *Ficus deltoidea*, other *Ficus* species and other angiosperms. The remaining bare areas of trunk were also recorded.

The heights of the oil palm trunks were measured using an adjustable pole and measuring tape and the percentage cover of each plant group was then estimated by eye. If there was overlap of different plant groups, the microepiphytes cover was omitted. For *Ficus* species, the heights where they appeared to have originated were recorded and herbarium samples were made for identification purposes. The identifications of *Ficus* species were made from herbarium specimens presented to botanists from Universiti Kebangsaan Malaysia (UKM). The growth forms of the *Ficus* species (creeper, strangler or true epiphyte) were also recorded. No attempt was made to further identify the bryophytes, pteridophytes and other angiosperms.

3.3.3 Ficus deltoidea assessment

For each individual of *F. deltoidea*, its variety was recorded as well as the sex if figs were present. The heights and diameters of thickest stems of the fig trees were also noted. Some *F. deltoidea* were inaccessible and their details could not be recorded. The diameter of the thickest stem was measured using Vernier calliper. For plant apparent sex ratio assessment, the data were collected on 11th March 2017.

3.3.4 Frugivore observations

In Batu Pahat plantation, six camera traps (Bushnell 24MP Trophy Cam HD Aggressor) were installed opposite large female *F. deltoidea* individuals that had high numbers of figs. This was to record any frugivores present on the *F. deltoidea*. The cameras were installed at heights that provided coverage of the whole crowns of the plants. The traps were left in place for two weeks from 2nd July 2017 until 15th July 2017. Any frugivorous birds and mammals were noted.

3.3.5 Statistical methods

A normality test using Kolmogorov-Smirnov was conducted for each analysis to see whether the distribution was normal or not. ANOVA was later used if the distribution was nomal while Kruskal-Wallis for non-normal distribution. Tukey and Mann-Whitney test were later conducted for normal and non-normal respectively to see the diffrences occurred between groups.

Differences in the proportions of the trunks covered by *Ficus* and *Ficus deltoidea* groups were analysed using non-parametric Kruskal-Wallis tests. The differences in height of origin of the *Ficus* species on the trunks were analysed Kruskal-Wallis again. For this analysis, the creeper *Ficus sagitatta* was excluded because it comes into contact with the trunks near or at ground level.

Differences in the numbers of individuals of two varieties and the frequency of their sexes at Batu Pahat were compared using Chi square tests. For each variety, two way contingency Chi square tests were performed to test against a 50:50 sex ratio. The height preferences of the two varieties of *F. deltoidea* and the male and female trees were analysed using ANOVA if the distribution was normal and Mann-Whitney tests if the distribution was not normal. A more detailed study was conducted of *F. deltoidea* var. *angustifolia* at the Banting plantation. The heights of male and female plants on the oil palms were compared using ANOVA. All test were all performed in SPSS Statistics 20.

3.4 Results

3.4.1 Ficus deltoidea var. angustifolia in plantations near Kuala Lumpur

There are 26 other oil palm plantations along the 10 km of road between Banting and Dengkil, and one residential area that is surrounded by some oil palms within their compound. From all the plantations observed, only 4 showed the existence of *F. deltoidea* var. *angustifolia* within 20 m from the main road. These were the start and end points of Banting (23 trees) and Dengkil (22 trees) plus the houses (5 trees) and Plantation 26 (18 trees) (Table 3.1). The rest of the plantations did not show any signs of *F. deltoidea* presence on the trunks of the oil palms when viewed from the road.

Plantation	Distance from previous	F. deltoidea		
	plantation (m)	var. angustifolia		
Banting plantation	0	23		
Plantation 1	500	0		
Plantation 2	47	0		
Plantation 3	110	0		
Plantation 4	290	0		
Plantation 5	200	0		
Plantation 6	650	0		
Plantation 7	170	0		
Houses	130	5		
Plantation 8	1300	0		
Plantation 9	300	0		
Plantation 10	280	0		
Plantation 11	200	0		
Plantation 12	170	0		
Plantation 13	400	0		
Plantation 14	650	0		
Plantation 15	200	0		
Plantation 16	500	0		
Plantation 17	400	0		
Plantation 18	600	0		

Table 3.1 The number of *Ficus deltoidea* var. *angustifolia* sighted within 20 m from the road.

Plantation 19	270	0
Plantation 20	210	0
Plantation 21	290	0
Plantation 22	140	0
Plantation 23	400	0
Plantation 24	200	0
Plantation 25	190	0
Plantation 26	500	18
Dengkil plantation	800	22
Total	10097	68

3.4.2 Epiphyte loads

All of the 1091 oil palm trunks at the five sites that were assessed supported at least two of the groups of epiphytes on their trunks, with a maximum of five groups. Five groups of epiphytes was the most common frequency at most sites. In all study sites some bare trunk was present on almost all the oil palm trees assessed (Table 3.2). Bryophytes and pteridophytes were also present on almost all trunks assessed. The only plantation that did not have other *Ficus* in addition to *F. deltoidea* on the trunks was in Batu Pahat.

The sites had different percentage trunk coverage by the groups of epiphytes (Figure 3.1, calculated by summing trunk coverage across trees). Bare trunks ranged from 3.2 % of the total trunk area (Banting) to 24.34 % (Dengkil). Bryophytes showed a high percentage coverage all sites ranging from 18.70% to maximum 33.24%. Pteridophytes also covered 30% or more of the trunks at sites other than Banting. Non-*Ficus* Angiosperms were less abundant, ranging from 5.42% to 25.83% coverage. *Ficus* other than *F. deltoidea* had the highest coverage at Banting and were absent from Batu Pahat, which had the highest coverage by *F. deltoidea* (Table 3.2. There was a significant difference in percentage coverage by *F. deltoidea* groups (Kruskal-Wallis, $\chi^2 = 97.35$, df = 4, P < 0.001) and *Ficus* group (Kruskal-Wallis, $\chi^2 = 324.04$, df = 4, P < 0.001) between sites.

Frequency of trees	Study sites				
	Banting	Dengkil	Batu Pahat	Bagan Serai	Tembila
Bare trunk	284	427	101	145	133
Ficus deltoidea	88	128	66	32	53
Other Ficus	74	311	0	46	9
Bryophytes	283	427	101	145	133
Pteridophytes	280	426	101	145	134
Other Angiosperms	185	237	87	110	126
Total trunks	284	427	101	145	134

Table 3.2. Frequency of bare trunks and epiphyte groups in five different plantations.



Figure 3.1 The cover of oil palm trunks by different epiphyte groups. Open bars = bare trunks, solid bars = *Ficus deltoidea*, grey bars = other *Ficus*, spotted bars = bryophytes, hashed bars = pteridophytes, horizontal lines = other Angiosperms.

3.4.3 Distribution of Ficus species

A total of 11 *Ficus* species were recorded as epiphytes or creepers in the plantations. Eight of the *Ficus* species were hemiepiphytes, one was a true epiphyte (three varieties) and one root climber (Table 3.3). During the study, four species produced figs and *F. deltoidea* was the only species that produced figs all the time (Table 3.3).

Dengkil had the largest number of *Ficus* species while Batu Pahat had the least, with only one species (two varieties) (Figure 3.2 - Figure 3.6). The most common *Ficus* species was *Ficus deltoidea* var. *angustifolia* except for at the Tembila site. At Johor a mean of more than one per host of this plant was recorded and some palms supported both of the *F. deltoidea* varieties. The maximum number of *F. deltoidea* present on one trunk were 9 trees where 5 of them were from var. *angustifolia* while the other 4 were from var. *deltoidea*. In Tembila, *F. deltoidea* var. *trengganuensis* was almost the only *Ficus* present. Here the number of var. *trengganuensis* per palm trunk ranged from 1-4.

F. microcarpa, F. obscura and *F. benjamina* were also quite common *Ficus* species at three study sites. The least overall proportion of *Ficus* present at all study sites was *F. virens*, only found in Tembila and the occurrence ratio was very low. There was a significant difference in the number *F. deltoidea* present per oil palm trunk at different sites (Kruskal-Wallis, $\chi^2 = 80.10$, df = 4, P < 0.001) with Batu Pahat having the highest mean Table 3.2) while the numbers of plants were rather similar at Banting, Dengkil and Bagan Serai.

<i>Ficus</i> species	Based on Berg a	Produced	
	Growth form	Maximum height (m)	figs ?
<i>F. deltoidea</i> var. <i>angustifolia</i> (Miq.) Corner	True epiphytes	7	Yes
F. deltoidea var. deltoidea Corner	True epiphytes	7	Yes
F. deltoidea var. trengganuensis Corner	True epiphytes	7	Yes
F. benjamina L.	Hemiepiphytes	35	No
<i>F. caulocarpa</i> (Miq). Miq.	Hemiepiphytes	20	No
<i>F. microcarpa</i> L.f.	Hemiepiphytes	30	No
<i>F. obscura</i> Blume	Hemiepiphytes	8	Yes
<i>F. religiosa</i> L.	Hemiepiphytes	25	No
F. sagittata J. Konig ex Vahl	Root-climber	Depends on host height	No
<i>F. sinuata</i> Thunb.	Hemiepiphytes	6	Yes
<i>F. tinctoria</i> G. Forst.	Hemiepiphytes	15	Yes
<i>F. viren</i> s Aiton	Hemiepiphytes	35	No

 Table 3.3. Characteristics of epiphytic *Ficus* species on the trunk of oil palms.



Figure 3.2 Frequencies of *Ficus* species in Banting plantation. *F. deltoidea* (*a*) = var. *angustifolia*, *F. deltoidea* (*d*) = var. *deltoidea*, *F. deltoidea* (*t*) = var. *trengganuensis*.


Figure 3.3 Frequencies of *Ficus* species in Dengkil plantation. *F. deltoidea* (a) = var. angustifolia, *F. deltoidea* (d) = var. deltoidea, *F. deltoidea* (t) = var. trengganuensis.



Figure 3.4 Frequencies of *Ficus* species in Batu Pahat plantation. *F. deltoidea* (a) = var. angustifolia, *F. deltoidea* (d) = var. deltoidea, *F. deltoidea* (t) = var. trengganuensis.



Figure 3.5 Frequencies of *Ficus* species in Bagan Seral plantation. *F. deltoidea* (a) = var. angustifolia, *F. deltoidea* (d) = var. deltoidea, *F. deltoidea* (t) = var. trengganuensis.



Figure 3.6 Frequencies of *Ficus* species in Tembila plantation. *F. deltoidea* (a) = var. angustifolia, *F. deltoidea* (d) = var. deltoidea, *F. deltoidea* (t) = var. trengganuensis.

3.4.4 Differences in establishment heights of *Ficus* on oil palms

In Banting, the trunks of oil palms ranged from 0.5 m to 8.8 m in height (5.68 ± 1.28 m) (Mean ± SD) (Figure 3.7). Most of the *Ficus* individuals of all species inhabited the middle parts of the trunks with the highest *F. deltoidea* var. *angustifolia*, found at 8.1 m above the ground. The small number of *Ficus caulocarpa* were usually towards the top of the trunks, with none lower than 4.2 m, but there was no significant difference in height occupied by the six *Ficus* taxa in this site (Kruskall-Wallis, $\chi^2 = 7.20$, df = 5, P > 0.05).

The oil palms at Dengkil ranged from 0.8 m to 9.1 m in height (6.71 ± 1.36 m) (Figure 3.8). As before most of the *Ficus* inhabited the middle parts of the trunks at around 5 m but there was a significant difference in establishment heights across the eight *Ficus* species (Kruskal-Wallis, $\chi^2 = 57.61$, df = 7, P < 0.001). Mann-Whitney tests revealed *that F. religiosa* was present significantly higher up the trunks than some of the other species, but not var. *angustifolia*.

Only two *Ficus* species were present at Batu Pahat (*F. deltoidea* var. *angustifolia* and *F. deltoidea* var. *deltoidea*). The overall trunk heights of the oil palm trees ranged from 0.5 to 9.5 m with a mean of 3.88 ± 1.95 m. The two *F. deltoidea* varieties had a similar range of occupancy from 0.1 to 9.3 m above the ground (Figure 3.9. There was no height preference between the two varieties of *F. deltoidea* (Mann-Whitney, Mdn = 3016.50, df = 1, P > 0.05).

The oil palm trunk heights at Bagan Serai ranged between 1.0 and 8.5 m (5.70 \pm 1.36 m). Five *Ficus* species were found in this site including a creeper species (*F. sagittata*), with most located between 3 and 4 m above the ground (Figure 3.10). There was no significant difference in heights occupied by the four *Ficus* taxa in this site (ANOVA, F = 1.82, df = 3, P > 0.05).

At Tembila, the oil palm trunks were shorter than at the other study sites as they ranged from 1.7 m to 4.8 m (mean = 2.77 ± 0.65 m). *F. virens* occupied the lower part of the trunks while most of the *F. microcarpa* were higher (Figure 3.11). There was a significant difference in establishment heights across the three species of *Ficus* using (Kruskal-Wallis, $\chi^2 = 9.0$, df = 2, P < 0.001) and Mann-Whitney tests revealed three different groups of *Ficus* were present based on their establishment heights.



Figure 3.7 The heights of apparent establishment sites of *Ficus* species in Banting plantation. *F. deltoidea* (a) = *Ficus deltoidea* var. *angustifolia*. The lines in boxes indicate the median, the boxes indicate upper and lower quartiles with extreme values at the end.



Figure 3.8 The heights of apparent establishment sites of *Ficus* species in Dengkil plantation. *F. deltoidea* (*a*) = *Ficus deltoidea* var. *angustifolia*. The lines in boxes indicate the median, the boxes indicate upper and lower quartiles with extreme values at the end. Values with different superscripts differed significantly in Mann-Whitney tests (P < 0.05).



Figure 3.9 The heights of apparent establishment sites of *Ficus* species in Batu Pahat plantation. *F. deltoidea* (*a*) = *Ficus deltoidea* var. *angustifolia*. *F. deltoidea* (*d*) = *Ficus deltoidea* var. *deltoidea*. The lines in boxes indicate the median, the boxes indicate upper and lower quartiles with extreme values at the end.



Figure 3.10 The heights of apparent establishment sites of *Ficus* species in Bagan Serai plantation. *F. deltoidea* (a) = *Ficus deltoidea* var. *angustifolia*. The lines in boxes indicate the median, the boxes indicate upper and lower quartiles with extreme values at the end.



Figure 3.11 The heights of apparent establishment sites of *Ficus* species in Tembila plantation. *F. deltoidea* (t) = *Ficus deltoidea* var. *trengganuensis*. The lines in boxes indicate the median, the boxes indicate upper and lower quartiles with extreme values at the end. Values with different superscripts differed significantly in Mann-Whitney tests (P< 0.05).

3.4.5 Ficus deltoidea var. angustifolia and Ficus deltoidea var. deltoidea

The only site with two varieties of *F. deltoidea* present was Batu Pahat, where they were both abundant. A total of 101 oil palm trunks supported 104 var. *angustifolia* and 64 var. *deltoidea*. For var. *angustifolia*, 34 males and 11 females were identified while for var. *deltoidea*, 21 males and 14 females were identified. Comparing the sex ratios of the two varieties on 11th March 2017 using a Chi Square test, there was no significant difference in sex ratios between species ($\chi^2 = 2.22$, df = 1, P > 0.05). The sex ratios within var. *angustifolia* was biased towards males at this time of year (Chi square, $\chi^2 = 6.81$, df = 1, P < 0.01 but for var. *deltoidea* no significant was detected for the sex ratio using Chi square again ($\chi^2 = 0.92$, df = 1, P > 0.05).

The two varieties occupied similar heights on the trees, with most located between 2 and 3 m above the ground (Figure 3.12). There was no height preference between the two varieties of *F. deltoidea* (Mann-Whitney, Mdn = 3016.50, df = 1, P > 0.05). Male and females from these two varieties also did not prefer different heights (ANOVA in var. *angustifolia* F = 0.26, df = 1,43, P > 0.05 and var. *deltoidea* F = 0.11, df = 1,32, P > 0.05).



Figure 3.12 Differences in heights on oil palm trunks between sexes of two varieties of *F. deltoidea* in Batu Pahat plantation. *F. deltoidea* (a) = *Ficus deltoidea* var. *angustifolia*, *F. deltoidea* (d) = *Ficus deltoidea* var. *deltoidea*.

3.4.6 Ficus deltoidea var. angustifolia in Banting

The only variety of *F. deltoidea* present In Banting was var. *angustifolia*. From a total of 113 plants, 28 were male, 24 were female, 30 did not produce any figs and 31 were too high to be assessed for their sex. Male trees had establishment sites between 0.4 to 7.5 m (3.18 ± 1.87 m) above the ground, while in females the range was from 0.2m to 7.3 m (3.18 ± 2.18 m). The sexes did not differ in the heights they occupied (ANOVA, F = 0.10, df = 1, 50, P > 0.05).

3.4.7 Frugivore observations

The observations were made for a two-week period from 2^{nd} July 2017 until 15th July 2017. Only one bird species was recorded, for a total of three times within the female *F. deltoidea* var. *deltoidea* and the same species was found on *F. deltoidea* var. *angustifolia* (five times). At least two individuals were present. The bird was identified by a local ornithologist (Dr Thanate Kerdkaew) as *Pycnonotus goiavier*, the yellow-vented bulbul (Figure 3.13). Both varieties showed synchronous fig phenology where all development phases were present on the tree during the observations. The camera traps also recorded a wild boar (*Sus scrofa*) but no other mammals. Feeding on the figs was not confirmed by the camera trap records.



Figure 3.13 *Pycnonotus goiavier* on a female *Ficus deltoidea* var. *deltoidea* (the red arrow).

3.5 Discussion

3.5.1 Epiphytes

Tropical environments with a high humidity favour a high diversity of epiphytes (Thomas and Packham, 2007). However, most organisms are present at low diversity and abundance in oil palm plantations when compared to natural forests and the expansion of oil palm plantations is a major reason for biodiversity loss in SE Asia (Foster et al., 2011). The presence of epiphytes in these plantations can help to partially mitigate this biodiversity loss without having any impact on productivity (Prescott et al., 2015). Bryophytes, pteridophytes and angiosperms are common epiphytes on trees in general (Muňoz et al., 2003; Mehltreter et al., 2005) and oil palm trunks specifically (Nadarajah and Nawawi, 1993; Prescott et al., 2015).

Our surveys of oil palm epiphytes cannot be seen as representative of plantations in general in Peninsular Malaysia as the plantations were selected only if they supported *F. deltoidea* as epiphytes. In all the plantations the percentages of bare trunk were low. Some of the bare trunks come from extensive shedding especially in very old oil palm trees but most comes from the trunk surface where no epiphytes were present.

The epiphyte communities on the oil palms differed between study sites and among tree within sites. Affield (2008) similarly found a high variability in species composition of epiphytes on *Metrosideros robusta* A. Cunn. (Myrtaceae) in different study sites and trees within sites and suggested this may be due to the irregularity of spore and seed colonisation and variable success in establishment on the host plants. He also postulated that the presence of vascular epiphytes depends on there also being non-vascular epiphytes on the trees, because a significant relationship between them was detected. Epiphytic mosses and lichens are said to aid vascular plant establishment as they provide a rough surface and moist environment for seed germination (Zotz, 1998).

In oil palm plantations, variable management practices are also likely to have influenced the composition of the epiphyte communities we recorded. Large companies often routinely clear epiphytes, but one of the five plantations included in this study is managed by a big company and had many epiphytes on the trees. Whether this was deliberate policy or an oversight by the company is unknown. Pteridophytes were important components of the epiphyte floras at all the sites other than Banting. Apart from at Banting, ferns covered more than 30% of the trunk surfaces. Mosses were also frequent, as were a variety of angiosperms. These included several species of fig trees, a group that are particularly important for biodiversity because of the number of animals that eat their fruits (Shanahan et al., 2001) and leaves (Ødegaard, 2000).

3.5.2 Ficus species

Globally, hemiepiphytic *Ficus* (strangler figs) include more described species (about 500) than non-hemiepiphytes (about 300 species) (Harrison, 2005), but most stranglers are monoecious, and there are more dioecious than monoecious species in SE Asia. Among the ten species of *Ficus* identified as epiphytes in this study, eight were hemiepiphytes. These included some of the more common hemiepiphytic *Ficus* present in Malaysia (Sreekar et al., 2010; Hodgkison et al., 2013; Badron et al., 2014). There were also the two varieties of *F. deltoidea* and the creeper *F. sagittata*, which ascended the palm trunks after germinating on the ground.

Ficus species can have roots up to 100 m long, but the epiphytic *F. deltoidea* in the oil palm plantations rarely had roots longer than 1 m. Only certain varieties of this species are sometimes or always epiphytic and it also commonly grows on rocks, (Berg and Corner, 2005). It grows more often as an epiphyte in the lowlands, and can be particularly abundant in riparian situations (Corner, 1969). The abundance of strangler figs may reflect their high tolerance to drought and other adaptations for the conditions that epiphyte seedlings are subjected to (Holbrook and Putz 1996a, 1996b; Hao et al., 2010, 2013, 2016). Hemiepiphytic figs are often bigger than non-hemiepiphytes and produce very large numbers of figs (Lambert and Marshal, 1991). These are often dispersed by birds, which are likely to be the main seed dispersal agents in oil palm plantations, so the seeds of epiphytic *Ficus* may also be deposited on to palms in larger numbers than the seeds of fig trees with other growth forms (Peh and Fong, 2003; Hao et al., 2013).

Larger palm trees were expected to support more epiphytic *Ficus* as their larger size can accommodate more individuals (Wagner et al., 2015) and fig trees are more likely to flourish upon larger host trees (Male and Robert, 2009). Larger palms are also likely to be older and to have had more time to be colonised. According to Boelter et al. (2014), soils and host size control the structure of epiphyte communities. In this study,

Banting, Bagan Serai and Batu Pahat are on peat soils, Dengkil is on a silt soil and Tembila is on a sandy soil. Peat soil has a fibre content, low permeability and low shear strength (Teong et al., 2016). The sample size is too limited to say whether soil may have influenced which *Ficus* species were on the palms, and the range of palm heights may have been too small to detect size effects, though the trees were significantly smaller at Tembila, where the tallest palm was only 4.8 m high and the sandy soil there has less ability in retaining water and nutrients (Weber et al., 2007).

Dengkil had the highest number of *Ficus* species growing on the oil palms (8 species) while the fewest species were recorded in Batu Pahat, with only one species. *F. deltoidea* var. *trengganuensis* has a very restricted range, and the plantation in Terengganu was the only one sampled within its range. The other species and varieties are more widely distributed and could potentially have been recorded at most or all of the sites.

The amount of light received during seed germination and the heights on the trunk that provide more water, nutrient and humus are important for the establishment of hemiepiphytes (Putz and Holbrook, 1986; Daniels and Lawton, 1991). Most individuals of all the *Ficus* species occupied medium to higher parts of the trunks, which is consistent with studies by Wyse and Burns (2011) and Benzing (2004) who found most epiphytes inhabit the upper regions of host trees. Within sites, there were rarely differences between the heights occupied by different *Ficus* species.

3.5.3 Ficus deltoidea as an epiphyte

The massive transformation from forest to oil palm plantation in Malaysia has provided *F. deltoidea* with more suitable habitat and it has probably extended its range. Before the transformation to oil palms, the distributions of the varieties of *F. deltoidea* in Peninsular Malaysia were mainly on the limestone hills, mountains, mossy forests, rocks and the upper dunes on the sea-shore. These are areas where the quality of the soil is poor and well-drained. Some of the varieties were predominantly epiphytes (var. *deltoidea*, var. *kunstleri*, var. *trengganuensis*) (Corner, 1969). At least three different forms of *F. deltoidea* were recorded to colonise oil palm plantations in Peninsular Malaysia namely var. *angustifolia* (in Johor, Perak and Selangor), var. *deltoidea* (Johor) and var. *trengganuensis* (Terengganu).

In the plantations surveyed near Kuala Lumpur, *F. deltoidea* var. *angustifolia* either existed in large numbers or there were none at all. Some plantations are likely to have had all their *Ficus* epiphytes removed, but this also suggests that the seeds are dispersed within a short range. Long distance dispersal can increase genetic diversity and reduce population genetic structuring (Mondragon and Calvo-Irabien, 2006), so the extent of seed dispersal can be assessed in the future using genetic techniques.

The higher number of *F. deltoidea* (belonging to two varieties) at Batu Pahat may be because the plantation is very old and the palm trees barely produce fruit. Due to this fact, the owners may neglect cutting the epiphytes since they no longer need to collect the palm fruit. This plantation also acts as source of *F. deltoidea* leaves for commercial products like tea and also for scientific research so cutting the fig trees would reduce their stocks.

The slow growth of *F. deltoidea* and its production of figs sequentially and asynchronously (Chapter 6) may be linked to their occurrence in nutrient poor habitats (Berg and Corner, 2005). Traits developed in response to poor soils, including the unusually large seeds of this species (Chapter 4) may have represented a preadaptation that allowed them to survive as epiphytes. Species with larger seeds have the ability to survive poorer quality environments like deep shade, drought, physical damage and competition with other vegetation (Westoby et al., 1996). Larger seeded species also tend to have a lower specific leaf area (SLA) (leaf area per leaf dry mass), which causes their slow growth. At least under glasshouse growing conditions, the seedlings of F. deltoidea var. angustifolia have stems (hypocotyls) of up to five centimetres before the cotyledons, which should increase their chance of reaching sunlight through competing vegetation (unpublished observations). This is made possible by the greater reserves they have available to support respiration. The unusually large seeds of *F. deltoidea* are surrounded by a mucilaginous exocarp and seeds kept in warm damp conditions are resistant to mould growth, unlike the other parts of the figs (S.G. Compton, personal communication). Mucilage has also been suggested in other *Ficus* species to inhibit germination until gut-passage by birds, and to act as an ant-attractant (Kauffman et al., 1991).

Only Batu Pahat supported two varieties of *F. deltoidea* within the same plantation. Some of the trunks supported both varieties and one trunk had a total of 9 individuals of both varieties. Living on the same trunk triggers competition for light, humidity and space (Sales et al., 2016), and although there was no significant difference between the varieties in terms of abundance and height preferences, most of the *F. deltoidea* sharing a trunk did not produce figs and were often smaller in size than elsewhere. Lack of nutrients may be inhibiting fig production.

3.5.4 Frugivores

Figs are eaten by many vertebrates, especially birds (Shanahan and Compton, 2001; Mackay et al., 2018). The most commonly recorded birds eating figs in SE Asia are bulbuls (Pycnonotidae). The small size of the figs of F. deltoidea suggests that their seeds are likely to be dispersed mainly by birds, but Orang Utans were the only published feeders on these figs in Malaysia (Shanahan et al., 2001). At Banting, two individuals of Pycnonotus goiavier (Yellow-vented bulbul) were recorded by the camera traps placed opposite a female F. deltoidea tree crown. However, as there was no film of the bulbuls eating the figs, it cannot be confirmed that this species eats the figs and is a likely seed dispersers for this species. A study by Rajpar and Zakaria (2010) found that this was the most abundant bird species in Paya Indah Wetland (Peninsular Malaysia). This is also a common species across southern Thailand and Malaysia (T. Kerdkaew personal communication). Many of the fig tree species recorded as epiphytes of the oil palms grow into large trees and only fruit once they have reached a large size (Berg and Corner, 2005). Plantation managers are unlikely to allow them to ever reach so big, so they are not likely to ever be allowed to provide figs that could be used by birds.

3.6 Conclusions

The diversity and abundance of fig trees and other epiphytes on the oil palm trunks was high. The plantations were not necessarily typical but show that given suitable management epiphytes can improve biodiversity in this form of monocultural agriculture. However many of the fig tree species recorded have to be large before reaching maturity and they would not normally be allowed to reach this size because they would be seen as damaging their host trees. *F. deltoidea* is the exception, and its abundant fruiting suggests it is particularly important for biodiversity in oil palm plantations.

Chapter 4

Characterisation of Ficus deltoidea varieties

4.1 Introduction

4.1.1 Ficus

Ficus (Moraceae) is among the most studied plant genera due to its unique closed inflorescence, variable breeding systems, different growth forms and unique relationship with its pollinators. With more than 800 described species (Weiblen, 2002), fig trees represent more than 75% of all the species in the family Moraceae (Zhang et al., 2018). This genus was also among the earliest domesticated crops and today is widely planted (Lansky et al., 2008), with the edible fig (*Ficus carica* L.) and ornamental species cultivated worldwide.

The genus *Ficus* as a whole is characterised by having an urn-shaped inflorescence commonly known as the fig or syconium (Chen et al., 2009). Entry to the flowers requires passage through a typically bract-lined ostiole (Thorogood et al., 2018). Male and female flowers or only female flowers are present in each fig. Fig trees have two breeding systems, dioecy and monoecy (Dunn et al., 2008). Roughly half of all fig species are monoecious, with individual inflorescences providing both female (seed production and dispersal) and male (pollen production and dispersal) functions (Herre et al., 2008). The remaining *Ficus* species are functionally dioecious.

In these species, there are two types of trees; female trees that produce only seedbearing fruits; and male trees with figs that produce only pollen and support development of pollinator wasp progeny to transport the pollen (Patel and Hossaert-McKey, 2000). In a dioecious fig tree, the female fig wasps that enter the female figs pollinate flowers but they produce only seeds (Ghana et al., 2015a). The long styles of flowers in the female figs and the structure of the stigmas prevent the wasps from galling the flowers and laying their eggs and so allow only for seed production (Kjellberg et al., 2014). On male trees, the female flowers are modified for receiving a pollinator egg and only fig wasp larvae develop inside. Even pollinated flowers with no eggs deposited do not produce seeds. Approximately one to several months after oviposition, which is usually in flowers that have also been pollinated, the male fig wasp offspring emerge from their galls and mate with the still gall-enclosed females (Liu et al., 2013). As in monoecious figs, all the female flowers inside a fig become receptive simultaneously several weeks before the male flowers mature (protogyny) (Patel, 1996). The galled ovules of the short styled flowers in male figs are each consumed by a single pollinator larva which completes its development there.

The outer wall (receptacle) of a fig varies in colour depending on the species and developmental stage (Lansky et al., 2008). The flowers inside figs are tiny and simplified, with no petals, but sometimes with tepals, which can be prominent in some *Ficus* species. Tepals are elements of the perianth that cannot be distinguished as either sepals or petals (Warner et al., 2009). Among plants in general, sepals typically have a protective function, while petals are involved in pollinator attraction (Bedinger et al., 2017). Tepals can be involved with one or both of these functions. In one study of tepal function in *Chamissoa altissima* (Jacq.) Kunth (Amaranthaceae), the tepals were found to have protective and photosynthetic functions before they obtain a red colour due to betalain pigments and developed a later function in seed dispersal (de Oliveira et al., 2010). Another study on early-flowering common snowdrop (Galanthus nivalis L.) recorded tepal function as attracting the pollinators and facilitating the photosynthetic process (Aschan and Pfanz, 2006). The tepals of this species contains chlorophyll that provides photo-assimilates to the flower and the developing seeds. Other functions for tepals have also been described, including pollen capture (Cornille et al., 2012) and possibly as footholds for insects when pollinating the flowers in the genus Oxygonum (Polygonaceae) (Hong et al., 1998).

4.1.2 *Ficus* pollination

Fig trees are pollinated exclusively by fig wasps (Agaonidae). The pollination mutualism between fig trees and fig wasps is one of the classic examples of obligate mutualism (Thorogood et al., 2018). The unique enclosed uniovulate flowers inside the figs act as the venue for the mutualism with the figs wasps (Cook and Rasplus, 2003). The fig trees depend on fig wasps to carry their pollen and produce seed,

whereas the fig wasps need the *Ficus* inflorescences (figs) to provide food for developing larvae and complete their life cycles (Herre et al., 2008). Pollen produced by a fig is transported by the pollinators that develop in the fig. The morphology of fig wasps is well adapted to their biological cycle starting from entering the figs, oviposition, pollination, mating and finally departure from their natal figs. The wings and the antennae also routinely detach easily when she enters the figs (Kjellberg et al., 2005b). Most of all, pollinating fig wasps are very uniform in that they have similar life cycles (Wiebes, 1982).

The pollen-bearing adult female fig wasp is called a foundress when she reaches a suitable fig and they are responsible for the pollination in the fig trees as well as the production of their progeny. The foundress detects volatiles emitted by receptive figs of their particular host species and locates them using their antennae (Hossaert-McKey et al., 2016). Once inside they pollinate the flowers after entry through a narrow often bract-lined ostiole (Nefdt and Compton, 1996) and at the same time she lays eggs in some of the flowers if she has not entered a female fig. The ostiole is the only route to the inner surface of the figs and the narrow slit-like condition of the ostiole only allows the specific pollinator to get through it (Liu et al., 2013). In many species the foundresses rarely or never emerge from the first figs they enter, but in other species several figs can be entered by a single foundress. After the eggs are laid, and after larval feeding, the fig wasp's offspring emerge, mate within the figs and the female offspring will carry the pollen out from their natal figs and disperse.

The wingless male wasps hatch first from their galls and search for galls containing females. The males bite holes into the female galls and insert their genitalia in order to mate (Cook and Segar, 2010). Female wasps then emerge into the fig cavity and at this time the male fig flowers are mature and have mature pollen. After the females collect some pollen either actively or passively, the male fig wasps bore a hole to let the pollen-bearing female fig wasps get out from the natal fig. The females then search for other receptive figs to deposit their eggs and pollinate the flowers (Nefdt and Compton, 1996). It used to be thought that the wingless male fig wasps always will spend their entire life in the figs (Compton and McLaren, 1989). However, there are a few cases where males go outside their natal figs and even enter other figs looking for mates (Greeff, 2002). They can even help protect the females from ant predation on the surface of the figs (Zachariades et al., 2010).

In some pollinator species, the female fig wasps actively collect pollen into thoracic pollen pockets and later deposit it deliberately onto flower styles (Jousselin et al., 2003). In others, pollination is passive via transport on the wasp's body, as in most other insects (Waser and Ollerton, 2006). This occurs when the foundress gets coated by dehiscing pollen shed by the anthers on their way out of her natal fig (Kjellberg et al., 2014). In fig wasps that display passive pollination, the pollen grains are trapped in the abdominal pleura and are released as a wasp's gaster swells in the water-saturated fig cavity atmosphere (Galil and Neeman, 1977). Fig trees with passive pollinators have mature anthers that burst open naturally and have figs with high ratios of anthers to female flowers and so produce more pollen per fig to facilitate passive collection by their pollination and have offspring that mostly develop in fertilized flowers (Kjellberg et al., 2001). This behaviour is assumed to have evolved because fertilized flowers provide better nutrients for fig wasp larvae (Parrish et al., 2003) and there is experimental evidence that this is the case (Tarachai et al., 2008).

4.1.3 Species concepts in Ficus

In 1686, the term species was proposed by John Ray in his book Historia Plantarum saying that a species is a group of individuals that can interbreed together despite having 'accidental' variations between them. A broader concept was proposed by Carolus Linnaeus in his book Species Plantarum (1753) who classified all the known living organisms by looking at their morphology. He also proposed a binomial system and renamed plants and animals in the two word latinized system. In plants, he differentiated between them based on floral morphology and sexual characters. The most widely accepted species concept was later proposed by Ernst Mayr in 1942. He described species as "groups of actually or potentially interbreeding natural populations which are reproductively isolated from other such groups". This concept is used as the basic definition of species today.

In recent classifications, defining a species also requires taking into account variation within and between the species (Aldhebiani, 2018). In classification systems of plants, the kingdom is divided further by classes, orders, families, genera and species. They can be further described by using variety, cultivar, hybrid and other notations. Varieties indicate variation in inheritable differences within species. The difference between sub-species and varieties is not clear, but varieties tend to be used more frequently

when dealing with cultivated plants and do not necessarily reflect natural variation within species.

Ficus is considered as a 'difficult' genus by many tropical botanists. This results from similarities in ecology and the sharing of common traits, together with often highly variable leaf morphologies (Harrison, 2005). In determining the boundaries of *Ficus* species, plant anatomy is used as the main reference and current species names are based mainly on anatomical features of leaves, inflorescences and flowers. Luckily, the highly specific interaction with particular species of their obligate pollinators, together with the current development of molecular techniques, has enhanced the modern classification of *Ficus* and provided additional ways of recognising boundaries between species.

The most widely used current classification of *Ficus* is based on Berg and Corner (2005). It includes a contribution by the late ECJ Corner, but the species boundaries were largely determined by the late C.C. Berg, who generally opted to assume that two morphologically distinct entities belonged to the same species, if intermediates were also present between them. Biological features, including the identity of pollinators (the behaviour of which largely determines gene flow in *Ficus*) were not taken into account. Consequently, sympatric biologically distinct taxa (with gene flow not taking place between them) are not always recognised as being distinct species using the current morphological species concept.

Specific volatile compounds are released from the figs to attract the fig wasps associated with each species of fig tree (Hossaert-McKey et al., 2016). The volatiles that attract fig wasps largely determine where fig pollen is deposited. Most fig wasps show a high degree of host-plant specificity and are known from only a single fig tree species. However, in some cases wasps may be shared across closely related fig species (Cook and Segar, 2010). A single fig tree species has up to eight or more pollinator species, but most probably have one or a small number of associated pollinators.

The external morphology of fig wasps reflects variation in fig structure, particularly features of the ostiole and the length of the styles. Female pollinators need special adaptations especially on the head to pass through the ostiole as they enter the fig

(Liu et al., 2013). They have a head that has sunken membranous region that allows it to be deformed when entering the ostiole while protecting parts of the antennae that do not detach (Boucek et al., 1981). The ostiole length reflects head shape as there is a positive correlation between head shape and fig size (van Noort and Compton, 1996).

The style length of the flowers also influences the morphology of the fig wasps. *Ficus* species with longer styles of female flowers in monoecious figs are associated with longer ovipositor lengths of their particular pollinators, though monoecious fig pollinators in general have longer ovipositors than those associated with dioecious figs (Nefdt and Compton, 1996; Compton et al., 2010). Successful entrance into the fig cavity and subsequent oviposition does not guarantee that a female can gall the flowers where her eggs are laid. The galling process is poorly understood (Elias et al., 2018), but galled ovules and their pedicels enlarge rapidly. This is believed to be in response to chemicals released by the ovipositing females and may reflect competition for space inside the figs (Ghana et al., 2015a). The larvae may continue to produce gall-inducing chemicals during their development.

For a viable seed to be generated, pollen compatibility is important. For example in *Flourensia cernua* (Asteraceae), when non-compatible male pollen and female stigmas expressed the cognate S-allele, it will produce none or very low numbers of seed due to inhibition at the S-locus (Ferrer et al., 2009). The S-allele is self sterility gene where the function of this gene is to prevent self fertilisation (Ferrer et al., 2009). Non-compatible Pollen incompatibility seems to be poorly developed in fig trees, perhaps because heterospecific pollen is rarely encountered due to the specificity of the plants' vectors. However, when pollen of other species makes its way into figs, hybridisation and introgression can occur.

In the fig and fig wasp cases, the reproductive success of any hybrids depends on the ability of the foundress offspring to develop in their host plants and whether the hybrids can produce viable seeds (Ghana et al., 2015b). In experimental crosses, hybrids between two closely related but not sympatric species *Ficus montana* Blume and *Ficus asperifolia* Miquel were generated by the routine pollinators of F. *montana* (*Kradibia tentacularis* (Grandi). F1 male plants were sterile as the ovules failed to enlarge and the embryos failed to develop. Conversely, F1 females produced a lot of viable seeds. So it is possible for hybrid male figs to be unsuitable for reproduction of a particular

species of fig wasp, but hybrid female figs can have high reproductive success. In this species pair, at last, introgression between the *Ficus* species is therefore more likely to occur through entry into female than male plants.

4.1.4 Ficus deltoidea

Ficus deltoidea Jack. sometimes known as the mistletoe fig because of one of its growth forms, is one of the more commonly found fig trees in South East Asia. Its distribution includes Thailand, Indonesia and Malaysia (Cardellina, 2012). They are locally known as "Mas Cotek" by Malaysians because of having gold (in Malay gold means emas) coloured fine spots on the surface of the leaves (Mat et al., 2012). In Thailand some varieties are referred to as 'golden tongue', perhaps because of the orange-brown undersides to the leaves. Traditionally, *F. deltoidea* is widely used to cure a variety of diseases such as rheumatism, diabetes, toothache, headaches, colds and sore throats using the powdered roots and leaves (Bunawan et al., 2014). More recently it has been shown that *F. deltoidea* has anti-hyperglycaemic (Adam et al., 2012), anti-diabetic, anti-oxidant (Misbah et al., 2013) and wound-healing properties (Abdulla et al., 2010). Adam et al. (2007) stated that all parts of *F. deltoidea* have potential medicinal properties

F. deltoidea is dioecious, with separate male and female plants that either support fig wasp pollinators or produce seeds, respectively. Female figs of this group are exceptional insofar as they only contain small numbers of flowers. This allows individual seeds to be far larger than is normal for fig trees and is a presumed adaptation for an epiphytic lifestyle, as it means that young seedlings have more resources available initially. The small size of the figs suggests that their seeds are dispersed mainly by birds, but Orang Utans are the only published feeders on the figs (Shanahan et al., 2001). The *F. deltoidea* complex includes varieties that are commonly grown as ornamentals, usually as pot plants. They can thrive in dry atmospheres because of their thick cuticle, and small number of stomata both of which reduce water loss (Hao et al., 2016). *F. deltoidea* can be propagated from seed but many of them are propagated through cuttings and almost all the plants offered for sale by nurseries are female (Starr et al., 2003). Little is known about the *Blastophaga* species that pollinate *F. deltoidea* figs, but there are indications that different varieties may support different species of pollinators (Chapter 5).

4.1.5 F. deltoidea classification

Corner (1969) asserted that the F. deltoidea complex displayed most clearly the interplay of speciation and geographical distribution within the genus. F. deltoidea is currently placed in Subgenus Ficus, Section Ficus, subsection Frutescentiae (Berg and Corner, 2005). The subsection was further divided into two species groups containing about 25 species (Shi et al., 2006). The small F. deltoidea species group is equivalent to series Erythrogyneae of Corner (1960) and contains three species (F. deltoidea, F. oleifolia and F. kofmaniae), distributed in SE Asia. These species are true, short-rooted epiphytes (as opposed to hemi-epiphytic stranglers) or epilithic and when growing on the ground are usually found on nutrient poor soils (Berg and Corner, 2005). A characteristic feature is the small number of flowers in their female figs with sometimes as few as a single unusually large seed being produced per fig (Corner, 1969). In the first two species the presence of modified 'furcate' leaves on mature plants is another feature. Leaf morphology and anatomy are often used for identification of *F. deltoidea* varieties because they are particularly discriminative (Nur Fatihah et al., 2014), but leaf shape often varies greatly between young and mature plants. Trees in section Ficus and series Erythrogyneae also have ostiolar male flowers and reduced root systems (Corner, 1969).

F. deltoidea is highly variable in anatomy and this led Corner (1960) to recognise 13 varieties under *F. deltoidea*, namely var. *angustifolia*, var. arenaria, var. *bilobata*, var. *borneensis*, var. *deltoidea*, var. *intermedia* var. *kinabaluensis*, var. *kunstleri*, var. *lutescens*, var. *motleyana*, var. *oligoneura*, var. *peltata*, and var. *trengganuensis*. Given the presence of intermediate individuals between varieties and further variation within varieties, Berg (in (Kjellberg et al., 2005b) simplified the intraspecific classification of *F. deltoidea*. Two subspecies were recognised based mainly on leaf differences, namely *F. deltoidea deltoidea* and *F. deltoidea motleyana*. Most of the varieties described by Corner were placed under *F. deltoidea deltoidea* except for var. *motleyana* and var. *olioneura* that were placed under subspecies *F. deltoidea motleyana*. However, most studies of *F. deltoidea* have continued to use the more fine-grained classification of Corner (1960, 1969) and this classification is retained within this thesis.

4.1.6 F. deltoidea varieties

In the early classification of (Corner, 1969) he considered that the most primitive or basal variety was var. *motleyana*. This variety is predominantly in Borneo and

resembles *Ficus oleaefolia* that is also common in Borneo (and Sumatra). He also noted that the most widespread variety was var. *angustifolia* and that this variety showed the most reduction in leaf, fig and seed size. Falling next in terms of widespread distributions is var. *deltoidea* and this variety was suggested as a possible ancestor of var. *angustifolia* and var. *lutescens*. In Peninsular Malaysia, seven varieties are endemic namely var. *angustifolia*, var. *bilobata*, var. *deltoidea*, var. *intermedia*, var. *kunstleri*, var. *motleyana*, and var. *trengganuensis* while the other Malaysian varieties can be found in Sabah and Sarawak. The varieties are usually easily distinguished by looking at their leaf structure (Figure 4.1). The leaf forms of male and female plants are not distinguishable. The only way to know the gender of the trees is by opening the figs and looking at their inner structure.

All varieties show a presence of fine spots with a golden colour on the surface of the leaves and the vernacular name of Mas cotek is given in Peninsular Malaysia because of it. Morphological studies of *F. deltoidea* show that all characters portray a high variability among the varieties (Awang et al., 2013). This variation between and within species may be due to cross-pollination, sexual recombination and mutations (Corner, 1997). All the fig species in the *F. deltoidea* species group exhibit red tepals, numerous ostiolar male flowers, a reduced root system and large seeds (Corner, 1969). Among 40 morphological characters used in differentiating the varieties in *F. deltoidea*, the three most significant characters were found to be the shape and size of the leaves, the number and colour of the dots on the leaf's surface and fig structure (Awang et al., 2013). Some of the varieties have long spathulate leaves when saplings and these include var. *deltoidea*, var. *angustifolia*, var. *intermedia* and var. *motleyana* (Corner, 1969). In addition, a recent study by Mat et al., (2012), found different leaf forms between juvenile and mature individuals of var. *bilobata* and var. *trengganuensis*.

A tepal is present which encloses the ovary and their colour and structure varies between varieties and sexes. In female *F. deltoidea* the figs are particularly simple, because they contain unusually few pistillate female flowers. There were also sterile male flowers in the female figs. In the male figs, there are staminate flowers to produce the pollen and pistillate flowers to support development of fig wasp offspring (Kjellberg et al., 2014). Male figs of *F. deltoidea* are more typical of figs in general, and contain many more flowers than the female figs. In male figs, the developing galls in *F. deltoidea* is unique externally, being rugose and angular, unlike other *Ficus* with rounded galls (F. Kjellberg pers. Communication).



Figure 4.1 Differences in leaf form between seven endemic varieties of *Ficus deltoidea* in Peninsular Malaysia. From top left, var. *trengganuensis*, var. *kunstleri*, var. *intermedia*, var. *angustifolia*, var. *deltoidea*, var. *bilobata* and var. *motleyana*.

4.2 Objectives

The group of fig trees comprising *F. deltoidea* and its relatives is one of the most interesting within the whole genus, displaying apparently recent evolution, diversification and novel growth forms in and around the Sunda Shelf. The taxonomic and biological status of these varieties is unclear, as is the extent to which different varieties support different fig wasp pollinators. Until now, most published studies discussed the differences in the leaf forms and the chemical and pharmaceutical properties of the leaves, but no study has been conducted on the differences in fig anatomy between varieties and their relationship with their associated *Blastophaga* fig wasps.

Whether the varieties are actually distinct biological species is unclear. *F. deltoidea* often exhibit different states of leaf characters between the young and mature plants that has often led to misleading identifications of the varieties. Whether the figs of varieties are different is unknown. Leaf forms do not influence gene flow but fig morphology does (S. G. Compton pers. Communication).

This study aims aimed to provide biological evidence relating to whether some of the Malaysian varieties of *F. deltoidea* are likely to be distinct species. Specific questions were (1) are there morphological differences (in the fig and the fig wasp (size of figs, floral numbers, ovipositor length, femur length) between different populations of *F. deltoidea* var. *angustifolia* in Peninsular Malaysia? (2) Are there morphological differences between the figs and the fig wasps of this and other varieties of *F. deltoidea*? (3) are any differences in morphology in the fig wasps related to variation in their host figs?

4.3 Methods

4.3.1 Study varieties

Morphological comparisons were made between the figs produced by seven varieties of *F. deltoidea*. The varieties were var. *angustifolia* (Miq.), var. *deltoidea* Corner, var. *trengganuensis* Corner, var. *kunstleri* King, var. *bilobata* Corner, var. *motleyana* (Miq.) and var. *intermedia* Corner. All these varieties are native to Peninsular Malaysia, but natural populations of only three varieties were sampled.

4.3.2 Study sites and sampling methods

The samples of var. *angustifolia*, var. *deltoidea* and var. *trengganuensis* were obtained from natural wild fig trees growing as epiphytes in oil palm plantations in Peninsular Malaysia. The samples of var. *kunstleri*, var. *bilobata*, var. *motleyana* and var. *intermedia* were obtained from potted trees housed in the Terengganu Germplasm Collection (TGC) in Terengganu and planted trees housed at Batu Pahat, Johor. Details of these locations including related climate data were provided in Chapter 2. All the samples were collected to see the differences in the fig's characterisation of all varieties of *F. deltoidei*.

Fig development stage classification follows Galil and Eisikowitch (1968). B-phase is the receptive (female) phase when the figs are ready to be entered by pollinator fig wasps. D-phase is when the next generation of wasps are mature and ready to emerge from the galls (E phase). At each study site, B-phase, D or E phase and aborting figs were selected at random and collected into plastic bags. Since fruiting of all the trees was asynchronous, the figs at different phases were collected from each trees at the same time. The male fig collections were taken at early D phase where no fig wasps including the male had emerged from their galls. In female figs the equivalent early E phase was sampled. Five trees of each sex were sampled at each site.

For F. deltoidea var. angustifolia, four different oil palm plantations provided the samples. The plantations were located in Banting and Dengkil (Selangor state), Batu Pahat (Johor state) and Bagan Serai (Perak state). 5 B-phase male and female figs and 5 early D-phase male/early E-phase female figs were collected from each tree. The B-phase and D/E-phase figs were collected from the same five trees. A total of 400 figs from 20 male and 20 female trees were brought to the Lab. to be dissected (Table 1.1). The same sample sizes were obtained at single sites for populations of var. deltoidea (at Batu Pahat in Johor,) and var. trengganuensis (at Tembila, Terengganu), providing totals of 100 figs for each variety (Table 4.1). The fig trees of the remaining varieties were growing in pots and did not show any signs of pollination. Approximate B phase and the largest figs present were sampled. The maximum diameters of their figs were measured. Sample sizes varied according to availability (Table 4.2). Preliminary studies where figs were opened in the field allowed the likely developmental stage of figs to be assessed visually. Any sampled figs that were found to be at other stages of development when opened were rejected. Figs at the appropriate stages on each tree were sampled at random.

Variety	Date	Location	Male			Female		
			Trees	B phase figs	D phase figs	Trees	B phase figs	D phase figs
angustifolia	15.04.2017	Bagan Serai	5	25	25	5	25	25
angustifolia	03.11.2016	Batu Pahat	5	25	25	5	25	25
angustifolia	20.07.2016	Banting	5	25	25	5	25	25
angustifolia	20.07.2016	Dengkil	5	25	25	5	25	25
deltoidea	10.11.2016	Batu Pahat	5	25	25	5	25	25
trengganuensis	28.10.2016	Tembila	5	25	25	5	25	25

Table 4.1 The collections of male and female figs of var. *angustifolia*, var. *deltoidea*, and var. *trengganuensis*, at each study site.

Table 4.2 The collections of male and female figs of six varieties of *F. deltoidea*. (TGC = Terengganu Germplasm Collection).

Variety	Date	Location	Male			Female		
			Trees	В	D	Trees	В	D
				phase	phase		phase	phase
				figs	figs		figs	figs
bilobata	28.10.2016	TGC	5	25	25	1	5	5
kunstleri	10.11.2016	Batu Pahat	2	10	10	5	25	25
kunstleri	28.10.2016	TGC	2	10	10	-	-	-
intermedia	28.10.2016	TGC	1	5	5	-	-	-
motleyana	28.10.2016	TGC	-	-	-	5	25	25

4.3.3 Fig characterisations

The diameters of the figs were measured using Vernier callipers, and they were then dissected to record their contents. The diameters were measured at a right angle to the ostiole-pedicel axis. Later, pi is used to calculate the surface area. With B-phase (receptive) female figs, the parameters counted and measured were the number of female flowers, the style lengths of the female flowers, the number of tepals, the lengths of the tepals and the diameter. The same features were recorded in B-phase male figs, together with the number of male flowers present. Each male flower has two anthers.

Tepal and style lengths were measured using an image analyser. The style length measurements followed those described by Nefdt and Compton (1996) where the style length was taken as the distance from the top of the stigma to where the style reached the edge of the ovary. Ghana et al. (2017) noted that maximum style lengths in both male and female figs are reached at about the time the figs are receptive (B phase). In female figs all the available styles at or near B phase were measured. The maximum lengths of five randomly chosen tepals were measured in each fig. The diameters of pollinated figs were measured at late D- or early E- phases as before.

No pollinated figs of some varieties were available. The diameters of these varieties were measured in the same way using the biggest figs on the trees. D- phase male figs of var. *angustifolia*, var. *deltoidea* and var. *trengganuensis* were examined to record pollinator offspring numbers. The following were recorded: the number of galls that contained adult pollinator offspring, the number of empty (failed) galls, maximum gall length and the number of unused flowers. No E phase male figs (where an exit hole was present) were examined. Each gall was squashed gently between two microscope slides to allow the sex of the wasp inside to be recorded (if any were present). No non-pollinator fig wasps were present in the figs of any of the varieties. E phase female figs for the pollinated varieties were examined and the seed lengths and number of seeds produced were counted.

4.3.4 Fig wasp characterisations

The pollinators of three varieties of *F. deltoidea* were examined to see any differences between them (var. *angustifolia*, var. *deltoidea* and var. *trengganuensis*). Ovipositor lengths were measured using pollinators from var. *angustifolia* figs collected at Bagan

Serai, Banting, Batu Pahat ande Dengkil. Batu Pahat also provided the figs for var. *deltoidea* and Tembila provided the figs for var. *trengganuensis*. Measurements were made using 10 females from each of 5 D phase male figs collected from five different trees of each variety (n = 250 females from each variety). The figs from the same tree were placed together in a container covered by fine mesh to allow the pollinators to emerge naturally. Individuals were selected at random from those that emerged from the figs. Ovipositors were measured by placing the females in a drop of water on a glass slide under a cover slip and squashing with moderate pressure to displace the ovipositor sheaths without breaking the ovipositor. The measurements were conducted using image-analysis software, as before.

Hind femur length was used as an indicator of overall body size because it is the longest leg segment, readily distinguishable and easy to measure. Measurements were made as before. From each of the four var. *angustifolia* study sites, groups of about 5-10 D phase figs from five different trees at each site were collected and placed in containers as before. Femurs of 30 individuals from each site were measured (six from each tree). Similar measurements were made using 30 randomly sub-sampled female fig wasps reared from figs of var. *deltoidea* and var. *trengganuensis*. The wasps were obtained from five trees of each variety.

4.3.5 Statistical methods

For the variation between four populations of var. *angustifolia* and variation across all varieties of *F. deltoidea*, ANOVA followed by Tukey tests were conducted to examine differences in means between populations and between trees within populations if the data were countinuos and had a normal distribution. Count data was analysed using GLM with Poisson error. For non-normal distributions, Kruskal-Wallis tests were conducted followed by Mann-Whitney tests. If there was over-dispersion in Poisson distributions for count data, the data were analysed using quasi-Poisson errors. Correlation tests were conducted to examine relationships between pairs of variables within the same variety only in var. *angustifolia* because four study site provided the samples. Linear regression were used together with ANOVA to examine relationships across and within variables where site is used as a random factor. GLMs, ANOVAs correlation and regression were all performed in R studio (1.0.153) and SPSS Statistic 20.

4.4 Results

4.4.1 Variation among populations of var. angustifolia

4.4.1.1 Variation in female figs of var. angustifolia

Receptive (B phase) female figs were small, with a diameter of about 4 mm (Table 4.3). There was no significance difference in diameter between sites (Kruskal-Wallis, $\chi^2 = 5.06$, df = 3, P > 0.05) and trees (Kruskal-Wallis, $\chi^2 = 29.51$, df = 19, P > 0.05). After pollination, at E phase they reached maximum diameters of 7 or 8 mm (Table 4.3). A significant difference between sites was found (ANOVA, F = 10.50, df = 3, 96, P < 0.01) and between trees (ANOVA, F = 5.01, df = 19, 80, P < 0.01).

From 3 to 6 female flowers were present in the female figs of var. *angustifolia* (Table 4.4). There were significant differences in flower numbers between sites (Kruskal-Wallis, $\chi^2 = 11.35$, df = 3, P < 0.05) and trees (Kruskal-Wallis, $\chi^2 = 30.95$, df = 19, P < 0.05). Mann-Whitney tests showed that only the Banting site differed from other sites. Usually four tepals were present in each female flower (Table 4.4), but their numbers ranged from three to five. Kruskal-Wallis followed by Mann-Whitney Test tests found a significant difference in tepal numbers between the Banting and other sites (Kruskal-Wallis, $\chi^2 = 22.01$, df = 3, P < 0.01), reflecting the smaller number of flowers in the figs at this site.There were also significant differences in tepal numbers between trees (Kruskal-Wallis, $\chi^2 = 48.06$, df = 19, P < 0.01).

Table 4.3. The diameters of receptive phase and maximum diameters of female var. *angustifolia* figs from four sites in Peninsular Malaysia. Values with different superscripts in a column differed significantly in Tukey tests (P< 0.01).

	n	n	Receptive dia	meter (mm)	Maximum diameter (mm)		
Sites	(trees)	(figs)	Mean ± SD	Range	Mean ± SD	Range	
Bagan Serai	5	25	4.2 ± 0.02	3.9-4.5	7.1 ± 0.03^{a}	6.7-7.8	
Banting	5	25	4.3 ± 0.03	3.9-4.7	7.4 ± 0.04^{b}	6.7-8.3	
Batu Pahat	5	25	4.3 ± 0.02	4.0-4.7	7.2 ± 0.04^{a}	6.7-8.1	
Dengkil	5	25	4.3 ± 0.03	3.9-4.7	7.1 ± 0.04 ^a	6.7-7.8	

Table 4.4. The numbers of female flowers and tepals in female figs of var. *angustifolia* from four sites. Values with different superscripts in a column differed significantly in Mann-Whitney tests (P < 0.05).

	Female flower number		Tepal Nur	nber	No. of tepals per flower		
Sites	Mean ± SD	Range	Mean ± SD	Range	Mean ± SD	Range	
Bagan Serai	4.96 ± 0.61^{b}	4-6	20.36 ± 2.74 ^b	16-27	4.11 ± 0.30^{b}	3.60-4.67	
Banting	4.28 ± 0.79^{a}	3-6	16.48 ± 3.18 ^a	12-24	3.87 ± 0.43^{a}	3.25-4.80	
Batu Pahat	4.48 ± 1.00^{b}	4-6	19.24 ± 1.90 ^b	15-24	4.03 ± 0.33^{b}	3.33-4.75	
Dengkil	4.80 ± 0.76^{b}	4-6	18.96 ± 3.12 ^b	12-24	3.95 ± 0.14^{b}	3.75-4.40	
Between one and six seeds were present in the pollinated female figs (Table 4.5). The highest mean number of seeds was in figs from Batu Pahat. Kruskal-Wallis tests detected a significant difference in the numbers of seeds in figs between sites (χ^2 = 14.53, df = 3, P < 0.05) and between trees (Kruskal-Wallis, χ^2 = 37.99, df = 19, P < 0.05). The shortest and longest seeds were from figs in Dengkil measuring 3.54 mm and 4.43 mm respectively (Table 4.5). Seed lengths varied significantly between sites (ANOVA, F = 8.12, df = 3,196, P < 0.01) and specifically between Bagan Serai and the three other sites. Seed lengths also varied significantly between trees (ANOVA, F = 3.68, df = 19,180, P < 0.01).

Unused flowers represent non-pollinated, undeveloped flowers within figs that contained seeds and so had been entered by pollinators. More unused flowers were present in female figs from Bagan Serai than elsewhere (Kruskal-Wallis, $\chi^2 = 21.23$, df = 3, P < 0.01) (Table 4.6). Trees also differed significantly in the numbers of unused flowers present in their figs (Kruskal-Wallis, $\chi^2 = 42.94$, df = 19, P < 0.05). These differences were reflected in differences in pollination efficiency (the proportion of female flowers that developed into seeds) with figs from Bagan Serai having a higher proportion of unused flowers than elsewhere (Table 4.6).

Table 4.5. The numbers of seeds and their maximum lengths in female figs of var. *angustifolia* from four sites. Values with different superscripts in a column differed significantly (Seed number Mann-Whitney tests, Seed length Tukey tests P< 0.05).

Sites	n trees	Seed numb	ber	n trees (seeds)	Seed length (mm)		
	(ligs)	Mean ± SD	Range	(seeds)	Mean ± SD	Range	
Bagan Serai	5 (25)	3.44 ± 1.29 ^a	1-6	5 (50)	3.90 ± 0.16^{a}	3.72-4.35	
Banting	5 (25)	3.56 ± 0.89^{a}	1-5	5 (50)	4.03 ± 0.18^{b}	3.67-4.42	
Batu Pahat	5 (25)	4.48 ± 1.00^{b}	1-6	5 (50)	4.09 ± 0.24^{b}	3.59-4.21	
Dengkil	5 (25)	4.08 ± 1.22^{ab}	1-6	5 (50)	4.04 ± 0.23^{b}	3.54-4.43	

Table 4.6.The unused female flowers in female figs of var. *angustifolia* from four sites. Values with different superscripts in a column differed significantly (Mann-Whitney tests, P< 0.05).

	n	n	Unused female flowers		Proportion of unused flowers	
Sites	(trees)	(figs)	Mean ± SD	Range	Mean ± SD	Range
Bagan Serai	5	25	1.52 ± 1.19^{a}	0-4	0.31 ± 0.24^{a}	0 - 0.80
Banting	5	25	0.56 ± 0.96^{b}	0-4	0.11 ± 0.19^{b}	0 - 0.80
Batu Pahat	5	25	0.32 ± 0.90^{b}	0-4	0.06 ± 0.18^{b}	0 - 0.80
Dengkil	5	25	0.72 ± 0.98^{b}	0-3	0.15 ± 0.22^{b}	0 - 0.75

4.4.1.2 Between-population variation in male figs of var. angustifolia

Receptive (B phase) male figs were small with a diameter of about four mm (Table 4.7). There were no significant differences in diameter between sites (Kruskal-Wallis, $\chi^2 = 6.10$, df = 3, P > 0.05) and trees (Kruskal-Wallis, $\chi^2 = 19.14$, df = 19, P > 0.05). There were also no significant differences in maximum diameters at early E phase between sites (ANOVA, F = 2.62, df = 3, 96, P > 0.05) and between trees (ANOVA, F = 1.24, df = 19, 80, P > 0.05).

Male flowers only exist in male figs. They ranged between 21 to 37 per fig (Table 4.8). There were no significant differences in male flower numbers between sites (ANOVA, F = 1.72, df = 3, 96, P > 0.05), but differences between trees (ANOVA, F = 5.69, df = 19, 80, P < 0.05) (Table 4.8). There were more female than male flowers in the male figs. Figs from Dengkil had the most variation in female flower numbers, from 94 to 259 in different figs. There were significant differences between sites (ANOVA, F = 8.18, df = 3, 96, P < 0.01) and trees (ANOVA F = 12.56, df = 19, 80, P < 0.01) (Table 4.8).

Reflecting the large variation in female flower numbers in male figs at Dengkil, there was also large variation in the numbers of tepals present in the figs (Table 4.9). ANOVA test showed there were significant differences in tepal numbers between sites (F = 8.87, df = 3, 96, P < 0.01) and between trees (ANOVA, F = 6.92, df = 19, 80, P < 0.01).

	n	n	Receptive diar	neter (mm)	Maximum diameter (mm)		
Sites	(trees)	(figs)	Mean ± SD	Range	Mean ± SD	Range	
Bagan Serai	5	25	0.41 ± 0.03	0.37-0.45	0.71 ± 0.03	0.67-0.78	
Banting	5	25	0.42 ± 0.03	0.38-0.47	0.74 ± 0.04	0.67-0.83	
Batu Pahat	5	25	0.43 ± 0.02	0.38-0.44	0.72 ± 0.04	0.67-0.81	
Dengkil	5	25	0.41 ± 0.03	0.36-0.44	0.71 ± 0.04	0.67-0.78	

Table 4.7 The diameter of receptive male var. *angustifolia* figs and their maximum diameters of male figs at four sites.

Table 4.8 The numbers of male and female flowers in male figs of var. *angustifolia* from four different sites. Female flower numbers with different superscripts differed significantly between sites (Tukey tests, P < 0.01).

	n	n	Male flower n	umbers	Female flower numbers		
Sites	(trees)	(figs)	Mean ± SD	Range	Mean ± SD	Range	
Bagan Serai	5	25	29.88 ± 3.54	21-34	173.96 ±16.44°	150-209	
Banting	5	25	29.64 ± 3.49	23-35	142.76 ± 12.21ª	121-163	
Batu Pahat	5	25	31.20 ± 3.60	25-41	156.08 ± 16.61 ^b	130-185	
Dengkil	5	25	28.64 ± 5.17	21-37	176.08 ± 48.44 ^c	94-259	

Table 4.9 The number of tepals in male figs of var. *angustifolia* from four sites. Tepal numbers with different superscripts differed significantly between sites (Tukey tests, P< 0.01).

	n	n	Tepal numbers			
Sites	(trees)	(figs)	Mean (± SD)	Range		
Bagan Serai	5	25	645.24 ± 100.93 ^b	557-830		
Banting	5	25	499.92 ± 70.29 ^a	336-590		
Batu Pahat	5	25	538.28 ± 69.64ª	436-708		
Dengkil	5	25	648.04 ± 209.64 ^b	288-1052		

4.4.1.3 Variation in Blastophaga sp. pollinators reared from var. angustifolia

Ovule gall length is related to the size of the pollinators that emerge from the galls. Gall lengths ranged from 0.83-1.19 mm (Table 4.10). The longest mean length was recorded from Dengkil (1.03 mm) and the shortest from Batu Pahat (0.92 mm). There were significant differences in gall sizes between sites (ANOVA, F = 19.02, df = 3, 196, P < 0.01) and trees (ANOVA, F = 8.57, df = 19, 180, P < 0.05, Table 4.10). Despite this variation in gall sizes, the lengths of the hind femurs of reared female fig wasp offspring (a measure of body size) showed very little variation. They did not differ significantly in length between sites (ANOVA, F = 1.25, df = 3,116, P > 0.05) and trees (ANOVA, F = 1.03, df = 19, 100, P > 0.05).

The contents of five figs from five different trees at each site were recorded (Table 4.11). Total female flower numbers varied greatly, ranging between 94 and 259. Empty galls may have had eggs laid by the pollinator but if so they had failed to develop successfully. Unused flowers had not developed and showed no sign that they had been galled. Around one third to one half of these flowers were galled by the pollinators (Table 4.11). Galls that failed to support successful pollinator development were much less common, with usually two or fewer present per fig, but large numbers of failed galls were present in the 25 figs at Banting. There were significant differences between sites in the numbers of successful galls (ANOVA, F = 3.21, df = 3, 96, P < 0.05) and also between trees (ANOVA, F = 2.45, df = 19, 80, P < 0.05, Table 4.11). The number of empty galls did differ between sites (Kruskal-Wallis, $\chi^2 = 15.86$, df = 3, P < 0.05). Mann-Whitney tests between sites showed that only empty gall numbers from Banting differed from other sites (Table 4.11). Trees from Banting were the only ones to differ significantly in their numbers of empty galls (Kruskal-Wallis, $\chi^2 = 56.79$, df = 19, P < 0.01). The numbers of unused flowers did not differ between sites (Kruskal-Wallis, $\chi^2 = 21.25$, df = 3, P < 0.01) but differed between trees (Kruskal-Wallis, χ^2 = 42.94, df = 19, P < 0.05).

Flower utilisation in the male figs was very similar at the four sites (Figure 4.2). About 5% of the female flowers were empty galls at Banting, more than at the other sites. This difference comes because of a single tree which had figs with high numbers of empty galls.

Table 4.10 The gall lengths in male figs of var. *angustifolia* and pollinator hind femur lengths from four sites. Values for gall length with different superscripts differed significantly (Tukey tests, P< 0.01).

Sites	n figs	Gall lengt	h (mm)		Femur length (mm)		
	(n galls)	Mean (± SD) Range		n fig wasps	Mean (± SD)	Range	
Bagan Serai	10 (50)	0.94 ± 0.10^{a}	0.77-1.13	30	0.20 ± 0.01	0.18-0.21	
Banting	10 (50)	0.99 ± 0.06^{b}	0.89-1.12	30	0.20 ± 0.01	0.18-0.21	
Batu Pahat	10 (50)	0.92 ± 0.07^{a}	0.76-1.08	30	0.20 ± 0.01	0.18-0.21	
Dengkil	10 (50)	1.03 ± 0.10^{b}	0.83-1.19	30	0.20 ± 0.01	0.18-0.21	

Table 4.11 The female flower numbers and utilisation of female flowers in male figs of var. *angustifolia* at four sites. Values with different superscripts in a column differed significantly (Tukey and Mann-Whitney tests, P < 0.05).

Sites	Total female flowers		Total Successful Galls		Empty Galls		Unused flowers	
	Mean	Range	Mean	Range	Mean	Range	Mean	Range
	(± SD)		(± SD)		(± SD)		(± SD)	
Bagan Serai	173.96 ± 16.44°	150 - 209	62.32 ± 23.11 ^{ab}	24 -120	0.12 ± 0.44 ^a	0 - 2	111.52 ± 22.06 ^a	53- 146
Banting	142.76 ± 12.21ª	121- 163	46.76 ± 26.99 ^a	29 -129	7.6 ± 14.91⁵	0 - 46	88.4 ± 24.8 ^b	34- 122
Batu Pahat	156.08 ± 16.61 ^b	130 - 185	58.76 ± 25.29 ^{ab}	26 -123	0.24 ± 0.6 ^a	0 - 2	97.08 ± 22.32 ^b	66- 138
Dengkil	176.08 ± 48.44°	94 - 259	71.92 ± 38.24 ^b	19 -151	1.08 ± 3.24 ^a	0 -16	103.08 ± 48.86 ^b	12- 213



Figure 4.2 Female flower utilisation in male figs of var. *angustifolia* at four sites. Hashed bars = successful galls, open bars = empty galls and solid bars = unused flowers.

4.4.2 Variation between varieties of F. deltoidea

4.4.2.1 Characterisation of female figs

Receptive figs of var. *trengganuensis* had the longest diameters (up to 9.40 mm) and receptive figs of var. *bilobata* had the smallest diameters (as small as 2.7mm) (Table 4.12). Two groups of figs are distinguishable based on the size of their receptive figs (Figure 4.3). Varieties with small figs include var. *angustifolia*, var. *deltoidea*, var. *bilobata* and var. *motleyana* while those with large figs are var. *trengganuensis* and var. *kunstleri*.

The receptive and maximum diameters of figs from six varieties were measured (Table 4.12). Only one tree of var. *bilobata* was available. Within the other varieties, there were no significant differences between trees in diameter at receptivity (Kruskal-Wallis P > 0.05). Significant differences were present in the diameters of figs at receptivity from different varieties (Kruskal-Wallis, $\chi^2 = 143.62$, df = 5, P < 0.001). There were significant differences between the maximum diameters of figs belonging to different varieties using log transformed data (ANOVA, F = 93.92, df = 5,199, P < 0.001), but within most varieties there were also significant differences between trees (Tukey tests, Table 4.12).

The number of female flowers in female figs across all varieties was low (Table 4.13). Varieties with smaller figs tended to have fewer flowers (**Error! Reference source not found.**). Uniquely, all the figs from var. *bilobata* contained only one female flower. The largest number of female flowers was in figs of var. *trengganuensis*. The varieties all differed significantly from each other in their flower numbers except for var. *angustifolia* and var. *motleyana* (GLM, $\chi^2 = 100.43$, df = 5, P < 0.001) but between trees in the same variety, all varieties showed a significant difference (P< 0.05). Longest style lengths were also found in figs of var. *trengganuensis* and the shortest in figs of var. *deltoidea* (Table 4.13). Most varieties differed in mean style lengths (ANOVA on log transformed, F = 694.70, df = 5, 527, P < 0.001). Within the same sites, the trees also differed significantly in style lengths.

Table 4.12 The diameters of receptive female figs and the maximum observed diameters of female figs of six varieties of *F. deltoidea*. Values with different superscripts in a column differed significantly (Tukey and Mann-Whitney tests, P < 0.001). * indicates that values differed significantly between trees of the same variety.

Variety	n (trace)	n (fiara)	Receptive diameter (mm)		Mature diameter (mm)	
	(trees)	(figs)	Mean ± SD	Range	Mean ± SD	Range
angustifolia	20	100	4.3 ± 0.03^{b}	3.9-4.7	$7.2 \pm 0.04^{b*}$	6.7-0.83
deltoidea	5	25	4.6 ± 0.02^{b}	4.4-5.0	8.1 ± 0.05 ^{c*}	7.2-0.91
trengganuensis	5	25	7.5 ± 0.01^{d}	5.6-9.4	$13.6 \pm 0.08^{d*}$	12.1-1.49
kunstleri	5	25	6.0 ± 0.11°	5.0-8.6	$12.9 \pm 0.15^{d*}$	11.2-1.62
bilobata	1	5	2.9 ± 0.02^{a}	2.7-3.1	3.4 ± 0.01^{a}	3.2-0.36
motleyana	5	25	4.5 ± 0.03^{b}	4.0-5.1	7.5 ± 0.08^{bc}	6.5-0.88

Table 4.13 The number of female flowers and the lengths of styles in female figs of six varieties of *F. deltoidea*. Values with different superscripts in a column differed significantly (Tukey tests, P < 0.001). * indicates that values differed significantly between trees of the same variety.

Variety	n (trace)	n (fiara)	Female flower	Female flower number		Style lengths	
	(trees)	(figs)	Mean ± SD	Range (mm)	Mean ± SD	Range (mm)	
angustifolia	20	100	$4.72 \pm 0.71^{a*}$	3-6	$1.66 \pm 0.26^{b*}$	1.02-2.12	
deltoidea	5	25	8.48 ± 1.36 ^{c*}	6-11	$1.33 \pm 0.19^{a*}$	0.96- 1.64	
trengganuensis	5	25	23.88 ± 4.21 ^{e*}	18-32	$3.22 \pm 0.44^{e*}$	2.21-4.21	
kunstleri	5	25	19.8 ± 1.8 ^{d*}	17-23	$2.77 \pm 0.25^{d*}$	2.18-3.47	
bilobata	1	5	1 ± 0 ^b	1-1	1.49 ± 0.13^{ab}	1.30-1.66	
motleyana	5	25	$5.88 \pm 0.6^{a*}$	5-7	2.31 ± 0.35 ^{c*}	1.48-2.89	



Figure 4.3 The female flowers in receptive female figs of *Ficus deltoidea* varieties. A = var. *angustifolia*, B = var. *deltoidea*, C = var. *trengganuensis*, D = var. *bilobata*, E = var. *kunstleri* and F = var. *motleyana*). The figs had not been pollinated.



Diameter at receptivity and the number of female flowers in female figs of *F. deltoidea* varieties. (\blacksquare) = var. *angustifolia*, (\bullet) = var. *deltoidea*, (\blacktriangle) = var. *trengganuensis*, (\bigstar) var. *bilobata*, (\bullet) var. *kunstleri* and (\diamond) var. *motleyana*).

Because the tepals enclosed each ovary, their numbers were strongly correlated with flower numbers (Table 4.14). Var. *bilobata* recorded the lowest tepal number with a mean of 4.6 per fig while the most tepals were recorded in var. *trengganuensis* with 87.08. The ranges of numbers of tepal per single female flower were between 3 - 5. All varieties had significantly different numbers of tepals (GLM, $\chi^2 = 822.1$, df = 5, P < 0.001). The shape of the tepals in female figs varied little across the varieties but tended to have a rounder shape compared to the long slender tepals in male figs (Figure 4.4). There was some variation in tepal lengths between varieties (ANOVA, F = 123.40, df = 5, 568, P < 0.001) (Table 4.14).

Seed lengths could only be measured for the three varieties where pollination took place, namely var. *angustifolia*, var. *deltoidea* and var. *trengganuensis*. The seeds of var. *angustifolia* were the smallest, averaging 4.02 mm and the longest were from var. *trengganuensis* with a mean of 4.54 mm (Table 4.15). Seed lengths varied significantly between varieties (ANOVA, F = 118.90, df = 2, 297, P < 0.001). Within trees, only var. *angustifolia* showed differences in seed length (P < 0.05). This might be due to the seed lengths in var. *angustifolia* being taken from four different sites. A summary table of plant traits in male figs, which clarify the many ways in which the different varieties varied was depicted in (Figure)

Table 4.14 The number of tepals and tepal lengths in female figs of six varieties of *F. deltoidea*. Values with different superscripts in a column differed significantly (Tukey tests, P < 0.001). * indicates that values differed significantly between trees of the same variety.

Variety	n (tracc)	n (fiara)	Tepal numbers		Tepal lengths (mm)	
	(trees)	(figs)	Mean ± SD	Range	Mean ± SD	Range
angustifolia	20	100	18.76 ± 3.9 ^{b*}	12-24	1.34 ± 0.16 ^{c*}	0.89-1.69
deltoidea	5	25	36.36 ± 5.61 ^{c*}	27-46	$1.12 \pm 0.2^{b*}$	0.96-1.83
trengganuensis	5	25	87.08 ± 13.02 ^{d*}	70-110	1.59 ± 0.19 ^{a*}	1.17-1.95
kunstleri	5	25	75.64 ± 5.32 ^{e*}	62-84	1.49 ± 0.16 ^{a*}	1.04-1.87
bilobata	1	5	4.6 ± 0.55^{a}	4-5	1.59 ± 0.08^{a}	1.38-1.73
motleyana	5	25	24.08 ± 2.71 ^{b*}	20-28	$1.16 \pm 0.16^{b*}$	0.85-1.44

Table 4.15. Seed lengths in three varieties of *F. deltoidea*. Values with different superscripts in a column differed significantly (Tukey tests, P < 0.001). * indicates that values differed significantly between trees of the same variety.

Variety	n (trees)	n (figs)	n (seeds)	Seed length (mm)	
				Mean ± SD	Range
angustifolia	20	42	200	$4.02 \pm 0.22^{a*}$	3.54-4.43
deltoidea	5	10	50	4.24 ± 0.28^{b}	3.70-4.64
trengganuensis	5	10	50	4.54 ± 0.18 ^c	4.16-4.93



Figure 4.4 The tepals in female figs of *F. deltoidea* varieties. A = var. *angustifolia*, B = var. *deltoidea*, C = v ar. *trengganuensis*, D = var. *bilobata*, E = var. *kunstleri* and F = var. *motleyana*.

		Varieties						
Characteristic	angustifolia	deltoidea	trengganuensis	bilobata	kunstleri	motleyana		
Receptive diameter (mm)	0.43 <u>+</u> 0.04	0.46 <u>+</u> 0.02	0.75 <u>+</u> 0.012	0.29 <u>+</u> 0.02	0.6 <u>+</u> 0.11	0.45 <u>+</u> 0.03		
Maximum diameter (mm)	0.71 <u>+</u> 0.04	0.81 <u>+</u> 0.05	1.36 <u>+</u> 0.08	0.34 <u>+</u> 0.01	1.29 <u>+</u> 0.15	0.75 <u>+</u> 0.08		
No. of female flowers	4.72 <u>+</u> 0.71	8.48 <u>+</u> 1.36	23.88 <u>+</u> 4.21	1 <u>+</u> 0	19.8 <u>+</u> 1.8	5.88 <u>+</u> 0.6		
No. of tepals	18.76 <u>+</u> 3.9	36.36 <u>+</u> 5.61	87.08 <u>+</u> 13.02	4.6 <u>+</u> 0.55	75.64 <u>+</u> 5.32	24.08 <u>+</u> 2.71		
Style length (mm)	1.66 <u>+</u> 0.26	1.33 <u>+</u> 0.19	3.22 <u>+</u> 0.44	1.49 <u>+</u> 0.13	2.77 <u>+</u> 0.25	2.31 <u>+</u> 0.35		
Tepal length (mm)	1.34 <u>+</u> 0.16	1.12 <u>+</u> 0.2	1.59 <u>+</u> 0.19	1.59 <u>+</u> 0.08	1.49 <u>+</u> 0.16	1.16 <u>+</u> 0.16		
No. of seeds	3.93 <u>+</u> 1.17	6.68 <u>+</u> 2.08	16.16 <u>+</u> 7.34	0.00	0.00	0.00		

Table 4.16 Summary table of plant characteristics (mean \pm SD) in the female fig in six varieties of *F.deltoidea*.

4.4.2.2 Male Figs Characterisation

There was significant variation in the diameters of receptive figs (ANOVA on log transformed, F = 240.00, df = 5, 196, P < 0.001) and the maximum sizes of the figs (ANOVA on sqrt transformed, F = 1416.00, df = 5, 196, P < 0.001) produced by male plants of the six varieties of *F. deltoidea* (Table 4.16). Var. *bilobata* had the smallest receptive figs and var. *trengganuensis* the largest. Most of the pairwise comparisons showed a highly significant difference between varieties but no significant differences within trees of the same variety. Variation between trees of var. *intermedia* could not be assessed as only one tree was available. The aborting figs of var. *bilobata* and var. *trengganuensis* again had the smallest and the biggest maximum diameters respectively. Most varieties varied in the maximum size of their aborting figs (Table 4.16). Only two varieties showed significant differences between trees from the same variety.

The male flowers in male figs were consistently located near the ostiole while the female flowers were present in the basal part of the figs (Figure 4.5). The number of female flowers in male figs across all varieties was high compared to the female figs where varieties with smaller figs tended to have fewer flowers (Figure)The most male flowers were in figs of var. *kunstleri* and the least in var. bilobata (Table 4.17). The numbers of flowers varied significantly between (GLM, χ^2 = 119.57, df = 5, P < 0.001). Within a variety, significant differences were found between trees (except for var. deltoidea). Female flower numbers were lowest in var. *bilobata* (range = 24-69) and highest in var. *kunstleri* (range = 427–1362) (Table 4.17). There were significant differences overall (GLM, χ^2 = 156.3, df = 5, P < 0.001) with most combinations differ significantly. Anther to ovule ratios were highly variable between figs of the same variety but are consistent with all varieties the being passively pollinated (



Diameter at receptivity and the number of female flowers in male figs of *F. deltoidea* varieties. (\blacksquare) = var. *angustifolia*, (\bullet) = var. *deltoidea*, (\blacktriangle) = var. *trengganuensis*, (\bigstar) var. *bilobata*, (\bullet) var. *kunstleri* and (\triangledown) var. *intermedia*).

Table 4.18).

Table 4.16 The minimum and maximum diameters of male figs of six varieties of *F. deltoidea*. Values with different superscripts in a column differed significantly (Tukey tests, P < 0.001). * indicates that values differed significantly between trees of the same variety.

Variety	n (troop)	n (figo)	n Receptive diameter (mm)		Maximum diameter (mm)	
	(liees)	(ligs)	Mean ± SD	Range	Mean ± SD	Range
angustifolia	20	100	0.42 ± 0.03^{b}	0.37-0.47	0.63 ± 0.03^{b}	0.55-0.69
deltoidea	5	25	$0.47 \pm 0.02^{c*}$	0.42-0.49	$0.84 \pm 0.04^{\circ}$	0.76-0.90
trengganuensis	5	25	0.61 ± 0.07^{d}	0.52-0.80	1.31 ± 0.08 ^e	1.20-1.42
kunstleri	4	20	$0.64 \pm 0.08^{d*}$	0.51-0.81	$1.18 \pm 0.06^{d*}$	1.10-1.28
bilobata	5	25	0.30 ± 0.02^{a}	0.27-0.33	$0.39 \pm 0.04^{a*}$	0.34-0.49
intermedia	1	5	0.41 ± 0.01^{b}	0.40-0.42	0.58 ± 0.04^{b}	0.55-0.62

Table 4.17 The numbers of male and female flowers in six varieties of *F. deltoidea*. Values with different superscripts in a column differed significantly (Tukey tests, P < 0.001). * indicates that values differed significantly between trees of the same variety.

Variety	n (trace)	n (fiara)	n Male flower number		Female flower number		
	(trees)	(figs)	Mean ± SD	Range	Mean ± SD	Range	
angustifolia	20	100	29.84 ± 4.06 ^{c*}	20-41	162.22 ± 30.43 ^{a*}	94-259	
deltoidea	5	25	50.41 ± 5.65^{a}	42-61	157.41 ± 26.70 ^{a*}	108-239	
trengganuensis	5	25	143.68 ± 26.00 ^{b*}	91-188	828.48 ± 123.22 ^b	526- 1024	
kunstleri	4	20	176.75 ± 61.49 ^{b*}	79-285	874.90 ± 327.72 ^{b*}	427- 1362	
bilobata	5	25	13.64 ± 1.89 ^{d*}	11-17	46.08 ± 12.04 ^{c*}	24-69	
intermedia	1	5	49.80 ± 6.22^{a}	42-57	77.00 ± 6.36^{d}	71-86	



Figure 4.5 The location of male and female flowers in the male figs of *Ficus* deltoidea complex. A = var. angustifolia, B = var. deltoidea, C = var. trengganuensis, D = var. bilobata, E = var. intermedia and F = var. kunstleri.



Diameter at receptivity and the number of female flowers in male figs of *F. deltoidea* varieties. (\blacksquare) = var. *angustifolia*, (\bullet) = var. *deltoidea*, (\blacktriangle) = var. *trengganuensis*, (\bigstar) var. *bilobata*, (\bullet) var. *kunstleri* and (\triangledown) var. *intermedia*).

Variety	n (trees)	n (figs)	Ratio of anthers to ovules	
			Mean ± SD	Range
angustifolia	20	100	0.19 ± 0.03	0.14-0.26
deltoidea	5	25	0.31 ± 0.05	0.24-0.46
trengganuensis	5	25	0.18 ± 0.04	0.12-0.25
kunstleri	4	20	0.21 ± 0.04	0.16-0.28
bilobata	5	25	0.31 ± 0.09	0.20-0.58
intermedia	1	5	0.65 ± 0.11	0.59-0.79

Table 4.18 The ratio of anthers to ovules in male figs of varieties of *F. deltoidea*.

Similar to the female figs, tepals enclosed the ovules in the male figs (Figure 4.6). The number of tepals present reflected ovary numbers with var. *kunstleri* and var. *bilobata* again recording the highest and the lowest numbers of tepals respectively (Table 4.19). There was a significant difference across all varieties (GLM, $\chi^2 = 161.33$, df = 5, P < 0.001) but most pairs did not have different tepal numbers. With tepal lengths the pattern was the same, with var, *kunstleri* figs having the longest tepals and var. *bilobata* the shortest (Table 4.19). Tepal lengths varied significantly (ANOVA on log transformed data, F = 485.50, df = 5, 1044, P < 0.001) and there were differences between most pairs of varieties (Tukey tests) as well as between trees of the same variety.

The style lengths of female flowers in male figs ranged from 0.09 mm – 0.29 mm (Table 4.20). The shortest mean style lengths were in var. *bilobata* figs while var. *trengganuensis* figs had the longest mean style length. Most varieties differed in style lengths (ANOVA, F = 1768.00, df = 5, 1044, P < 0.001). Gall length provides an indirect measure of fig wasp body size. Gall lengths were longest in var. *deltoidea* figs with and the shortest in var. *angustifolia* (Table 4.21). There were significant differences in the sizes of galls in the figs of the three varieties (ANOVA, F = 181.20, df = 2, 297, P < 0.001). Tukey tests showed that gall sizes in var. *angustifolia* and var. *trengganuensis* did not differ significantly unlike the other two combinations. Within varieties, only var. *trengganuensis* did not show any significant variation between trees. A summary table of plant traits in male figs, which clarify the many ways in which the different varieties varied was depicted in (Table)



Figure 4.6 The tepals in male figs of varieties in the *Ficus deltoidea* complex. (A = var. *angustifolia*, B = var. *deltoidea*, C = var. *trengganuensis*, D = var. *bilobata*, E = var. *kunstleri* and F = var. *intermedia*).

Table 4.19 The numbers and lengths of tepals in male figs of six varieties of *F. deltoidea*. Values with different superscripts in a column differed significantly (Tukey tests, P < 0.001). * indicates that values differed significantly between trees of the same variety.

Variety	n	n "	Tepal Nu	Imbers	Tepal lengtl	ns (mm)
	(trees)	(figs)	Mean ± SD	Range	Mean ± SD	Range
angustifolia	20	100	582.87 ± 140.69 ^{a*}	336-1052	$0.48 \pm 0.10^{b*}$	0.26-0.74
deltoidea	5	25	620.74 ± 111.39ª	399-866	0.81 ± 0.12 ^{a*}	0.33-1.02
trengganuensis	5	25	3065.44 ± 644.342 ^b	1976-4006	0.68 ± 0.11 ^{c*}	0.43-0.99
kunstleri	4	20	3310 ± 1268.68 ^{b*}	1257-4988	0.91 ± 0.14 ^{d*}	0.54-1.19
bilobata	5	25	155.2 ± 44.14 ^{c*}	88-257	0.34 ± 0.04 ^{e*}	0.21-0.44
intermedia	1	5	296.8 ± 16.84 ^d	276-304	$0.54 \pm 0.09^{a*}$	0.33-0.73

Table 4.20 The lengths of the styles of female flowers in male figs of six varieties of *F. deltoidea*. Values with different superscripts in a column differed significantly (Tukey tests, P < 0.001). * indicates that values differed significantly between trees of the same variety.

Variety	Style lengths (mm)						
	n (trees)	n (figs)	n (flowers)	Mean ± SD	Range		
angustifolia	5	25	250	0.20 ± 0.05^{b}	0.17-0.24		
deltoidea	5	25	250	$0.15 \pm 0.02^{a*}$	0.12-0.19		
trengganuensis	5	25	250	0.24 ± 0.01 ^{c*}	0.21-0.29		
kunstleri	4	20	100	0.15 ± 0.01 ^{a*}	0.13-0.19		
bilobata	3	15	150	0.12 ± 0.06^{d}	0.09-0.13		
intermedia	1	5	50	0.18 ± 0.01 ^e	0.15-0.21		

Table 4.21 The gall lengths of female flowers in three varieties of *F. deltoidea*. Values with different superscripts in a column differed significantly (Tukey tests, P < 0.001). * indicates that values differed significantly between trees of the same variety.

Variety	Gall lengths (mm)							
	n (trees)	n (trees) n (figs) n (galls) Mean ± SD Range						
angustifolia	20	20	200	0.96 ± 0.11 ^{a*}	0.7-1.13			
deltoidea	5	5	50	1.27 ± 0.13 ^{b*}	1.07-1.59			
trengganuensis	5	5	50	1.00 ± 0.05^{a}	0.94-1.13			

Table 4.23 Summary table of plant characteristics (mean \pm SD) in the male fig	JS
of six varieties of <i>F. deltoidea.</i>	

	Varieties						
Characteristic	angustifolia	deltoidea	trengganuensis	bilobata	kunstleri	intermedia	
Receptive diameter (mm)	0.41 <u>+</u> 0.04	0.47 <u>+</u> 0.02	0.61 <u>+</u> 0.07	0.30 <u>+</u> 0.02	0.64 <u>+</u> 0.08	0.41 <u>+</u> 0.01	
Maximum diameter (mm)	0.62 <u>+</u> 0.04	0.84 <u>+</u> 0.04	1.31 <u>+</u> 0.08	0.39 <u>+</u> 0.04	1.18 <u>+</u> 0.06	0.58 <u>+</u> 0.04	
No. of male flowers	29.84 <u>+</u> 4.06	50.41 <u>+</u> 5.65	143.68 <u>+</u> 26.00	13.64 <u>+</u> 1.89	176.75 <u>+</u> 61.49	49.80 <u>+</u> 6.22	
No. of female flowers	162.22 <u>+</u> 30.43	163.59 <u>+</u> 26.70	831.88 <u>+</u> 123.22	46.08 <u>+</u> 12.04	874.90 <u>+</u> 327.72	77.00 <u>+</u> 6.36	
No. of galls	59.94 <u>+</u> 29.96	144.59 <u>+</u> 28.08	182.56 <u>+</u> 111.47	-	-	-	
No. of empty galls	2.26 <u>+</u> 8.14	1.30 <u>+</u> 1.90	24.60 <u>+</u> 19.49	-	-	-	
No. of unused female flowers	100.02 <u>+</u> 32.23	9.74 <u>+</u> 10.46	621.32 <u>+</u> 131.50	46.08 <u>+</u> 12.04	874.90 <u>+</u> 327.72	77.00 <u>+</u> 6.36	
No. of tepals	582.87 <u>+</u> 140.69	620.74 <u>+</u> 111.39	3065.44 <u>+</u> 644.342	155.2 <u>+</u> 44.14	3310 <u>+</u> 1268.68	296.8 <u>+</u> 16.84	
Style length (mm)	0.20 <u>+</u> 0.05	0.15 <u>+</u> 0.02	0.24 <u>+</u> 0.01	0.12 <u>+</u> 0.06	0.15 <u>+</u> 0.01	0.18 <u>+</u> 0.01	
Tepal length (mm)	0.48 <u>+</u> 0.10	0.81 <u>+</u> 0.12	0.68 <u>+</u> 0.11	0.34 <u>+</u> 0.04	0.91 <u>+</u> 0.14	0.54 <u>+</u> 0.09	
Gall length (mm)	0.96 <u>+</u> 0.11	1.27 <u>+</u> 0.13	1.00 <u>+</u> 0.05	-	-	-	

4.4.2.3 Fig wasp Characterisation

Fig wasps reared from figs of var. *angustifolia*, var. *deltoidea* and var. *trengganuensis* were the only pollinators available. Their ovipositor lengths overlapped (Table 4.22) but the mean length of the ovipositors of pollinators reared from var. *angustifolia* was significantly shorter than those reared from the remaining varieties (ANOVA, F = 1076, df = 2, 747, P < 0.001) (Figure 1.6). There were significant differences between the mean ovipositor lengths of wasps that emerged from different trees belonging to var. *trengganuensis* (ANOVA, F = 5.73, df = 4, 245, P < 0.01), but not different trees belonging to the other varieties (var. *angustifolia* (ANOVA, F = 1.00, df = 4, 245, P > 0.05) and var. *deltoidea* (ANOVA, F = 1.75, df = 4, 245, P > 0.05).

Significant differences were present in the hind femur lengths between pollinators of the three varieties (Kruskal-Wallis, $\chi^2 = 118.15$, df = 2, P < 0.001). Pollinators from var. *angustifolia* were the smallest and those from var. *trengganuensis* the largest (Table 4.23). Within-variety differences in femur length from different trees were not significant with var. *angustifolia* (ANOVA, F = 0.75, df = 19, 100, P > 0.05), var. *deltoidea* (ANOVA, F = 0.69, df = 4, 25, P > 0.05) and var. *trengganuensis* (ANOVA, F = 1.49, df = 4, 25, P > 0.05). The ratios of ovipositor to femur lengths (Figure 1.8) were very similar in the fig wasps reared from the three varieties.

The strong relationship between femur length and ovipositor length present is very clear when varieties are compared, Linear regression, $R^2 = 0.61$, F = 282.96, df = 178, P < 0.001). Within varieties, there was no relationship between the femur length and the ovipositor length of the female pollinator.

Table 4.22 The ovipositor lengths of female fig wasps reared from three varieties of *F. deltoidea*. Values with different superscripts in a column differed significantly (Tukey tests, P < 0.001). * indicates that values differed significantly between trees of the same variety.

Variety	Ovipositor lengths (mm)						
	n (trees)	n (trees) n (figs) n (wasps) Mean ± SD Rang					
angustifolia	5	25	250	0.44 ± 0.02^{a}	0.40-0.49		
deltoidea	5	25	250	0.54 ± 0.03^{b}	0.48-0.61		
trengganuensis	5	25	250	0.51 ± 0.02 ^{c*}	0.44-0.55		

Table 4.23 The femur lengths of female fig wasps reared from three varieties of *F. deltoidea*. Values with different superscripts in a column differed significantly (Mann-Whitney, tests, P < 0.001).

Variety	Femur lengths (mm)						
	n (trees)	n (trees) n (figs) n (wasps) Mean ± SD Ran					
angustifolia	20	20	200	0.20 ± 0.01^{a}	0.18-0.21		
deltoidea	5	5	50	0.25 ± 0.01 ^b	0.24-0.26		
trengganuensis	5	5	50	0.22 ± 0.01°	0.21-0.24		



The relationship between femur length and ovipositor length of female pollinators in *F. deltoidea* varieties. (\blacksquare) = var. *angustifolia*, (\bullet) = var. *deltoidea*, (\blacktriangle) = var. *trengganuensis*, (\bigstar).

trengganuensis

Variety	n (femur)	n (ovi)	Pearson's r	P value
angustifolia	120	120	0.002	0.98
deltoidea	30	30	0.225	0.23

25

0.175

0.36

5

Within-variety relationships between the femur length and ovipositor length of female pollinators using Pearson correlation.

4.4.3 Fig and fig wasp inter-relationships

4.4.3.1 Style length and ovipositor length relationships

Measurements of the style lengths in male figs of the three varieties and the ovipositor lengths of their respective pollinators showed that all the ovipositors were long enough to reach all the ovaries of the female flowers in all varieties (Figure 4.7). This meant that even the shortest ovipositor length (a 0.4 mm ovipositor from *Blastophaga* sp. var. *angustifolia*) was capable of penetrating the full length of even the longest styles of female flowers in male figs of all six varieties of *F. deltoidea* (Figure 4.8). On the other hand, none of the ovipositors were able to reach to the end of the much longer styles in any of the female figs and the ovipositors of the fig wasps showed that (Figure 4.9).



Figure 4.7 Hind femur lengths (open bars), ovipositor lengths (hashed bars) and ratios of ovipositor to femur lengths (solid bars) of fig wasps reared from three varieties of *F. deltoidea*. Values with different superscripts differed significantly (Mann-Whitney tests, P < 0.001).



Figure 4.8 A comparison between ovipositor lengths (open bars) and style lengths (solid bars) of female flowers in female figs of three varieties of *F. deltoidea* (Tukey tests, P < 0.001).



Figure 4.9 A comparison between ovipositor lengths (open bars) and style lengths (solid bars) of female flowers in male figs of three varieties of *F. deltoidea*. Values with different superscripts differed significantly (Tukey tests P < 0.001).



The relationship between the ovipositor length of *Blastophaga* sp. (solid bars) and style lengths in male figs (open bars) and female figs (hashed bars) of *F. deltoidea* var. *angustifolia*.



The relationship between the ovipositor length of *Blastophaga* sp. (solid bars) and style lengths in male figs (open bars) and female figs (hashed bars) of *F. deltoidea* var. *deltoidea*.



The relationship between the ovipositor length of *Blastophaga* sp. (solid bars) and style lengths in male figs (open bars) and female figs (hashed bars) of *F. deltoidei* var. *trengganuensis*.



Figure 4.10 The relationship between the ovipositor length of *Blastophaga* sp. (solid bars) and style lengths in male figs of *F. deltoidea* (open bars). A = var. *angustifolia*, B = var. *deltoidea*, C = var. *trengganuensis*.

4.4.4 Flower utilisation and pollination

All three varieties had more than 50% of the flowers in female figs pollinated (Figure 4.11). Var. *angustifolia* had 83.26% of its flowers producing seeds followed by var. *deltoidea* (78.77%) and var. *trengganuensis* (67.67%). The small figs of var. *angustifolia* and var. *deltoidea* contained few female flowers compared to the big figs in var. *trengganuensis*. They produced correspondingly fewer seeds (Table 4.24, GLM, $\chi^2 = 36.72$, df = 2, P < 0.001). Within varieties, only var. *deltoidea* did not differ significantly between trees in terms of seed numbers.

Differences between varieties in the contents of male figs are summarised in Figure 4.12. Male figs contained many more female flowers than female figs and differences in the numbers of flowers galled and fig wasp offspring generated by the figs reflected in the differences in flower numbers between varieties (Table 4.25, Figure 4.13). There was a significant difference in the total fig wasp offspring generated in the three varieties (GLM, $\chi^2 = 34.38$, df = 2, P < 0.001). This resulted from fewer offspring in the var. *angustifolia* figs than in the figs of the other varieties (Mann-Whitney tests).

Not all the galled flowers supported successful pollinator development. These formed empty galls, which represented only a small proportion of the total galled flowers (Table 4.25). The number of empty galls varied significantly between varieties (GLM, $\chi^2 = 349.7$, df = 2, P < 0.001) because var. *trengganuensis* figs contained many more empty galls than the other varieties (Mann-Whitney tests). Within varieties, the numbers of empty galls varied between trees of var. *angustifolia* were significantly different (Kruskal-Wallis, $\chi^2 = 56.77$, df = 19, P < 0.001).

The mean number of male offspring per fig varied from about 10 to 50 across the different varieties (Table 4.25). The number of male fig wasp offspring differed between varieties (GLM, $\chi^2 = 261.01$, df = 2, P < 0.001). Within varieties, only var. *angustifolia* did not also have a significant difference in male numbers between trees (Table 4.25). The mean numbers of female offspring per fig varied from about 50 to 135 in the three varieties. Significant differences were present in the numbers of female offspring in the figs of var. *angustifolia* compared with the other two varieties (GLM, $\chi^2 = 589.00$, df = 2, P < 0.001). Only var. *deltoidea* did not also have significant between-trees differences in the numbers of females present (Table 4.25).

The numbers of unused female flowers differed significantly between all combinations of the three varieties (Poisson GLM, $\chi^2 = 18155.11$, df = 2, P < 0.001). Within each variety, only trees of var. *angustifolia* differed significantly (ANOVA, F = 3.64, df = 19, 80, P < 0.001). The mean sex ratio (proportion of male offspring) was greatest in var. *trengganuensis* (0.29 ± 0.21) (Mean ± SD) followed by var. *angustifolia* (0.18 ± 0.19) and var. *deltoidea* (0.16 ± 0.07) (Figure 4.14) and they differ significantly (Kruskal-Wallis, $\chi^2 = 9.02$, df = 2, P < 0.05).



Figure 4.11. The number of female flowers in female figs (open bars) and the number of seeds generated (solid bars).

Table 4.24 The number of seeds in three figs of three varieties of *F. deltoidea*. Values with different superscripts differed significantly (Tukey Tests P< 0.05). * indicates that values differed significantly between trees of the same variety.

Variety	Number of seeds (mm)						
	n (trees) n (figs) Mean ± SD Range						
angustifolia	20	100	3.93 ± 1.17 ^{a*}	1-6			
deltoidea	5	25	6.68 ± 2.08^{b}	3-10			
trengganuensis	5	25	16.16 ± 7.34 ^{c*}	2-28			



Figure 4.12 A comparison of female flower utilisation by fig wasps in male figs of three varieties of *F. deltoidea*. Open bars = male offspring, grey bars = female offspring, black bars = empty galls, hashed bars = unused flowers.
Table 4.25 The contents of male figs (Mean \pm SD) of three varieties of *F. deltoidea*. Values with different superscripts in a row differed significantly (tests vary, see above, P< 0.001). * indicates values differed significantly between trees within the same variety.

Contents/Variety	angustifolia	deltoidea	trengganuensis
Female offspring	49.68 ± 26.34 ^{a*}	122.70 ± 27.30 ^b	133.04 ± 99.15 ^{b*}
Range	0-114	70-174	31-299
Male offspring	10.26 ± 10.16 ^a	23.67 ± 11.25 ^{b*}	49.52 ± 48.40 ^{c*}
Range	0-51	3-56	5-161
Total offspring	59.94 ± 29.96 ^{a*}	146.37 ± 29.90 ^b	182.56 ± 111.47 ^{b*}
Range	18-151	73-203	59-433
Empty galls	2.26 ± 8.14 ^{a*}	1.30 ± 1.95 ^a	24.60 ± 19.49^{b}
Range	0-46	0-7	9-104
Unused flowers	100.02 ± 32.23 ^{a*}	9.74 ± 10.46 ^b	621.32 ± 131.50°
Range	12-213	0 - 37	353-797
Total female flowers	162.22 ± 30.43 ^a	157.41 ± 26.70 ^{ab}	828.48 ± 123.22 ^b
Range	94-259	108-239	526-1024



Figure 4.13 Female flower numbers in male figs of three varieties of *F. deltoidea* (open bars) and the numbers of flowers galled (offspring + empty galls) by *Blastophaga* foundresses (solid bars).



Figure 4.14 The offspring sex ratios in relation to total offspring in three varieties of *Ficus deltoidea*. (Mann-Whitney test, P < 0.05).

4.4.5 Inflorescence design in closely related fig trees: The effects of having different numbers of flowers in the figs

4.4.5.1 Relationship between fig flower numbers and other characters

The strong relationship between maximum diameter of female figs and the number of female flowers present is very clear when varieties are compared (



Figure 4.15, Linear regression, $R^2 = 0.84$, F = 1091.87, df = 204, P < 0.001). Within varieties, there was no relationship between the numbers of flowers in a female fig and the maximum diameter they reached (Table 4.26).

In the male figs, there was again a strong relationship between the maximum diameter of the figs and the number of female flowers (Figure 4.16, Linear regression, $R^2 = 0.75$, F = 610.52, df = 201, P < 0.001). Similar result insofar with female figs for the within varieties, where there was no relationship between the numbers of flowers in a male fig and the maximum diameter they reached (Table 4.27).



Figure 4.15 The relationship between maximum diameter of female figs and the number of female flowers in *F. deltoidea* varieties. (\blacksquare) = var. *angustifolia*, (\bullet) = var. *deltoidea*, (\blacktriangle) = var. *trengganuensis*, (\bigstar) var. *bilobata*, (\bullet) var. *kunstleri* and (\blacklozenge) var. *motleyana*).

Table 4	4.26	With	in-va	ariety	rela	tio	nships	between	the	maximu	m di	amete	r of
female	figs	and	the	numb	oers	of	female	flowers	they	contain	using	g Pear	son
correlat	tion.												

Variety	n (trees)	n (figs)	Pearson's r	P value
angustifolia	20	100	-0.14	0.15
deltoidea	2	27	0.08	0.70
trengganuensis	5	25	-0.23	0.27
kunstleri	5	25	-0.31	0.13
motleyana	5	25	0.27	0.19



Figure 4.16 The relationship between maximum diameter and the number of female flowers in male figs in *F. deltoidea* varieties. (\blacksquare) = var. *angustifolia*, (\bullet) = var. *deltoidea*, (\blacktriangle) = var. *trengganuensis*, (\bigstar) var. *bilobata*, (\bullet) var. *kunstleri* and (∇) var. *intermedia*).

Table 4.27 Within-variety	relationships betv	veen the maximu	um diameter o	of male
figs and the numbers of	female flowers the	y contain using I	Pearson corre	elation.

Variety	n (trees)	n (figs)	Pearson's r	P value
angustifolia	20	100	0.05	0.64
deltoidea	2	27	0.24	0.24
trengganuensis	5	25	-0.16	0.43
kunstleri	4	20	0.02	0.92
bilobata	5	5	-0.24	0.18
intermedia	5	25	-0.40	0.51

When figs of all the varieties are considered together, there was a strong relationship between numbers of male and female flowers in male figs (Figure 4.17, Linear regression, $R^2 = 0.90$, F = 1827.06, df = 201, P < 0.001). The relationship between numbers of male and female flowers in male figs was not strong, but they were positively correlated in two varieties (Table 4.28). The ratio of male to female flowers numbers was exceptionally high in var. *intermedia*, where more male than female flowers were present (Table 4.29). The rest typically had far more female than male flowers, but one fig of var. *bilobata* was an exception.

Female pollinator offspring provide a measure of the reproductive success of each male fig. When figs of all pollinated varieties are considered together, there was a strong relationship between numbers female flowers and number of female pollinators in male figs (Figure 4.18-Figure 4.20). Linear regression, $R^2 = 0.21$, F = 40.68, df = 151, P < 0.001). There was a positive relationship between the number of female flowers in the male figs of var. *angustifolia* and the numbers of female pollinator offspring that had developed there (Table 4.30). No significant relationship was found in the other two varieties.



Figure 4.17 Relationships between the number of male flowers and the number of female flowers in male figs (\blacksquare) = var. *angustifolia*, (\bullet) = var. *deltoidea*, (\blacktriangle) = var. *trengganuensis*, (\bigstar) var. *bilobata*, (\bullet) var. *kunstleri* and (∇) var. *intermedia*. Each data point represents the individual fig.

Variety	n (trees)	n (figs)	Pearson's r	P value
angustifolia	20	100	0.95	<0.001
deltoidea	5	27	0.34	0.08
trengganuensis	5	25	0.05	0.79
bilobata	5	25	0.35	0.09
kunstleri	4	20	0.87	<0.001
intermedia	1	5	-0.23	0.71

Table 4.28 Within-variety relationships between the number of male flowers and the number of female flowers in male figs of *Ficus deltoidea* using Pearson correlation.

Variety	n	n	Ratio of male flower number	
	(trees)	(figs)		
			Mean ± SD	Range
angustifolia	20	100	0.19 ± 0.03	0.13-0.26
deltoidea	5	25	0.31 ± 0.05	0.24-0.46
trengganuensis	5	25	0.18 ± 0.04	0.12-0.25
bilobata	5	25	0.31 ± 0.09	0.21-0.58
kunstleri	4	20	0.21 ± 0.04	0.15-0.28
intermedia	1	5	0.65 ± 0.11	0.53-0.79

Table 4.29.	The ratio o	f anthers to	ovules in	n male figs	of six	varieties	of	F .
deltoidea.				U				



Figure 4.18 Relationships between the number of female flowers and the number of female pollinator offspring in male figs of var. *angustifolia*. Each data point represents one fig.



Figure 4.19 Relationships between the number of female flowers and the number of female pollinator offspring in male figs of var. *deltoidea*. Each data point represents one fig.



Figure 4.20 Relationships between the number of female flowers and the number of female pollinator offspring in male figs of var. *trengganuensis*. Each data point represents one fig.

Table	4.30	The	relations	hips	between	numbers	of	female	flowers	and	the
numbe	er of f	emal	e pollinat	or off	spring in	varieties o	of F	icus del	toidea.		

Variety	n (trees)	n (figs)	Pearson's r	P value
angustifolia	20	100	0.27	0.01
deltoidea	5	27	0.31	0.12
trengganuensis	5	25	0.13	0.54

The number of female flowers in male figs and the numbers galls and bladders produced were positively correlated across the three pollinated varieties (Figure 4.21, Linear regression, $R^2 = 0.43$, F = 111.79, df = 151, P < 0.001). In two of the varieties the total galled flowers (successful and failed galls (bladders) combined) there were more numerous in figs containing a greater number of female flowers (Table 4.31). This relationship was not present in var. *trengganuensis*

Varieties with more female flowers produced more seeds per fig (Figure 4.22, Linear regression, $R^2 = 0.76$, F = 466.91, df = 149, P < 0.001). This relationship was also present within the figs of two of the varieties, but not var. *angustifolia* (Table 4.32).

The relationship between the number of female pollinator offspring in a fig and the number of male flowers available to generate pollen for them to carry varied between varieties (Figure 4.23, Linear regression, $R^2 = 0.16$, F = 29.46, df = 151, P < 0.001). No significant relationship between number of male flowers in a fig and the number of female pollinators generated to transport the pollen was found in all varieties (Table 4.33. Ratios of the male flowers to the female pollinators was highest in var. *trengganuensis*, where a relatively small proportion of the female flowers in some of the figs had supported pollinator development (Table 4.34).



Figure 4.21. Relationships between the combined number of successful and failed galls (bladders) and the number of female flowers in male figs. (\blacksquare) = var. *angustifolia*, (\blacktriangle) = var. *trengganuensis*, (\bullet) = var. *deltoidea*. Each data point represents one fig.

Table 4.31 Correlations	between the	total number	of temale	nowers and	the
number of galls (succes	sful and empty	/ combined) in	male figs of	f Ficus delto	idea
using Pearson correlation	on.				

Variety	n (trees)	n (figs)	Pearson's r	P value
angustifolia	20	100	0.42	< 0.001
deltoidea	5	27	0.63	< 0.001
trengganuensis	5	25	0.30	0.14



Figure 4.22 Relationships between the mean number of seeds per fig and the number of female flowers in female figs. (\blacksquare) = var. *angustifolia*, (▲) = var. *trengganuensis*, (\bullet) = var. *deltoidea*). Each data point represents one fig.

Table 4.32 Correlations between the number of female flowers and the number	ber
of seeds produced in the female figs of <i>Ficus deltoidea</i> .	

Variety	n (trees)	n (figs)	Pearson's r	P value
angustifolia	20	100	-0.12	0.24
deltoidea	5	27	0.49	0.014
trengganuensis	5	25	0.61	0.001



Figure 4.23 Relationships between the numbers of female pollinator offspring with the number of male flowers in the male figs where they developed. $(\bullet) = var$. angustifolia, (•) = var. deltoidea, (\blacktriangle) = var. trengganuensis). Each data point represents a single fig.

the number of male flowers in male figs using Pearson correlation.					
Variety	n (trees)	n (figs)	Pearson's r	P value	
angustifolia	20	100	0.10	0.30	

Table 4.33 Correlations between the number of female pollinator offspring and the number of male flowers in male figs using Pearson correlation.					
Variety	n (trees)	n (fias)	Pearson's r	P value	

Variety	n (trees)	n (figs)	Pearson's r	P value
angustifolia	20	100	0.10	0.30
deltoidea	5	27	-0.18	0.39
trengganuensis	5	25	-0.10	0.67

Table 4.34 The ratios of male flowers to female pollinator offspring in three varieties of *Ficus deltoidea*.

Variety	n (trees)	n (figs)	Male flowers per female pollinators offspring	
			Mean ± SD	Range
angustifolia	20	100	0.86 ± 0.88	0.00-6.00
deltoidea	5	25	0.43 ± 0.12	0.12-0.70
trengganuensis	5	25	1.93 ± 1.52	0.25-5.00

4.5 Discussion

Seven of the 13 varieties in the *F. deltoidea* complex recognised by Corner (1969) are present in Peninsular Malaysia. They formed the basis of this comparative study, which aimed to assess the extent of the variation present in their figs. Natural populations of three varieties were also investigated, allowing their pollination biology and pollinators to be compared.

4.5.1 Variation between populations of *F. deltoidea* var. angustifolia

Var. *angustifolia* is the most widespread variety of *F. deltoidea*. It has the most reduction in leaf, fig and seed size and is considered the most derived variety within the *F. deltoidea* complex as described by (Corner, 1969). Comparisons of four populations of var. *angustifolia* in Peninsular Malaysia, which were separated by up to 552 km, found few differences in the structure and contents of their figs. The number of tepals in the female figs varied, and this may have reflected variation in the proportion of flowers that were fully developed. Seed lengths also varied, which may reflect soil nutrient variation (Suleman et al., 2013b) and some variation was also present in the number of unused flowers in female figs (but not the numbers of seeds produced). Differences in foundress numbers may have been responsible.

Populations differed in the total numbers of female flowers (and the tepals they contain) in their male figs. There may be a genetic basis to these differences as long distance gene flow may be rare in this species (Chapter 5). Soil and environment conditions may also have been responsible (Suleman et al., 2013b). The Dengkil plantation is on a silty soil, The Tembila plantation is on a sandy soil and the others on peat soil. Peat soils are very efficient as carbon and energy reservoirs and have very good water retention (Rezanezhad et al., 2016). The type of soil might influence the height of the host trees as Tembila showed the shortest oil palm trunk heights (Chapter 3) which might indicate that ability to retain water and nutrients is poor Weber et al. (2007). The effect of the soil might influence the individual fig characters as well as crop size.

The size of the pollinator females (as measured by gall size, ovipositor lengths and hind femur lengths) did not vary between populations. The number of galled flowers in male figs from different populations differed significantly. The extent of galling depends on the ability of foundresses to oviposit in the female flowers because galling is

apparently initiated at the same time as oviposition. There is often a positive correlation between the number of foundresses entering figs and the number of flowers that are galled even if individual clutch sizes per female (Raja et al., 2008a, Suleman et al., 2013b). Foundress numbers are difficult to estimate in this species because they regularly re-emerge from figs (Chapter 7) but it is likely that the number entering male figs varied between populations. This variation was not seen in seed production, perhaps because female figs contain so few flowers that entry by a single foundress results in pollination of almost all the flowers.

4.5.2 Variation between varieties of F. deltoidea

The varieties of *F. deltoidea* are distinguishable morphologically based on their size, shape and coloration of the leaves (Nur Fatihah et al. 2014; Mat et al. 2012; Zimisuhara et al., 2015). The figs of some varieties also differ considerably in diameter. Overlap in characters has led to the variation being recognised in the form of many discrete varieties rather than species (Corner 1969) or most varieties not being recognised as distinct (Berg and Corner 2005).

The most obvious differences between the seven varieties (in both plant sexes) were fig sizes and the number of flowers they contain. Varieties with small figs also have small leaves and plants with big figs have big leaves. Varieties with small figs include var. *angustifolia*, var. *deltoidea*, var. *bilobata*, var. *intermedia* and var. *motleyana* while those with big figs include var. *trengganuensis* and var. *kunstleri*. This observation is concordant with the evolutionary trends in *F. deltoidea* described by (Corner, 1969). Morphological features that were less variable (but differed between some pairs of varieties) were tepal lengths in both sexes, style lengths in both sexes, seed lengths in female figs and gall lengths in the male fig. Significant differences between varieties were found in all the traits examined in both male and female figs, but the sampling was limited and the extent of within variety variation is only known in var. *angustifolia*.

The variation between the seven varieties examined and the smaller within variety variation agrees with the leaf and fig characterisation of varieties of *F. deltoidea* by Awang et al. (2013) and Zimisuhara et al. (2015). A molecular approach using chloroplast DNA (cDNA) and nSSR markers from 24 individuals of four varieties of *F. deltoidea* showed there was a complicated diversity present in this complex and divided them into two large groups reflected in leaf size (Tnah et al., 2016). They also

concluded that var. *angustifolia* and var. *deltoidea* formed a distinct monophyletic group while var. *kunstleri* and var. *lutescens* were paraphyletic.

There was considerable variation in the numbers of flowers within the male figs but all the varieties had far more female flowers in male than female figs. This is in contrast to more usual pattern. Tarachai et al. (2008) for example recorded a mean of 858 female flowers in female figs of *Ficus hirta* Vahl. compared with a mean of 799 female flowers in male figs of this species. The unusual situation in *F. deltoidea* results from the unusually small numbers of flowers in the female figs (Corner, 1969). The extreme situation was found in female figs of var. *bilobata*, where they only produced a single female flower in each fig. Consistently high ratio of anthers to ovules confirmed that the varieties are likely to be passively pollinated.

4.5.3 Variation between fig wasps from three varieties of *Ficus deltoidea*

The pollinators of only three varieties were available for comparison. In contrast to their host plants, differences between pollinator females were small, although significant differences were present in the lengths of their ovipositors and hind femur. Significant differences were also present in the sizes of the galls they generated in different varieties. Gall size and hind femur length provide indications of the size of the insects. The ratio of ovipositor length did not differ between the pollinators of different varieties. The head shape also different between varieties (pers. Observation) where var. deltoidea tend to have m ore square head shape

In monoecious *Ficus* species mean style lengths are correlated with the ovipositor lengths of a tree's pollinator fig wasp, though not all individuals have ovipositors of sufficient length to oviposit in all the flowers (Nefdt and Compton, 1996). The male trees in dioecious species have figs with flowers that have far shorter styles than those in female figs. The styles grow in length faster in female figs and in both sexes reach maximum length during receptivity before shrinking afterwards (Ghana et al., 2017). As in other dioecious fig trees, the ovipositors of the pollinators of the three varieties were too short to reach the ovules of flowers in female figs but could reach all of the ovules in male figs.

The shortest mean ovipositor length was found in var. *angustifolia* which has small figs and the longest ovipositors were from var. *deltoidea*, which also has small figs. The ovipositor lengths of the pollinator of var. *trengganuensis*, which has big figs, were in between. The shortest ovipositor length (in pollinators from var. *angustifolia*) was still sufficient to penetrate the longest style lengths in male figs (in var. *trengganuensis*). This means the style length differences do not act as a barrier preventing pollinators ovipositing in figs of other varieties. In most *Ficus* species, non-pollinating fig wasps develop alongside pollinators in the figs and their host preferences can give an indication of differences between similar host plants. However, in the three *F. deltoidea* varieties no non-pollinating fig wasps were present.

4.5.4 Variation in reproductive output in different varieties of *F. deltoidea*

Factors that determine the number of fig wasp offspring in figs include the number of female flowers (Suleman et al., 2013a), the size of the foundresses (Herre et al., 1989), the number of foundresses entering the figs (Jousellin et al., 2001; Kjellberg et al., 2001) and non-pollinating fig wasp numbers (Tarachai et al., 2008). Higher temperatures lower the egg-laying capabilities of the pollinators that survive to enter figs (Wang et al., 2011) and temperatures during the development of fig wasp offspring influence offspring survival (Krishnan et al., 2014). Seed numbers are influenced by female flower numbers (Suleman et al., 2013a), the number of foundresses entering the figs (Jousselin et al., 2001) and the amount of pollen carried by the foundresses (Kjellberg et al., 2014). In the passively pollinated *Ficus carica*, only 10% of pollen is successfully transported away from the natal figs and the first foundresses to emerge load more pollen and are capable of pollinating more flowers (Kjellberg et al., 2014).

In this study, the varieties with figs containing more flowers produced more seeds and fig wasp offspring. The efficiency of seed production (the proportion of female flowers producing seeds) was consistently high as all three varieties recorded more than 60% of their flowers being pollinated. *Ficus hirta* Vahl. has female figs with many more flowers and seed production efficiency had a mean at nearly 50% (Yu et al., 2008). The highest efficiency was from var. *angustifolia*. Female figs of *F. deltoidea* contain so few flowers that it might be expected that a single foundress could easily pollinate all the flowers, even if they are passive pollinators like *Blastophaga* sp. This suggests either that a single foundress adequately pollinates var. *angustifolia*, but not var. *deltoidea* and var. *trengganuensis* that have more flowers, or that irrespective of

foundress numbers, pollination rates are more efficient in figs with fewer flowers. Figs with a higher number of flowers like var. *deltoidea* and var. *trengganuensis* will need more foundresses to enter the figs, which the plant can control to some extent by extending the time that the ostiole stays open after the first foundress has entered.

Male figs of var. *deltoidea* had the highest efficiency. The number of foundresses that entered each fig is likely to be the major reason for the higher efficiency in var. *deltoidea*. The high density of the *F. deltoidea* in Batu Pahat (Chapter 2 and Chapter 3) should help increase pollinator densities, but in a species like *F. deltoidea* where one foundress can pollinate several figs (Chapter 7), it may be that survival of foundresses walking between figs was also important. Jousellin et al. (2001) inferred that having more foundresses entering a single fig gives more advantage to the figs than the fig wasps as they bring more pollen, but also increase competition for oviposition sites. Having foundresses that can re-emerge from figs may reduce this difference.

4.6 Conclusions

(1) No significant morphological differences were detected between the figs from different populations of *F. deltoidea* var. *angustifolia* in Peninsular Malaysia.

(2) Morphological differences were present between the figs of different varieties, especially in the numbers of flowers that the figs contained, but the fig wasps associated with different varieties were very similar in size.

(3) No differences in the size in the fig wasps could be related to variation in their host figs and style lengths do not act as barrier to pollinator oviposition in different varieties.

Chapter 5

Host specificity of Ficus deltoidea pollinators

5.1 Introduction

5.1.1 Specificity of interactions between plants and insects

Interactions between plants and herbivores are often highly species-specific. Among tropical feeding guilds, granivores (seed eaters) showed the highest specificity while root feeders showed the least (Novotny et al., 2010). Among the interactions between plant and herbivores, pollination is a non-antagonistic relationship (mutualism) and co-evolution between these two groups has often led to speciation and an increase in biodiversity (Bruce, 2015). The rewards provided by the plants shape the behaviour of their pollinators (Nepi et al., 2018). Plants evolved their floral traits (colour, shape, odour, size) (Reverté et al., 2016) and floral rewards (pollen, sugar, oils) to attract potential pollinators (Gardener and Gillman, 2002). Some plants have few species of host specific pollinators, while others attract many species of pollinators that are not host specialists. The cues and rewards they provide shape the nature of this interaction (Ashman, 2009).

Nursery pollination differs from conventional pollination because the rewards for the pollinators are the sites for mating and developing their offspring (Sakai, 2002). The specificity in nursery pollination is often high, with one plant species pollinated by only or a few insect pollinators (Rodriguez et al., 2017). Among nursery pollination systems the fig tree (*Ficus*, Moraceae) and fig wasp (Agaonidae) mutualism displays this high specificity. Coevolved mutualisms often exhibit high levels of partner specificity. Obligate pollination mutualisms, such as the fig –fig wasp and yucca–yucca moth systems, represent remarkable examples of such highly species-specific associations. Mutualism favours higher host specificity than does antagonism in plant –herbivore interaction.

5.1.2 *Ficus* pollination

In the fig tree and fig wasp nursery pollination mutualism, the fig wasp progeny feed on the endosperm tissue of the galled ovary (Cook and Segar, 2010) and they are included under group one as classified by Sakai (2002) (see Chapter 1) because the offspring are seed/ovule parasites where the pollinator offspring feed on galled ovules which may or may not have been pollinated. The pollination of figs begins away from the trees when female fig wasps detect volatiles emitted by the receptive figs of their host tree (Wang et al., 2013). The scent acts as signal to the wasp that the fig is receptive and ready to be pollinated (Hossaert-McKey, 2016). This specific scent is usually developmental stage and species specific (Chen et al., 2009).

Once a fig wasp has landed on a receptive fig, it often examines the surface of the fig before deciding to go in and may be responding to physical and surface chemical cues. Their morphology needs to be well adapted to their host figs in order to gain access to the florets via the narrow ostiole (van Noort, 2003), which is the link between the outside and the flowers on the inside of the figs (Liu et al., 2013). The specific pollinator will have a head shape that is correlated to fig size (van Noort and Compton, 1996). Female fig wasps possess several adaptations allowing them to pass through the ostiole such as flattened heads, modified mandibles (Compton, 1993), and spikes on certain body parts (Harrison, 2005). This ostiole acts as a physical barrier preventing females of non-adapted pollinator species from gaining entry into the fig cavity, from where the ovules are accessible (van Noort, 2003; Liu et al., 2003).

During passage through the ostiole, pollinator females lose their wings and most of their antennae (van Noort, 2003). They do not need the wings and all their antennae once they are in the fig cavity, and their loss ensures that the insects cannot fly off again to other trees. Soon after the entrance of the pollinators, the ostiole will close up and seal off the fig cavity (Nefdt and Compton, 1996). Once in the fig cavity, the pollinators need to deposit their eggs into the female flowers and also to gall them. Gall formation involves expansion of the ovary. Gall production by insects is typically highly species specific, and the ability to gall may be an important factor in the host specificity of fig wasps (Ghana et al., 2015a).

An early record of pollinator sharing was in 1989 where the pollinator of *Ficus burtt-davyi* Hutch (*Elisabethiella baijnathi* Wiebes) was introduced to Kirstenbosch Botanical Garden in Cape Town. Not only did it pollinate the *F. burtt-davyi*, it also

pollinates and reproduces in two other *Ficus* species namely *Ficus* craterostoma Warb. ex Mildbr. & Burret and *Ficus* stuhlmannii Warb. that were planted in the garden (van Noort, 2003). The seeds were not viable beyond the cotyledon stage (van Noort, 2003). However, the wasps reproduced in all three figs but they can only exit the figs of *F. burtt-davyi*. The wasps were able to reproduce and offspring emerged into the cavity of *F. craterostoma* and *F. stuhlmannii* figs but failed to exit. This suggests another way in which host specificity is maintained

In monoecious fig trees, the styles of the female flowers vary in length. The shorter ones will mainly support the development of fig wasp progeny while the longer styles will support the production of seeds (Herre et al., 2008). In dioecious fig trees, the female fig wasps that enter female figs pollinate the female flowers and these produce seed. The pollinators gain nothing by entering the female figs as no offspring will be produced (Hossaert-McKey, 2016). Pollinators that enter the figs of male trees can deposit their eggs and transform the ovule into a gall where the larvae will feed (Ghana et al., 2015a). For this purpose, they must have an adequate ovipositor length that suits the style length of the receptive female flowers (Nefdt and Compton, 1996; Weiblen 2002; van Noort, 2003). In dioecious fig trees, all female flowers in the male figs are short-styled making them suit the pollinator's ovipositor length (Nefdt and Compton, 1996).

5.1.3 The breakdown of host specificity

It used to be thought that a single *Ficus* species was routinely pollinated by only one specific fig wasp. This was known as the 'one-to-one' rule (Rasplus, 1996; Weiblen, 2002). Some exceptions were known, but the development of a modern molecular approach has shown that one fig tree species often has more than one pollinator (Cook and Segar, 2010) and in rare instances a single fig-wasp can pollinate several fig trees (pollinator sharing) (Kerdelhue et al., 1997; Su et al., 2008) making the relationship less specific than early research suggested.

Co-pollination is a term used when one plant species is associated with more than one pollinator. Based on molecular studies, the occurrence in *Ficus* is more than 30% and this figure will inevitably increase (Yang et al., 2015). One classic finding of co-pollination was from (Michaloud et al., 1985), which discovered two pollinating wasps (*Courtella camerunensis* Wiebes and *Courtella gabonensis* Wiebes) were able to pollinate *Ficus ottoniifolia* Miquel in Gabon (Africa). However, the fig trees occupy

different geographical locations. In the rain forest, *C. camerunensis* tend to be more abundant than *C. gabonensis* while in open habitat, the situation was reversed with the almost total exclusion of *C. camerunensis*. On the other hand, in the savannah of lvory Coast, *C. gabonensis* appear to be the sole pollinator for *F. ottoniifolia* with exclusion of *C. camerunensis*. This result contributed to the suggestion by Kerdelhue et al., (1997) that different geographic distributions of pollinators can lead to two wasp species pollinating the same *Ficus* species.

Ficus sur Forsskal, another African species, also has more than one fig wasp that pollinates their flowers (Ceratosolen flabellatus Grandi and Ceratosolen silvertrianus Grandi). They can be either sole pollinators or can co-occur in the same fig (Kerdelhué et al., 1997). Other research by Kerdelhue et. al. (1997) and Proffit and Johnson (2009) discovered that another pollinator for *F. sur* was *Ceratosolen capensis* Grandi. The pollinators can exist in the same crops but rarely in the same figs. Inter-specific competition was not recorded in this study as the production of seeds and wasp offspring were similar between a fig that was pollinated by a single wasp species and the presence of both fig wasp species in the same fig (Kerdelhue et al., 1997). The figs are big and can accommodate lots of pollinator foundresses. Whether or not competition occurs will depend on the total number of foundresses that enter. At low densities there will be no competition. Cornille (2012) has proposed that one barrier that can mediate inter-specific competition between two pollinators is if the ostiole will close rapidly once entered, and the arrival times for the species varies. Liu et al., (2013) noted that attractant volatile compounds decrease significantly after foundress entry making it hard for later pollinators to detect.

In the African tree *Ficus sycomorus* L. two different fig wasps colonise the fig trees namely *Ceratosolen arabicus* Mayr and *Ceratosolen galili* Wiebes (Compton et al., 1991). However, only *C. arabicus* pollinates the flowers while *C. galili* does not pollinate. *C. galili* was found to also enter figs of *Ficus mucoso* Ficalho (van Noort, 2003). *C. galili* was found to be smaller than C. *arabicus*. This might be the reason why *C. galili* is able to develop in un-pollinated flowers of *F. sycomorus*. A study by Liu et al., (2013) has found that being smaller has advantages for fig wasps entering figs and they can penetrate the ostiole more easily.

A breakdown of host specificity occurs when a fig wasp responds to the volatile chemicals emitted by non-usual host receptive figs, successfully gains access to the fig cavity via the ostiole and lastly, is able to reproduce inside (van Noort, 2003). Some fig trees produce less specific stimuli that can attract more than one pollinator. Geographical area and season might also cause scent variation even in the same species (Rodriguez et al., 2017). This variation in the floral volatiles may allow a few species of pollinators to co-exist in a single fig tree (Wang et al., 2016). A study by Ware and Compton (1992) found that receptive *Ficus lutea* Vahl were entered by females of Elisabethiella stuckenbergi Grandi, Elisabethiella baijnathi Wiebes and Ceratosolen capensis Grandi. All those pollinators were not the normal pollinator of F. lutea. This case was considered as showing mistakes that happens due to the absence of the normal pollinator and the figs remaining receptive for longer (Liu et al., 2013). However, the ostiole of *F. lutea* may have acted as barrier and prevented *E. baijnathi* from entering the figs, leaving only two pollinators able to enter through the ostiole. However, only E. stuckenbergi was able to reproduce (Ware and Compton, 1992). In a study by Wang et al., (2016), the pollinator of *Ficus hainanensis* Merr. and Chun was attracted to the volatiles emitted by the receptive figs of Ficus auriculata Lour. However, the normal pollinators of F. auriculata were not attracted to the receptive figs of F. hainanensis, which makes the relationship asymmetric.

Ficus microcarpa Blume is associated with several pollinator fig wasps within its native area (Wang et al., 2015) and a single *Ficus* tree can have up to four different agaonid species (Compton et al., 2009). In Uganda, *Alfonsiella brongersmai* Wiebes, *Alfonsiella natalensis* Wiebes, *Elisabethiella allotriozoonoides* (Grandi) and *Alfonsiella longiscarpa* Joseph were found in figs of *Ficus natalensis* Hochst. and three of the pollinators generated viable seed. Studies by other researchers also discovered different pollinators of *F. natalensis* apart from those four agaonid wasps. A study by Cornille et al., (2012) discovered a single pollinator wasp *Elisabethiella stuckenbergi* Grandi was able to pollinate two different host figs. The first host fig was *F. natalensis* that also has another normal pollinator which is *Elisabethiella socotrensis* (van Noort, 2003) and they can co-occur in the same fig. Another fig species pollinated by *E. stuckenbergi* was *Ficus burkei* (Miq.) Miq. that also has a different routine pollinator (*Alfonsiella brongersmai* Wiebes) in East Africa (Cornille et al., 2012). Similarities in odours released by *F. burkei* and *F. natalensis* might be the reason why they can share pollinators (Cornille et al., 2012).

Some fig wasp species acquire new host species and build new relationships. Wachi et al. (2016) showed that two fig tree species (*Ficus formosana* Maxim. and *Ficus tannoensis* Hayata) have lost their native pollinators and have acquired new pollinators from *Ficus erecta* Thunb. A single *Blastophaga* now pollinates three different fig tree species. Liu et al. (2015) also found that absence of the routine pollinator and host shifting from *Ficus heterosyla* Merr. resulted in new pollinators of *Ficus squamosa* Roxburgh. The male trees of *F. heterostyla* have female pollinators that emerge at the same time as receptive figs of *F. squamosa*. Similarities in flowering phenology may have led to this pollinator sharing in *F. heterostyla* and *F. squamosa* in China (Liu et al., 2015).

Most of the examples of pollinator sharing come from monoecious fig trees (Wachi et al., 2016) but recently a number of breakdowns have been found in dioecious species such as *F. squamosa* (Wang et al., 2016). Pollinator host changes can lead to speciation as the fig wasps adapt to their novel host figs (Wachi et al., 2016). Hossaert-McKey et al. (2010) proposed that co-pollinators were more likely where (1) fig trees live in the same geographical area (2) there is similarity in flowering phenology between species (3) there are similarities in fig morphology (size) and (4) there are similarities in the volatiles emitted by their receptive figs. Liu et al., (2015) added an absence of native pollinators to this list.

5.1.4 Specificity of pollination in *Ficus deltoidea* varieties

Ficus deltoidea Jack. (Moraceae) is distributed across Southeast Asia. Berg and Corner (2005) place *F. deltoidea* in Subgenus *Ficus*, Section *Ficus*, and Subsection *Frutescentiae*. It is mainly distributed in Peninsular Malaysia, Sumatra, Java and Borneo, where it is most diverse and may have originated (Corner, 1969). *F. deltoidea* is dioecious, where the male and female functions are separated. Some varieties of *F. deltoidea* are unusual in that they are true epiphytes whereas most *Ficus* species are stranglers (hemiepiphytes) or free-standing trees (Starr et al., 2003). As well as being epiphytic, this complex contains varieties that can be epilithic (growing on the surface of rocks) or terrestrial shrubs and treelets (Berg and Corner, 2005). Individuals of the same variety can have different growth forms. They are well adapted as epiphytes by having slow growth and large seeds to overcome the nutrient scarcity where they grow (Berg and Corner, 2005).

The numerous varieties under *F. deltoidea* are discriminated through their leaf morphology and anatomy (Nur Fatihah et al., 2014). The most prominent leaf characteristics that differentiate varieties in *F. deltoidea* are their pinnate venation, obtuse deltoid leaf and forked vein (Corner, 1969). Within the same variety, they can have different leaf characters in young and mature leaves (Awang et al., 2013). The varieties also exhibit variation in their figs and flowers (Berg and Corner, 2005). A morphological study by Nur Fatihah et al. (2014) recognised two sub-species and seven varieties of *F. deltoidea* from Peninsular Malaysia. These are var. *deltoidea* Corner, var. *angustifolia* (Miq.) Corner, var. *trengganuensis* Corner, var. *bilobata* Corner, var. *intermedia* Corner, var. *kunstleri* (King) Corner, and var. *motleyana* (Miq.) Corner. Among these varieties, variety *angustifolia* is the most widespread and has been considered as the most derived while variety *motleyana* may be close to the ancestors of all the varieties (Corner. 1969).

Little information is recorded on the pollinator or pollinators of *F. deltoidea*. A general pattern regarding the Subgenus and section Ficus has been established where dioecious Subgenus Ficus section Ficus is usually pollinated by a fig wasp from the genus Blastophaga (Wiebes, 1979). Blastophaga guadrupes Mayr is the only recorded pollinator of Ficus deltoidea. It probably pollinates var. lutescens Desf. as it was collected in Java and Sumatra (Wiebes, 1993). In a study by Laman and Weiblen (1998), two different varieties of *Ficus deltoidea* (var. *borneensis* and var. *motleyana*) present in Gunung Palung, Indonesia were also recorded as pollinated by Blastophaga quadrupes. However, in apparent contradiction to this and Corner's (1969) conclusions, there is indirect evidence that some at least of the varieties of F. deltoidea may be biologically distinct species, because different species of pollinator fig wasps have been reared from them in Brunei (F. Kjellberg, personal communication to S.G. Compton 2016). However, this is not conclusive, because a single Ficus species can support more than one pollinator species (Cook and Segar, 2010), often in different areas of the plant's range or when plants are growing in different habitats (Souto-Vilaros et al., 2018).

Floral and inflorescence differences (Chapter 4) also suggest that different pollinators may be associated with different varieties of *F. deltoidea* and that this taxon may be a complex of closely related but biologically distinct species. The taxonomic and biological status of these varieties is unclear, as is the extent to which different varieties support different fig wasp pollinators. There are numerous studies on the pharmaceutical properties of *F. deltoidea* varieties, but no study has been conducted

on the composition of the attractant volatiles they release or whether pollinators from one variety are only attracted by the volatiles of that variety.

5.2 Objectives

In this study, the names of *F. deltoidea* varieties are based on Corner (1969). The following questions concern the specificity of its fig wasps and whether they are associated with particular varieties. Specifically, (1) what is the host specificity of the pollinator of *Ficus deltoidea* var. *angustifolia* and (2) will it also pollinate other varieties? We also examine questions related to the pollination of this variety of *F. deltoidea* that provide context for the specificity of its pollination. (3) Do the fig wasps prefer male figs over female figs? (3) How willing are pollinators to move from their natal male trees?

5.3 Methods

5.3.1 Study species

Plants in containers were used in the experiments together with plants belonging to variety *angustifolia* that were growing in situ as epiphytes on oil palms. Pot-maintained plants belonging to seven varieties of *Ficus deltoidea* were obtained on loan from the germplasm collection of Universiti Sultan Zainal Abidin (UniSZA), Tembila campus in Terengganu, an eastern coast state of Peninsular Malaysia. The varieties were var. *angustifolia*, var. *deltoidea*, var. *trengganuensis*, var. *kunstleri*, var. *bilobata*, var. *motleyana* and var. *intermedia*. At the time of collection, none of the figs on the plants in the germplasm collection had been pollinated. They varied from one to 10 years old. The trees originated from cuttings from several places in Malaysia namely Johor, Kelantan, Melaka, Terengganu, Sabah and Sarawak. They were maintained in either 39 cm or 49 cm diameter clay pots. The media for all the fig trees was a combination of top soil, pome, rice husk and peat soil. All of the collection were given NPK Green organic fertilizer and were watered regularly.

In addition, four pot-maintained female trees of *F. deltoidea* var. *angustifolia* were obtained on loan from the glasshouse of Universiti Kebangsaan Malaysia (UKM) and three male plants of this variety were bought from Batu Pahat, Johor. The male trees

from UKM were bought from MARDI Serdang in 2013 while the Batu Pahat trees were propagated through cuttings from KESEDAR, Kelantan in 2006.

The collections from UKM and Batu Pahat were maintained in 1 litre poly planter bags with peat soil before being transferred to clay pots with the same diameters as in Terengganu germplasm collection (TGC) collection. Once they were transferred to the TGC, they were given the same fertilizer and the same media. All the trees were healthy and producing new figs before the experiments were conducted.

5.3.2 Study sites

Experiments were carried out at three Peninsular Malaysian study sites. The first was an oil palm plantation in Kampung Olak Lempit which is in Banting district in the state of Selangor. It is located approximately 50 kilometres from Kuala Lumpur city centre. The climate of the area is described in Chapter 2. The plantation is about 2.1 hectares in area and was planted in 2001. This plantation is managed by a small-scale farmer who lives on the site. There are about 113 epiphytic individuals of *F. deltoidea* var. *angustifolia* on the 285 oil palm trees in the plantation (Chapter 3). These include some big male trees of var. *angustifolia* that were convenient for experimental purposes. The phenology of the fig trees is described in Chapter 6.

The second study site was at the Terengganu germplasm collection (TGC) of Universiti Sultan Zainal Abidin (UniSZA). This collection was established in 2006 and contains 165 medium-large individuals of *Ficus deltoidea* varieties and many smaller plants. It contains eight varieties: var. *angustifolia*, var. *deltoidea*, var. *trengganuensis*, var. *kunstleri*, var. *bilobata*, var. *motleyana*, var. *intermedia* and var. *borneensis* Corner. Of all these varieties, only var. *borneensis* is not native in Peninsular Malaysia. They were propagated from cuttings taken from all over Peninsular Malaysia while the var. *borneensis* were received from the Sarawak Biodiversity Centre. Because no permission to use the var. *borneensis* was granted and this variety is not native in Peninsular Malaysia, the experiment only used the other seven varieties. All the pots were maintained in clay pots.

The third study site was at Parit Maimon, situated in Batu Pahat district in the state of Johor. This site is about 239 km (150 miles) south of Kuala Lumpur city centre. Within this site, two epiphytic varieties of *F. deltoidea* (var. *angustifolia* and var. *deltoidea*) have colonised oil palm trunks. They are also planted in the ground within the site's

residential area together with two other varieties, namely var. *kunstleri* and var. *trengganuensis*.

5.3.3 Host choice by Blastophaga sp. (var. angustifolia)

In this study, the pollinators of *F. deltoidea* var. *angustifolia* were used in several manipulation experiments. *Ficus* species in subgenus *Ficus* and section *Ficus* are pollinated by fig wasps belonging to the genus *Blastophaga*. The identity of the pollinator of var. *angustifolia* is unknown so the pollinator of this variety is called *Blastophaga* sp. (var. *angustifolia*) here. The pollinator of var. *deltoidea* is referred to in a similar way.

5.3.3.1 Transplant experiment one

Twenty-five *Ficus deltoidea* individuals from seven varieties were obtained on loan from TGC and moved by lorry to the oil palm plantation in Kampung Olak Lempit, Banting. The distance between TGC and Banting was about 566 km. The plants were of a convenient size, in pots of 39 cm and 49 cm diameter, and all were producing figs. The varieties were var. *angustifolia*, var. *deltoidea*, var. *trengganuensis*, var. *kunstleri*, var. *bilobata*, var. *motleyana* and var. *intermedia* (Table 5.1). Two males and two females of var. *deltoidea*, var. *trengganuensis* and var. *kunstleri*, and two males and one female of var. *bilobata* were available. Only two and one trees respectively of var. *motleyana* and var. *intermedia* (East of var. *motleyana* and var.

Two big male epiphytic *Ficus deltoidea* var. *angustifolia* individuals (D7 and D22) with abundant figs were chosen as potential donors of pollinators to other varieties (Table 5.1). The two focal trees were typical of plants at the Banting site (Chapter 6), with non-synchronous production of figs and all the developmental phases present at any one time (Chapter 6). Each tree had more than 5000 figs present at the time of the experiment. The basal height from the ground for D7 was 2.2 m, while for D22 it was 2.6m. The distance between all the transplants in pots and the donor trees ranged between 1 m to 4 m while the figs on different varieties were as little as 20 cm apart. One female and one male plant from each variety were placed in a circle around the bases of the oil palms supporting the two epiphytic male var. *angustifolia* (Figure 5.1) to see whether the pollinators will enter all the varieties or only specific varieties. Individuals of var. *angustifolia* acted as controls. Plants belonging to the same variety were placed apart, to avoid any directional bias.

	• •		•	
No.	Variety	Tree code	Sex	Donor tree
1	angustifolia	FDA 1	Male	D7
2	angustifolia	FDA 2	Male	D22
3	angustifolia	FDA 6	Female	D7
4	angustifolia	FDA 7	Female	D22
5	deltoidea	FDD 1	Male	D7
6	deltoidea	FDD 2	Male	D22
7	deltoidea	FDD 3	Female	D7
8	deltoidea	FDD 4	Female	D22
9	trengganuensis	FDT 1	Male	D7
10	trengganuensis	FDT 2	Male	D22
11	trengganuensis	FDT 3	Female	D7
12	trengganuensis	FDT 4	Female	D22
13	kunstleri	FDK 1	Male	D7
14	kunstleri	FDK 2	Male	D22
15	kunstleri	FDK 3	Female	D7
16	kunstleri	FDK 4	Female	D22
17	bilobata	FDB 1	Male	D7
18	bilobata	FDB 2	Male	D22
19	bilobata	FDB 3	Female	D7
20	motleyana	FDM 1	Female	D7
21	motleyana	FDM 2	Female	D22
22	intermedia	FDI 1	Male	D7

Table 5.1 *Ficus deltoidea* individuals obtained from TGC and positioned in pots around potential pollinator donor male individuals of var. *angustifolia* growing as epiphytes in an oil palm plantation in Banting.



Figure 5.1 Varieties of *Ficus deltoidea* encircling large epiphytic male *Ficus deltoidea* var. *angustifolia*. Left donor tree D7, right donor tree D22.

The potted plants were moved into position around donor trees using a wheelbarrow. A total of 12 pots were placed around tree D7 and 10 pots around tree D22. For donor tree D7, the pots were placed in an arc of about 180 degree around the oil palm facing the area of the donor where most figs were. The pot plants encircled D22, forming an arc of about 220 degree. Once in position, all the figs on the potted plants were removed to stimulate production of new figs. Removal took two people about three days. The starting date of the experiment was taken to be once all the figs were pulled off (15th January 2017). They remained in place until 15th May 2017, when the experiment was terminated.

The trees in pots were watered regularly, and all survived throughout the experiment. They also all produced new figs except for plant FDK 1 (var. *kunstleri*). The plants were checked twice a month for the presence of any pollinated figs. At the end of the experiment, the numbers of figs at each developmental phase were also counted. Any DE phase figs were scored fortnightly throughout the experiment to make sure that none were missing at the end because the figs had been eaten by birds. To determine whether any figs were being entered by pollinators, but failing to develop, 50 aborting figs of each individual plant were brought to the Lab. and examined using a binocular microscope to check for the existence of wings in fig ostioles and wasp bodies inside the figs.

5.3.3.2 Transplant experiment two

Three additional male *F. deltoidea* var. *angustifolia* (with no pollinated figs) were brought to the Banting oil palm plantation from TGC to act as pollinator donor trees in experiment two (plants FDA 3, FDA 4, and FDA 5). They were placed next to three big male epiphytic *F. deltoidea* var. *angustifolia* individuals growing as epiphytes on oil palms (not D7 and D22) to allow their figs to be entered by pollinators. Together with potted plants FDA1 and FDA2 this produced five male var. *angustifolia* with figs containing developing pollinator offspring.

After four months, the potted plants (all varieties) were taken by lorry back to TGC, where experiment two was conducted from 16th May 2017 until 16th August 2017. A month earlier, all the figs on the plants in the TGC (of all varieties) had been removed to stimulate production of new figs. This removal of figs was carried out by four assistants from UniSZA. For this experiment, the five male var. *angustifolia* were used as a source of pollinators. The five male plants of var. *angustifolia* with fig wasps were

put together with the other varieties adjacent to the rest of the germplasm collection (Figure 5.2). The pots were placed back to their original places in the collection because most of the other pots were too large to be moved conveniently.

Data similar to that collected in Experiment one were recorded at the end of the experiment, namely how many figs were available to be entered and how many figs developed in response to entry. Fifty figs from each fig tree of all the varieties were again collected and examined under a binocular microscope for the presence of pollinator wings and fig wasp bodies.

5.3.3.3 Transplant experiment three

Experiment three was a continuation of experiment two. This experiment was carried out after the entire TGC collection was moved to an area in UniSZA about 500 m away. This allowed the experiment to be conducted more systematically because the experimental design was given to the UniSZA staff ahead of the move and the staff unloaded and arranged all the pots accordingly. Seven pollinated male trees of *Ficus deltoidea* var. *angustifolia* were used as donor trees. Five of the trees were the donor trees used in experiment two while two male trees had acquired the pollinators in experiment two. Male and female individuals of var. *angustifolia* with no pollinated figs acted as controls. Six other varieties used in this experiment were var. *deltoidea*, var. *trengganuensis*, var. *kunstleri*, var. *bilobata*, var. *motleyana* and var. *intermedia*. The arrangement of the clusters of plants is shown in Figure 5.2.

Arrangements of different combinations depended on the availability of recipient plants (A-G in Figure 5.3). The experiment was conducted simultaneously around the seven donor plants. The donor trees were placed in the middle of different combinations of varieties and sexes and controls were present except in combinations F and G. The distances apart of donor plants were not less than 10 m to reduce the chance of pollinators moving between tree cluster combinations. All the pots were placed as close as possible to the donor trees (all were less than 1 m away and the fig plants were touching if possible). This was to increase the chances of fig wasps moving from donor plants.



Figure 5.2. The arrangement of pots containing varieties of *Ficus deltoidea* in the Terengganu Germplasm Collection (TGC). The var. *angustifolia* that were releasing pollinators were FDA 1, FDA 2, FDA 3, FDA 4 and FDA 5.


Figure 5.3. The arrangement of pots containing *Ficus deltoidea* varieties during experiment three at UnisZA.

5.3.3.4 Transplant experiment four

This manipulation experiment was similar to experiment three, with all the pots positioned as before. However, in this experiment, all the receptive figs on the donor tree were removed to prevent the wasps that emerged from entering figs on their natal plants. As before, clusters F and G did not offer any receptive figs of var. *angustifolia* to emerging pollinators, while the other versions again offered emerging pollinators a choice between male and female figs of var. *angustifolia*.

5.3.4 Host choice by Blastophaga sp. (var. deltoidea)

This experiment was conducted in the Banting oil palm plantation using epiphytic var. *angustifolia* plants growing on oil palms as pollinator recipients and pollinators emerging from figs of var. *deltoidea* as donors. The experiment started with bagging pre-receptive figs (A phase) within fine mesh bags to prevent the entry of any pollinators. A total of 27 bags were placed around stems on 5 different trees bearing multiple (9-18) figs. Two of the five trees were female, three were male. The A phase figs from var. *angustifolia* usually take less than a week to become receptive (B phase). During this period mature male figs of var. *deltoidea* were collected from Batu Pahat in Johor.

At Batu Pahat (Johor), mature D/E phase figs of var. *deltoidea* were removed from the trees and brought to a Lab. In Kuala Lumpur where they were kept in containers covered by fine mesh. Once the fig wasps had emerged from the figs they were quickly brought to the plantation where a single foundress was released into each bag where receptive figs were available. The bags were then closed to prevent other pollinators from entering. 30 days later, the bags were opened to check for the presence of wings and the bodies of foundresses and record any developmental signs that pollinators had entered

5.3.5 Statistical methods

Chi-square tests were conducted to determine whether there were significant differences in frequencies of entry of male and female figs of var. *angustifolia*. Chi-square tests were also conducted to examine differences between entry rates of the same gender from different varieties. All analyses used SPSS Statistics 20.

5.4 Results

5.4.1 Host choice by Blastophaga sp. (var. angustifolia)

5.4.1.1 Transplant experiment 1

After four months, the total numbers of figs on the transplanted potted plants were counted. The figs present on different plants at the end of the experiment ranged from 0 to 4672 (Table 5.2). The only fig tree that did not produce figs was plant FDK 1 (var. *kunstleri*). As mentioned in Chapter 2, different varieties of *Ficus deltoidea* produce either big or small figs. The smallest figs are on var. *bilobata*, followed by var. *angustifolia*, var. *deltoidea*, var. *intermedia* and var. *motleyana*. Conversely, var. *kunstleri* and var. *trengganuensis* produce large figs. The varieties with smaller figs tended to produce more figs.

In this experiment, the largest number of figs found was in var. *bilobata*, with more than 4000 figs. Some plants of var. *angustifolia* and var. *deltoidea* also had more than 2000 figs each. Numbers of replicate individuals were small, but male trees tended to have more figs than females (var. *bilobata*, var. *deltoidea*, var. *angustifolia* and var. *kunstleri*) but var. *trengganuensis* females had more figs. In var. *motleyena* and var. *intermedia*, since only one gender was there, the comparison cannot be made.

The developmental phases of the figs on the male transplants at the end of the experiment showed that only var. *angustifolia* had C phase figs (entered and developing) (Figure 5.4). The experiment was terminated before any of these figs had released pollinator offspring (DE phase). The only phases on plants of the other varieties were either AB (immature or ready to be pollinated) or aborting. Male trees of var. *angustifolia* recorded the highest percentage of AB phase figs (73.6%) and C phase (25.6%). C phase was entirely absent in the other varieties. Aborting phase figs were in a majority on all the other varieties, but were just 0.9% of the total on the var. *angustifolia* plants.

Recipient variety	Don or Tree	Tree No.	Sex	Total number of figs present on the plants during		Figs	
				Beginni	End	Entered	Un-entered
				ng			
angustifolia	D7	FDA 1	Male	0	2175	445	1730
	D22	FDA 2	Male	0	201	162	39
	D7	FDA 6	Female	0	1652	63	1589
	D22	FDA 7	Female	0	1767	88	1679
deltoidea	D7	FDD 1	Male	0	2430	0	2430
	D22	FDD 2	Male	0	493	0	493
	D7	FDD 3	Female	0	54	0	54
	D22	FDD 4	Female	0	248	0	248
trengganuensis	D7	FDT 1	Male	0	215	0	215
	D22	FDT 2	Male	0	76	0	76
	D7	FDT 3	Female	0	836	0	836
	D22	FDT 4	Female	0	202	0	202
kunstleri	D7	FDK 1	Male	0	0	0	0
	D22	FDK 2	Male	0	145	0	145
	D7	FDK 3	Female	0	10	0	10
	D22	FDK 4	Female	0	125	0	125
bilobata	D7	FDB 1	Male	0	4591	0	4591
	D22	FDB 2	Male	0	4672	0	4672
	D7	FDB 3	Female	0	35	0	35
motleyana	D7	FDM 1	Female	0	82	0	82
	D22	FDM 2	Female	0	124	0	124
intermedia	D7	FDI 1	Male	0	135	0	135

Table 5.2. The entry of pollinators into figs of *Ficus deltoidea* varieties positioned around epiphytic male var. *angustifolia* in Banting plantation.

Results for female trees of *F. deltoidea* were similar to those of the males, insofar as only figs of var. *angustifolia* provided evidence of pollination, but the females of this variety had a much higher proportion of AB figs than the males (91.1%) and far fewer pollinated figs (1.8% C and 2.6% DE phases). Only 4.5% of the figs of female var. *angustifolia* were aborting, which suggests that many of the female figs were still available to be pollinated. Frequencies of female AB phase figs in the other varieties ranged from 18.5% to 29.8%, and all the remaining figs on these plants were aborting (Figure 5.5).

Among the transplanted individuals, only var. *angustifolia* had pollinated figs (Figure 5.6). All individuals of this variety had some figs that were pollinated. In FDA 1 (male), 445 figs from 2175 figs were entered (20.5 %) and in FDA 2 (also male), 162 figs from 201 figs were entered (80.6 %). Among female trees 63 figs from 1652 figs were entered on plant FDA 6 (3.8 %) and 88 figs from 1767 figs on FDA 7 (5.0 %). The pollinated figs on female trees included both C and DE phases. The entry rates into male transplants (FDA 1 and FDA 2) around different donor trees (D7 and D22) were significantly different ($\chi^2 = 5.07$, df = 1, P < 0.01). A similar difference was found among the female transplants FDA6 and FDA7 around different donor trees (Chi square $\chi^2 = 9.904$, df = 1, P < 0.05). So for both sexes, higher entry rates were achieved around donor tree D22 than donor tree D7.

Entry rates for the two male var. *angustifolia* were higher than for the two conspecific females. For FDA 1 (male, 20 un-entered, 445 entered) and FDA 6 (female, 92 un-entered, 63 entered) around donor tree D7 (Chi square, $\chi^2 = 238.05$, df = 1, P < 0.01. Around donor tree D22 rates for male FDA 2 were 1 un-entered and 162 entered compared with 62 un-entered and 88 entered for female plant FDA7 (Chi square $\chi^2 = 80.57$, df = 1, P <0.001). The sexual difference in entry rates around donor tree D22 appeared to be independent of crop sizes, which were almost the same on the male and female plants.

Between zero and 58 aborting phase figs from all varieties except var. *angustifolia* were dissected, with sample sizes varying according to availability (Table 5.3). This was to check whether pollinators had entered the figs, but they had subsequently failed to develop. None of the figs had wings present in their ostioles or contained the bodies of pollinator wasps.



Figure 5.4. Phases of fig development in male *Ficus deltoidea* figs positioned around male var. *angustifolia* in an oil palm plantation. Hatched bars = AB phase, Black bar = C phase, grey bars = Aborting. Combined data – sample sizes vary between varieties.



Figure 5.5. Phases of fig development in female *Ficus deltoidea* figs positioned around male var. *angustifolia* in an oil palm plantation. Hatched bars = AB phase, Black bar = C phase, open bar = DE phase, grey bars = aborting. Combined data – sample sizes vary between varieties.

163



Figure 5.6 Pollination rates in four transplanted *Ficus deltoidea* var. *angustifolia*. Solid bars = male trees, open bars = female trees.

Variety	Tree No.	Donor tree	Sex	No of figs collected	Figs with wings/wasps
	FDD1	D7	Male	54	0
doltoidoa	FDD2	D7	Male	54	0
Genoldea	FDD3	D22	Female	51	0
	FDD4	D22	Female	57	0
	FDT1	D7	Male	55	0
trenaganuensis	FDT2	D22	Male	56	0
trenggandensis	FDT3	D7	Female	55	0
	FDT4	D22	Female	51	0
	FDK1	D7	Male	0	0
kunstleri	FDK2	D22	Male	53	0
Kanotion	FDK3	D7	Female	10	0
	FDK4	D22	Female	50	0
	FDB1	D7	Male	56	0
bilobata	FDB2	D22	Male	58	0
	FDB3	D7	Female	28	0
motlevana	FDM1	D7	Female	52	0
moloyana	FDM2	D22	Female	50	0
intermedia	FDI1	D7	Male	54	0

Table 5.3. The contents of aborting figs collected from transplanted trees placed around male var. *angustifolia*.

5.4.1.2 Transplant experiment 2

The transplanted *F. deltoidea* were present at Terengganu Germplasm Centre (TGC) for three months. The number of plants available varied between varieties (Table 5.4). At the end of this period all the figs on the 66 fig trees were counted and examined for evidence of fig wasp entry and pollination. The number of figs on individual plants ranged from 0 (one of the var. *kunstleri*) to over 5000 (a var. *bilobata* individual).

The developmental phases of the figs on the recipient male plants (Figure 5.7) showed that only var. *angustifolia* had been entered by pollinators because some C was present. The only developmental phases on plants of varieties other than var. *angustifolia* were either AB (immature or receptive to pollination) or aborting. AB phase male figs were present on all varieties. This included 60.07% of the figs on male trees of var. *angustifolia*. C phase figs were only found on male trees of var. *angustifolia* with 1.67% of the total figs present. No DE phase was present on all varieties. Aborting figs were present in all varieties. This included 38.26% of the figs on the var. *angustifolia* plants. Among the donor var. *angustifolia* C and DE figs ranged from 4.79% to 65.54% of the total figs on different trees (Table 5.4. The entry into figs of *Ficus deltoidea* varieties in the Terengganu Germplasm Collection (TGC) after sources of var. *angustifolia* fig wasps were added.).

Results for female trees were different to those of the males. None of the female trees showed any evidence of pollination, including recipient female var. *angustifolia* (Figure 5.8). The only phases recorded on all female trees were AB (young and receptive) together with aborting figs. The highest proportion of AB phase figs was in var. *bilobata* (69.6%) while the highest percentage of aborting figs was recorded in var. *kunstleri* (98.2%). This indicate that var. *bilobata* may have only had receptive figs available towards the end of the experiment.

Variety	Tree No.	Sex	Total number of figs present on the plants during the experiment		F	igs
			Beginning	End	Entered	Un-entered
angustifolia	FDA 1	Male	2175	2943	1031	1912
	FDA 2	Male	201	2876	509	2367
	FDA 3	Male	0	2533	288	2245
	FDA 4	Male	0	2333	1529	804
	FDA 5	Male	0	2899	139	2760
	FDA 6	Female	1652	1673	0	1673
	FDA 7	Female	1767	1266	0	1266
	FDA 8	Male	0	1939	0	1939
	FDA 9	Male	0	2071	213	1858
	FDA 10	Male	0	2989	0	2989
	FD 11	Male	0	3787	0	3787
	FDA 12	Male	0	54	0	54
	FDA 13	Female	0	1470	0	1470
deltoidea	FDD 1	Male	2430	1922	0	1922
	FDD 2	Male	493	597	0	597
	FDD 3	Female	54	84	0	84
	FDD 4	Female	248	294	0	294
	FDD 5	Male	0	1574	0	1574
	FDD 6	Male	0	230	0	230
	FDD 7	Male	0	572	0	572
	FDD 8	Male	0	369	0	369
	FDD 9	Male	0	22	0	22
	FDD 10	Female	0	530	0	530
	FDD 11	Female	0	2337	0	2337
	FDD 12	Female	0	420	0	420
	FDD 13	Female	0	747	0	747
	FDD 14	Female	0	313	0	313
trengganuensis	FDT 1	Male	215	222	0	222
	FDT 2	Male	76	96	0	96
	FDT 3	Female	836	867	0	867

Table 5.4. The entry into figs of *Ficus deltoidea* varieties in the Terengganu Germplasm Collection (TGC) after sources of var. *angustifolia* fig wasps were added.

	FDT 4	Female	202	183	0	183
	FDT 5	Male	0	1183	0	1183
	FDT 6	Male	0	885	0	885
	FDT 7	Male	0	217	0	217
	FDT 8	Male	0	1019	0	1019
	FDT 9	Male	0	319	0	319
	FDT 10	Female	0	304	0	304
	FDT 11	Female	0	592	0	592
	FDT 12	Female	0	440	0	440
	FDT 13	Female	0	189	0	189
	FDT 14	Female	0	1229	0	1229
kunstleri	FDK 1	Male	0	0	0	0
	FDK 2	Male	145	350	0	350
	FDK 3	Female	10	25	0	25
	FDK 4	Female	125	94	0	94
	FDK 5	Male	0	210	0	210
	FDK 6	Female	0	806	0	806
	FDK 7	Female	0	891	0	891
	FDK 8	Female	0	700	0	700
	FDK 9	Female	0	1441	0	1441
	FDK 10	Female	0	686	0	686
bilobata	FDB 1	Male	4591	3281	0	3281
	FDB 2	Male	4672	4809	0	4809
	FDB 3	Female	35	217	0	217
	FDB 4	Male	0	3361	0	3361
	FDB 5	Male	0	4703	0	4703
	FDB 6	Male	0	5110	0	5110
	FDB 7	Male	0	3490	0	3490
	FDB 8	Male	0	2861	0	2861
motleyana	FDM 1	Female	82	102	0	102
	FDM 2	Female	124	155	0	155
	FDM 3	Female	0	366	0	366
	FDM 4	Female	0	317	0	317
	FDM 5	Female	0	97	0	97
	FDM 6	Female	0	159	0	159
intermedia	FDI 1	Male	135	92	0	92



Figure 5.7 The developmental phases of male *Ficus deltoidea* figs recorded at the end of experiment two. Hashed bars = AB phase, Black bar = C phase, open bar = DE, grey bars = aborting.



Figure 5.8 The developmental phases of female *Ficus deltoidea* figs recorded at the end of experiment two. Hashed bars = AB phase, grey bars = Aborting. No C or DE phase figs were recorded.

Among the ten male var. *angustifolia* plants, five had figs containing fig wasp offspring at the start of experiment two (Table 5.2). Between 2.1% and 99.0% of their figs had been entered at the start of the experiment (Figure 5.9). At the end of the experiment all the five male trees that had initially had fig wasps continued to have entered figs present (the originally occupied figs had completed their development and fallen from the trees by this time). Male plants with high entry rates at the start of the experiment tended to be the ones that also had many entered figs at the end of the experiment (Figure 5.9).

Among the remaining five male plants, FDA 9 was the only individual to acquire pollinators during the course of the experiment, with 213 of its 2071 figs entered by pollinators. FDA 9 had the highest density of pollinator-releasing trees near it of the five trees, and was located less than 2 meters from four of the male plants that were releasing pollinators (FDA 1, FDA 2, FDA 3, and FDA 4). The three female trees (FDA 6, FDA 7 and FDA 13) were entirely un-pollinated at the end of experiment two Table 5.4). As in experiment one, aborting figs from each plant of each variety were dissected, with sample sizes varying according to availability (Table 5.5). This check for whether or not any fig wasp pollinators had entered the figs found that neither wings nor wasp bodies were present in the ostioles and central cavities of any of the aborting figs.



Figure 5.9. Initial and final entry rates on the male trees of var. *angustifolia* in experiment two. Solid bars = entry rates at the start of the experiment, open bars = entry rates at the end of the experiment. FDA 1, FDA 2, FDA 3, FDA 4 and FDA 5 had previously-entered figs at the start of the experiment. FDA 9 was the only control tree with figs that were entered.

Variety	Tree No	Sex	No. of figs collected	Presence of wings/ wasps
	FDD 1	Male	55	0
	FDD 2	Male	52	0
	FDD 3	Female	56	0
	FDD 4	Female	57	0
	FDD 5	Male	55	0
	FDD 6	Male	56	0
daltaidaa	FDD 7	Male	58	0
denoidea	FDD 8	Male	53	0
	FDD 9	Male	22	0
	FDD 10	Female	52	0
	FDD 11	Female	53	0
	FDD 12	Female	59	0
	FDD 13	Female	57	0
	FDD 14	Female	57	0
	FDT 1	Male	54	0
	FDT 2	Male	55	0
	FDT 3	Female	52	0
	FDT 4	Female	56	0
	FDT 5	Male	52	0
	FDT 6	Male	55	0
trenaganuensis	FDT 7	Male	56	0
trengganaensis	FDT 8	Male	55	0
	FDT 9	Male	54	0
	FDT 10	Female	58	0
	FDT 11	Female	59	0
	FDT 12	Female	52	0
	FDT 13	Female	55	0
	FDT 14	Female	56	0
	FDK 1	Male	0	0
kunstleri	FDK 2	Male	50	0
	FDK 3	Female	22	0

Table 5.5 The contents of aborting figs collected from recipient plants in experiment two.

	FDK 4	Female	51	0
	FDK 5	Male	52	0
	FDK 6	Female	55	0
	FDK 7	Female	56	0
	FDK 8	Female	59	0
	FDK 9	Female	55	0
	FDK 10	Female	51	0
	FDB 1	Male	55	0
	FDB 2	Male	54	0
	FDB 3	Female	56	0
bilobata	FDB 4	Male	55	0
	FDB 5	Male	53	0
	FDB 6	Male	57	0
	FDB 7	Male	55	0
	FDB 8	Male	54	0
	FDM 1	Female	52	0
motleyana	FDM 2	Female	55	0
	FDM 3	Female	50	0
	FDM 4	Female	57	0
	FDM 5	Female	58	0
	FDM 6	Female	52	0
intermedia	FDI 1	Male	55	0
	FDI 2	Male	37	0

5.4.1.3 Transplant experiment 3

This experiment had a more systematic placement of the experiment plants, but produced similar results to experiment two. Only var. *angustifolia* had C phase figs, which indicated the entry of pollinators. About 70% of the male figs in the recipient trees of this variety had been entered, but no figs produced by the other varieties had been entered (Figure 5.10). A similar result was found with female figs, with only figs of var. *angustifolia* pollinated, but the percentage of entered figs was only about 18% (Figure 5.11).

Between zero and 56 aborting figs were examined from each variety, depending on availability (Figure 5.6). One individual of var. *motleyana* and one of var. *intermedia* did not produce any figs. There was no sign of wings or pollinator bodies being present in any of the dissected figs.



Figure 5.10 The phases of development of male figs placed around var. *angustifolia* that were releasing pollinators, scored at the end of experiment three. Hashed bars = AB phase, black bars = C phase, grey bars = aborting figs. No DE phase figs were recorded.



Figure 5.11 The phases of fig development in female figs placed around var. *angustifolia* that were releasing pollinators, scored at the end of experiment three. Hashed bars = AB phase, black bars = C phase, grey bars = aborting figs. No DE phase figs were recorded.

Variety	Tree No.	Donor tree	Sex	No of figs collected	Figs with wings/wasps
	FDD1	FDA 1	Male	52	0
deltoidea	FDD2	FDA 9	Male	55	0
denoidea	FDD3	FDA 1	Female	50	0
	FDD4	FDA 10	Female	53	0
	FDT1	FDA 2	Male	55	0
tronggonuonsis	FDT2	FDA 9	Male	56	0
trenggandensis	FDT3	FDA 2	Female	55	0
	FDT4	FDA 10	Female	51	0
	FDB1	FDA 3	Male	55	0
bilobata	FDB2	FDA 9	Male	53	0
	FDB3	FDA 3	Female	54	0
	FDK1	FDA 4	Male	4	0
kunstlari	FDK2	FDA 9	Male	1	0
Kunstien	FDK3	FDA 4	Female	7	0
	FDK4	FDA 10	Female	21	0
motlevana	FDM1	FDA 5	Female	0	0
motleyana	FDM2	FDA 10	Female	11	0
intermedia	FDI1	FDA 5	Male	0	0
intermedia	FDI 2	FDA 9	Male	37	0

Table 5.6. The contents of aborting figs collected from potted trees placed around male var. *angustifolia* that were releasing pollinators.

5.4.1.4 Transplant experiment 4

This experiment was similar to experiment three in terms of the experimental design. However, the receptive figs on all the donor trees were removed before this experiment started. This forced the fig wasps to fly away from their natal trees and make choices about entering figs elsewhere. Under these conditions, small numbers of the emerging female fig wasps were found to enter male figs of another variety as well as male figs of var. *angustifolia* (Figure 5.12). Some of the figs on the donor var. *angustifolia* had reached receptivity and being entered at the end of the experiment.

The fig wasps had entered 3 figs from a total of 39 figs of var. *deltoidea*. The three entered figs of var. *deltoidea* were located on a branch that was touching the donor tree. All three entries were from cluster F of the experiment, where the combination of adjacent plants did not include any control var. *angustifolia*. The three figs of var. *deltoidea* that had been entered were dissected. Wings were found in each fig. The ovules had been galled, but the galls were all empty with no sign of larval development. The entered figs were in mid C phase and usually, in that phase, the larvae of fig wasps can be seen clearly when galls are dissected.

For the female figs, the results were similar to those of the three previous experiments (Figure 5.13). No variety apart from var. *angustifolia* showed signs of fig wasp entry or pollination. From zero to 56 aborting figs were dissected. No aborting figs showed the presence of the wings and fragments of fig wasp body (Table 5.7).

Some of the male figs from var. *angustifolia* had already reached DE phase. Based on personal observations, the development of figs on the potted trees was fast compared to the wild epiphytic fig trees. The nutrients provided by the media might have caused this, or higher temperatures.



Figure 5.12 The phases of fig development in male figs recorded at the end of experiment four. Hashed bars = AB phase, black bars = C phase, grey bars = aborting figs. No DE phase figs were recorded.



Figure 5.13 The phases of fig development in female figs recorded at the end of experiment four. Hashed bars = AB phase, black bars = C phase, white bars = DE phase, grey bars = aborting figs.

Variety	Tree No.	Donor tree	Sex	No of figs collected	Figs with wings/wasps
	FDD1	FDA 1	Male	55	0
deltoidea	FDD2	FDA 9	Male	39	3
	FDD3	FDA 1	Female	52	0
	FDD4	FDA 10	Female	54	0
	FDT1	FDA 2	Male	54	0
trengganuensis	FDT2	FDA 9	Male	53	0
	FDT3	FDA 2	Female	52	0
	FDT4	FDA 10	Female	54	0
	FDB1	FDA 3	Male	54	0
bilobata	FDB2	FDA 9	Male	53	0
	FDB3	FDA 3	Female	55	0
	FDK1	FDA 4	Male	54	0
kunstlari	FDK2	FDA 9	Male	51	0
kunstien	FDK3	FDA 4	Female	12	0
	FDK4	FDA 10	Female	55	0
motlevana	FDM1	FDA 5	Female	54	0
moucyana	FDM2	FDA 10	Female	55	0
intermedia	FDI1	FDA 5	Male	55	0
intermedia	FDI 2	FDA 9	Male	38	0

Table 5.7. The contents of aborting figs collected from potted trees placed around pollinated male var. *angustifolia* in Experiment 4.

From a total of four donor trees, the fig wasps entered figs on all four male recipient plants, but only three of the female recipients (Figure 5.14). Pollination rates of female var. *angustifolia* figs were lower than in males. Between 5-45% of the male figs were entered compared 0 to 10% of the figs on female trees (Figure 5.14). The crop sizes of male recipient var. *angustifolia* ranged from 11 to 2115 figs while in female recipient trees the range was from 175 to 1984 figs.

There was a negative relationship between crop size of the recipient male trees and the percentage of available figs entered (Figure 5.15). A similar relationship was present in female figs, though the proportion of figs pollinated was never high. These results reflect the small numbers of pollinators that were present. No figs on female plants were entered by pollinators emerging from donor FDA 1 but this tree had 25% of total number of entered figs in male tree. The highest number of entered figs comes from FDA 2 with 106 figs in the male tree. This compares with 34, 18 and 58 figs around donor trees FDA1, FDA3 and FDA5 respectively.

The number of entered figs varied from 5 to 106 figs on recipient male trees while on female trees the range was from zero to 58 figs (Figure 5.16). The male trees around donor trees FDA 1 and FDA 2 had a higher total number of figs entered than female trees, whereas more pollinators entered more female figs than male figs around donor trees FD3 and FD5 (Figure 5.16). However, the proportion of male figs entered was at least slightly higher than with female figs around all four donor trees (Figure 5.17). In terms of proportions of available figs, the figs entered on male trees ranged from 0.05 to 0.46 while on female trees, the range was from 0 to 0.11. The highest proportion of entered figs was around donor tree FDA 5, where 5 out of 11 male figs were entered. Chi square tests showed there was a significant difference between entered and not entered figs in male and female figs from donor tree FDA 1 (χ^2 = 420.19, df = 1, P < 0.05), FDA 2 (χ^2 = 115.35, df = 1, P < 0.01) and FDA 4 (χ^2 = 10.25, df = 1, P < 0.01). Around donor tree FDA 3, the difference was not significant (χ^2 = 0.91, df = 1, P > 0.01).



Figure 5.14 The developmental phases of figs on male and female var. *angustifolia* located around four different donor trees (FDA 1-3 & 5). Hashed bars = AB phase, black bars = C phase, white bars = DE phase, grey bars = aborting figs. Four male and four female plants were positioned around each donor tree.



Figure 5.15. The relationship between recipient tree crop size and the percentage of entered figs from different donor trees. (Grey colour = male trees, white colour = female trees, (\blacktriangle) = donor tree FDA 1, (\blacklozenge) = FDA 2, (\blacksquare) = FDA 3, (\bullet) = FDA 5).



Figure 5.16. The numbers of entered male and female figs around different donor trees (Solid bars = male recipient trees, open bars = female recipient trees). No female figs were entered around tree FDA1.



Figure 5.17 The relationship between the proportion of entered figs and different donor trees. (Solid bars = male trees, open bars = female trees).

Host choice by Blastophaga sp. (var. deltoidea)

From 9-18 receptive figs of var. *angustifolia* were in the fine mash bags and available for entry by the single *Blastophaga* sp. (var. *deltoidea*) female that was placed in each (Table 5.8). The figs were brought to the Lab. fourteen days later to check for wings and foundress bodies. None of the confined figs were entered by the introduced foundresses, with no sign of wings and wasp bodies in the confined figs. Similar experiments carried out using pollinators obtained from figs of var. *angustifolia* routinely resulted in pollinator entry (Chapter 7).

Tree	Sex	Bag	Fig	js	Figs with
		-	Receptive	Entered	wings/wasps
FD1	Female	1	12	0	0
		2	14	0	0
		3	13	0	0
		4	9	0	0
		5	15	0	0
		6	10	0	0
FD6	Male	1	14	0	0
		2	13	0	0
		3	11	0	0
		4	11	0	0
		5	13	0	0
FD7	Male	1	17	0	0
		2	18	0	0
		3	15	0	0
		4	13	0	0
		5	16	0	0
FD22	Male	1	11	0	0
		2	15	0	0
		3	15	0	0
		4	14	0	0
		5	11	0	0
FD40	Female	1	15	0	0
		2	18	0	0
		3	16	0	0
		4	14	0	0
		5	9	0	0
		6	11	0	0

	lumbers of confined figs of var. angustifolia a	vailable for entr
--	---	-------------------

5.5 Discussion

Ficus deltoidea often exhibits variable leaf and fig characters that have made delimitation of the varieties difficult. Because of intermediates in morphology, Berg and Corner (2005) considered all variation within *F. deltoidea* to be within a single species, but more recent morphological studies conducted by Mat et al., (2012) and Nur Fatihah et al. (2014) confirmed that varieties can vary greatly in their leaf morphology. Whether the varieties are actually distinct biological species is unclear. Whether varieties have different species of host specific associated pollinators provides an indication of likely barriers to gene flow between them. Among the factors that maintain specificity in the fig and fig wasp mutualism are fig wasp behavioural responses and morphological adaptations towards the structure of the figs (Liu et al., 2013). In closely related fig tree species, pollinator host specificity is maintained by combinations of long range cues from floral scent, short-range contact cues and physical matching between the fig wasp ensures that genetic integrity is maintained (van Noort, 2003).

5.5.1 Host choice by Blastophaga sp. (var. angustifolia)

5.5.1.1 Host preference between varieties

In Experiment one, all recipient trees of variety *angustifolia* had some figs where seeds or pollinator offspring were developing, but no similar development was recorded among figs of any other varieties. Dissections of figs showed there was no sign of fig wasp females entering the figs of these varieties, so their lack of development was the result of pollinators from var. *angustifolia* failing to enter, rather than an inability to gall flowers of male figs or to pollinate flowers in female figs. This result suggested that figs of different varieties may emit different attractant volatiles and that pollinator sharing by different varieties is likely to be rare, even when they are living sympatrically (Moe et al., 2011).

In the second experiment, competition for pollinators from receptive var. *angustifolia* figs was much lower than in experiment one, because far fewer receptive figs of the usual host were available for entry. However, far fewer pollinators were also being released in the area. Under these conditions, only male trees of var. *angustifolia* were pollinated, and no females of this variety or individuals of either sex belonging to other varieties were entered by pollinators. This suggests that pollinator specificity can be

maintained away from natural conditions (Cornille et al., 2012), but the results were limited by the small number of pollinators that dispersed from their natal figs

Pollination rates among male var. *angustifolia* were higher among those plants that had fig wasps emerging from them and male FDA 9 was the only plant to acquire pollinators that had certainly flown from other male plants only a metre or so away. This result has parallels with a study by van Noort (2003) where he recorded that on an African tree with asynchronous fig development, the fig wasps tended to pollinate figs on their natal trees and that this reduced their flying costs as they only needed to fly a short distance.

Some fig wasps are able to move pollen between flowering fig trees that are separated by many kilometres (Harrison and Rasplus, 2006). Ahmed et al. (2009) showed that *Ceratosolen arabicus* Mayr females can reach trees with receptive figs160 km from their natal trees, even if the flight period of fig wasps can last for just a few hours (Liu et al., 2013). However, Duthie and Nason (2016), suggest that foundress density increases when tree connectivity is higher, because pollinator mortality will increase with distance travelled. The fig wasps from monoecious trees often have wider dispersal ranges compared to the fig wasps associated with dioecious fig tree species such as *F. deltoidea* (Borges 2016).

The entry of small numbers of fig wasps from var. *angustifolia* into var. *deltoidea* figs in experiment four showed that under these experimental conditions the fig wasps will enter figs of other varieties. Most of the pollinators did choose to enter figs of their typical host. It seems likely that emerging females enter the first receptive figs they encounter, which are usually the figs on their natal trees. However, in this experiment there were no receptive figs on the donor trees and this made the fig wasps need to fly to the other trees to find receptive figs. The figs of var. *deltoidea* are the most like var. *angustifolia* in their external appearance and size, compared to the other varieties, and this might be the reason why the fig wasps entered the figs of var. *deltoidea* rather than the other varieties (Chapter 4). Alternatively, the volatiles released by this variety may be the most similar to those of var. *angustifolia*. However, a conclusion cannot be made as the fig wasps might have entered this particular variety because its figs were at their peak attractiveness whereas other varieties may have had figs that were a little older or younger.

Regardless of why only figs of var. *deltoidea* were entered, this result showed that the var. *angustifolia* pollinator can enter male var. *deltoidea* figs, and that the ostiole does not present a physical barrier. The fig wasps were also able to gall the flowers. Galling is initiated at the time of oviposition, so eggs were probably laid in these figs, because the style lengths of the flowers are suitable for this pollinator (Ghana et al., 2015a) (Chapter 4). The initial formation of galls requires the injection of a secretion from the foundress into the ovules. The gall tissue later become the source of food for the developing larva (Martinson et al., 2014). However, even though galls developed, no sign of larval development were detected in any of the galls in the three figs of var. *deltoidea*. They had become empty galls (bladders) and clearly if eggs had been laid than any larvae were unable to develop successfully. This will generate a strong selection pressure to avoid male figs var. *deltoidea* in areas where the two varieties grow close together.

The specificity of the interaction between *Blastophaga* sp. and var. *angustifolia* is maintained because few pollinators were willing to enter var. *deltoidea* figs and no larvae successfully developed inside the figs of var. *deltoidea*. This in turn will make it also less likely that they will enter female figs of var. *deltoidea* and this will cut down the chances of hybrids being formed. None of the female figs of var. *deltoidea* were pollinated, so whether hybrid seeds are viable is unknown. Indications that the pollinator of var. *deltoidea* similarly avoids figs of var. *angustifolia* are provided after the following section.

5.5.1.2 Preferences between male and female plants

Because of the relative costs of seed and pollen production, in plants in general, malefunctioning individuals are expected to allocate more resources to secondary floral structures and products than female individuals (Delph and Lively, 1992). In *Ficus*, the pollinator is not expected to be able to distinguish the sex to maintain the mutualism. Otherwise the pollinator could turn into a parasite that avoids female trees (Kjellberg et al., 1987). The conflict between what is best for the fig tree and what is best for the pollinators is thought to have led to inter-sexual mimicry, with figs on male and female plants producing the same volatiles and other cues to make it difficult for pollinators to avoid female figs (Soler et al., 2002; Hossaert-McKey et al., 2016). This mimicry is strengthened by vicariant selection, because it is in the interest of both male and female plants that the pollinators cannot distinguish between male and female figs (Grafen and Godfray, 1991). Almost all the control plants (var. *angustifolia* with receptive figs and no emerging fig wasps) had at least some of their figs entered by pollinators. The frequency of pollination was lower on female plants. This suggests that the pollinator of var. *angustifolia* can detect differences between figs on male and female plants and prefers the cues provided by male figs. Entering a female fig guarantees that a foundress fig wasp will produce no offspring (Cook and Rasplus, 2003; Borges, 2016) but benefits the natal male tree from where the fig wasp emerged, because its pollen will be generating seeds. Conversely, a fig wasp that enters a male fig can reproduce but provides no benefits to its natal tree if it entered figs on another male tree. If it enters figs on its natal tree then there are nonetheless postponed potential benefits because one foundress can produce numerous offspring, each of which has a chance to fly away and enter female figs.

Several studies have found that male and female figs release similar volatiles and pollinators cannot distinguish between them (Hossaert-McKey 2016; Rodriguez et al. 2017). However, in *Ficus carica* L. the pollinators prefer to enter male figs rather than female figs (Anstett et al., 1998). It was suggested that in this species inter-sexual mimicry is not important, because very few receptive male figs are available at the time of year when female figs are receptive (Hossaert-McKey et. al., 2016). This is not the case with var. *angustifolia*, which produces figs asynchronously on both male and female trees, so this hypothesis fails to explain the preference shown by its pollinator for male figs. The mean diameter of receptive male and female figs is similar (Chapter 4) so the figs seem to offer similar visual cues. A difference in volatiles may be involved.

5.5.2 Host choice by Blastophaga sp. (var. deltoidea)

A molecular study of four varieties of *Ficus deltoidea* (var. *angustifolia*, var. *bilobata*, var. *trengganuensis* and var. *kunstleri*) suggested that there is gene flow among varieties of this species (Zimisuhara et al., 2015). Based on the behaviour of the pollinators of var. *angustifolia* and var. *deltoidea* any gene flow between these two varieties is likely be very limited if it occurs at all, because pollinators of both varieties were generally unwilling to enter figs of other varieties.

5.6 Conclusions

With its asynchronous crops, var. *angustifolia* often offers the chance for populations of its pollinators to cycle on a single male tree. Fig wasps tend to enter the first figs they have encountered regardless of the sexes and diameter of the figs (Moore et al., 2003). Taking time to make a choice of which fig to enter will cost them time and thus increase the possibility of reproductive failure (Moore et al., 2003). The pollinators of var. *angustifolia* are reluctant to fly away from their natal trees, even if receptive figs on other trees are only a metre or two away. If they do disperse from their natal trees they prefer to enter figs on male trees. This behaviour clearly benefits the insects, and has the consequence that pollinator limitation of seed production is very strong in natural populations (Chapter 6).

Chapter 6

Phenology of *F. deltoidea* var. *angustifolia* in a Malaysian oil palm plantation

6.1 Introduction

6.1.1 Phenology

The tracking of suitable environmental conditions through time is an essential adaptation for plants and animals (Diekmann, 1996). The word phenology is derived from Greek 'phainein' which means to show or appear (Fenner, 1998). Phenology is the study of the periodical (often seasonal) timing of life cycle events. In plants this commonly includes events such as flowering, fruiting, leaf production and leaf fall and when they occur in relation to biotic and abiotic factors (Harrison et al., 2000). Phenology can be considered within an individual and within a population (Elzinga et al., 2007). Phenology studies usually integrate variation within plants and populations with variations of climate (Visser et al., 2010). The timing of vegetative growth and reproduction is an important adaptation to seasonality and can be predictable from year to year but variation in temperature and other climatic variables such as rainfall can generate variation between years in flowering times (Inouye, 2008).

The study of phenology can deliver information on many aspects of plant biology, including development and patterns of plant growth and the extent of the influence of environment and selective pressures on flowering and fruiting behaviour (Zhang et al., 2009). Data on phenology can also be used in land-use planning, agricultural control, protection of species and notification of pollen release, which has implications for

human health (Ruml and Vulic, 2005). Long term observations on phenology can also be used to discover patterns resulting from climate change.

Temperature and the availability of water and light are often the critical environmental factors that determine patterns of phenology (Ulian et al., 2013; Krishnan et al., 2014). Phenological patterns reflect both responses to environmental triggers and the selection pressures that have favoured these responses. These include both abiotic and biotic influences such as those generated by pollinators and other mutualists.

6.1.2 Ficus phenology

The phenological patterns of fig trees, particularly those involving fruiting and leaf production, are central to their pollination biology and broader ecological significance for animals. Fig trees (*Ficus*, Moraceae) are involved in a highly specific and often one-to-one relationship with pollinating fig wasps (Agaonidae) that breed in their figs. Consequently, their fruiting phenology has a critical influence on populations of their pollinators. Any mismatch in timing will then have a detrimental effect on seed production as well as their pollinators (Zhao et al, 2014). Comparative phenological studies can reveal how the mutualism functions and how it has evolved (Bronstein and McKey, 1989; Patel, 1996)

Most phenological studies have involved monoecious rather than dioecious fig tree species (Herre, 1989; Harrison et al., 2000). Monoecious fig trees are often characterised by having synchronous fruiting within individual trees and asynchrony between trees in a population (Herre et al., 1996). Dioecious fig trees respond differently to seasonal variability than monoecious figs due to the separation of seed and wasp production between trees (Kjellberg and Maurice, 1989). In dioecious species there are often phenological differences between male and female individuals, but there needs to be reproductive coordination between the sexes (Zhao et al, 2014).

Relationships between leaf and fruiting phenology are variable. Individuals of some species do not produce young leaves and figs at the same time (Compton, 1993) while others produce young leaves and figs simultaneously (Peng et al., 2009). Some trees have synchronous leaf production but asynchronous fruiting phenology. For example, *Ficus racemosa* Linn. in China can display synchronous flushing and senescence of leaves, but they produce new figs all year long (Zhang et al., 2006). Local

environmental conditions can alter phonological patterns. An example is where drought can trigger fig trees to lose their leaves to minimize water loss (Compton, 1993).

The phenology of fig trees is likely to be different in seasonal and aseasonal habitats, with more synchrony in the former (Spencer et al., 1996). A common pattern in monoecious species is for figs to be produced throughout the year with seasonal peaks during the warmest periods and low numbers of tree fruiting during cold seasons (Kjellberg and Maurice, 1989). This allows populations of their pollinators to be maintained throughout the year. In dioecious species seasonal or dry environments should favour fig production by female trees to be concentrated during periods when fruit dispersal and seed germination and likely to be most successful (Patel, 1996). For example, (Patel and McKey, 1998) showed that female trees of *Ficus exasperata* Vahl. and *Ficus hispida* L.f. have more mature figs during the wet season when conditions are most suitable for germination.

Female trees have no role in maintaining pollinator populations, though they do act as traps for the pollinators that they attract (Suleman et al., 2011). The reproductive success of male trees depends on their production of pollen-carrying fig wasps that enter figs on female trees. Reflecting this, male trees of *F. hispida* produce more figs during the dry and early wet season and male *F. exasperata* release fig wasps at the time of peak production of receptive female figs (Patel and McKey, 1998). Limited out-of-peak-season fig production on male trees ensures that their pollinator populations are maintained throughout the year.

The effects of seasonal variation are usually less prominent close to the equator and this leads to more asynchrony in leafing and fruiting at individual and population levels (Galil and Eisikowitch, 1968; Zhang et al., 2006). This results in temporal overlap between the sexes in dioecious species (Patel, 1996) with mature and receptive figs produced continuously throughout the year (Suleman et al., 2011). The need to maintain pollinator populations throughout the year helps makes monoecious fig trees keystone in tropical areas as they provide food to frugivores at times when the other plants are not fruiting (Shanahan et al., 2001). Male dioecious fig trees must also maintain fig wasp populations, but their figs are not normally eaten by vertebrates. Figs of female dioecious fig trees are attractive to vertebrate frugivores, but their greater seasonality may reduce their significance as keystone resources.

6.1.3 Ficus deltoidea

Ficus deltoidea is a small dioecious fig tree from SE Asia (Corner, 1969). It is highly variable and numerous varieties have been described, some of which are likely to be different biological species. It is unusual within its genus because its female figs contain an unusually small number of flowers that generate atypically large seeds. It is also unusual because some varieties grow as true epiphytes. *Ficus deltoidea* also displays extreme variation in leaf forms between varieties and can also produce leaves that differ in shape on young and mature individuals (Mat et al., 2012).

Hetrophylly is defined as being variable leaf morphology within a single species. It is often exhibited in response to varying environmental conditions (Kuwabara et al., 2003). Variable environmental factors that promote heterophylly include light intensity and quality, ambient temperatures and water availability (Nakayama et al., 2017). Plant hormones involved in heterophylly are ethylene that influences leaf elongation (Kuwabara and Nagata, 2006), gibberellins that cause expansion of the leaf lamina (Sun, 2010) and auxins that generate outgrowths of the leaf lamina (Scarpella et al., 2010).

In *F. deltoidea,* heterophylly is reflected in the production of leaves of different size or shape. Elongate and deltoid leaves are produced. Juvenile individuals tend to have elongate leaves while mature trees tend to have deltoid leaves or a mixture of the two (S.G. Compton personal communication). Reasons for the change in leaf form with maturity are unknown but having fewer leaves with more surface area may help small trees to capture light and carbon (Pearcy et al., 2004). Plants with only juvenile leaves rarely produce figs under field conditions but will do so in containers (personal observations of var. *angustifolia*). Plants grown from cuttings do not revert to elongate leaves (S. G. Compton personal communication).

Corner (1969) recorded elongate juvenile leaves in several varieties of *F. deltoidea* (var. *angustifolia*, var. *deltoidea*, var. *intermedia* and var. *motleyana*). Mat et al. (2012) similarly recorded elongated juvenile leaves in var. *angustifolia*, var. *bilobata*, var. *intermedia* and var. *trengganuensis*. No other study of heterophylly was conducted after that.

6.2 Objectives

The phenology of any variety of *F. deltoidea* has not been recorded previously. We (1) examined the leaf and fruiting phenology of var. *angustifolia*, a variety that exhibits heterophylly, when growing as an epiphyte in an oil palm plantation in Peninsular Malaysia. (2) Examined the extent of within and between individual asynchrony in fig production and how crop and fig wasp population sizes responded to seasonal changes. (3) Asked whether the phenology of male and female plants differs and (4) recorded the extent to which seed production is limited by pollinator availability and whether more isolated female trees were more likely to suffer from a shortage of pollinators.

6.3 Methods

6.3.1 Study sites and sampling methods

In this study 39 dioecious *F. deltoidea* var. *angustifolia* individuals were monitored every two weeks for 15 months from 13th June 2016 until 21st August 2017. This provided a total of 34 dates when 21 male fig trees and 18 female fig trees were observed to determine their leafing and fruiting phenology. All the plants were epiphytes in an oil palm plantation at Banting in Selangor, Peninsular Malaysia. They were mature trees producing figs and were growing less than 2.5 meters from the ground, which provided easy access. The rest of the epiphytes either did not produce fruit or were too high to be assessed. The sexes of the plants were determined by opening a few figs on each and observing the number of female florets.

At the beginning of the study, the *F. deltoidea* var. *angustifolia* were mapped, located using GPS and tagged. The plants have a multi-stemmed, shrubby growth form that makes size estimates difficult. Measurements of the diameters of the thickest stems on each plant were made using callipers to provide an indication of plant size. The height of the basal part of each tree was recorded using an adjustable pole and measuring tape. Climate data (minimum and maximum temperatures, monthly rainfall totals) were obtained from the nearest meteorological station (KLIA) located 19 km from the site. Details were provided in Chapter 2. The observations were done visually. Visual assessments of the leaves and figs on the plants were made during each visit.
In some cases, the leafing and fruiting phenology of two plants growing on the same trunk was recorded.

6.3.2 Leaf phenology

During each visit the presence or absence of elongate (juvenile) and adult leaves was recorded (Figure 6.1) and leaf condition was scored as new, mature or senescent. Elongate leaves are produced by seedlings and young plants and are gradually replaced by smaller, more rounded leaves as plants get older. The proportion of leaves that were juvenile was assessed by visual inspection. The stage of development of the juvenile and adult leaves (combined) was also scored by visual inspection. New leaves were smaller and paler green in colour. Mature leaves were larger and darker green. Senescent leaves were yellow. If less than ten senescent leaves were present this state was recorded as 'rare'. Bare stems were also recorded. Juvenile leaves also scored as young, mature or senescing.

6.3.3 Fruiting phenology

The approximate numbers of figs on each tree were counted and the developmental phases were recorded. *F. deltoidea* produces two figs at the axil of each other. The sizes of fig crops were scored as being one of the followings <10, 11- 50, 51-100, 101-200, 201-500, 501-1000, 1001-2000 and more than 2001. Because of their relatively small crop sizes and the usually continuous presence of figs on the trees, large scale destructive sampling of figs to record their developmental stages was not considered appropriate. Estimates of the development of the figs were therefore based mainly on non-destructive visual assessments.

Before the start of routine monitoring, figs that covered the whole range of sizes were collected from male and female plants. After noting their external appearance (size, thickness, colour) the figs were opened, and their developmental phases were compared with their appearance. This preliminary study was used in building knowledge that allowed visual assessments to be made of the development of the figs without opening them.

Fig developmental phases were recorded based on Galil and Eisikowitch (1968). The details for fig developmental phases in male and female figs were described in Chapter 2. In both male and female figs, the colours were recorded as bright green, yellowish green and yellow. A bright green colour was associated with young figs at AB phase,

yellowish green was associated with C phase while yellow usually indicated that the figs were aborting. Size can also give an indication of a fig's developmental phase. AB phase figs of var. *angustifolia* usually lie within the range of 1-4 mm, figs with a range of 4-6 mm can belong to any phase, but figs with a diameter more than 6mm are usually either C or DE phase. AB phase figs are usually soft when squeezed, as are aborting figs. During C and DE phases, the texture is typically hard.

6.3.4 Fig development cycle

At the start of the study, ten random A phase figs on each tree were marked using a permanent fine marker. The diameter of each fig was then measured fortnightly and the developmental stage of the fig was recorded based on its appearance. For figs that failed to be pollinated, the duration they remained on the tree before being aborted was also recorded.

6.3.5 Statistical methods

The relationship between the crop size categories and the thickest stem diameter for both plants sexes were correlated using Spearman rank test. Similar tests were also used for the correlation between crop size categories with the basal height from the ground. Between sexes comparisons used Mann-Whitney U tests for proportional data and Chi-squared for binary data. All tests were performed in SPSS Statistics 20.



Figure 6.1 Leaf developmental phases and juvenile leaves of *F. deltoidea* var. *angustifolia*. From left, young, mature, senescent and elongate (juvenile) leaves. The size of senescent leaves was variable.

6.4 Results

One hundred and thirteen *F. deltoidea* var. *angustifolia* were present as epiphytes on the trunks of the oil palms, with 83 producing figs during the period of the study. Among the 113 trees, only 8 produced only juvenile leaves and none of these produced any figs. A further 105 trees had a mixture of juvenile and adult leaves or only adult leaves. The exact number of trees with only adult leaves was not recorded as some of the trees were too high to assess the overall crown. The sexes of 51 trees with accessible figs were determined. There were 27 male trees and 24 female trees. From these, the leaf and fruiting phenologies of 21 males and 18 females were recorded. The sexes, diameter of the thickest stem and the height of the basal parts of these var. *angustifolia* are recorded in Table 6.1. Most of the trees survived throughout the period of observations. However, before the last three observation dates, 9 trees (7 male and 2 female) were cut off by the farmer.

6.4.1 Leaf phenology

Among the 21 male trees, 9 consistently had a mixture of juvenile and adult leaves, 7 intermittently had adult leaves in combination with juvenile leaves and 5 trees had only adult leaves throughout (Table 6.2). The juvenile leaves on a tree ranged between 0 to 60% of the total (Table 6.3). Among female trees, 10 out of 18 consistently had a mixture of juvenile and adult leaves and 5 intermittently had this mixture while only three of them never had any elongate leaves. The juvenile leaves on the trees again ranged from 0-40% of the total leaves. Male and female trees had the same frequencies of leaf types (mixed, intermittent adult only and adult only) (Chi square, $\chi^2 = 0.66$, df = 2, P > 0.05). On plants with a mixture of leaves, the location of juvenile leaves was usually at the tips of new branches, but on older branches they were usually near the base of the branch (Figure 6.2). In both sexes, nearly mature leaves started to show golden spots. Leaves facing the sun also tended to be have some yellowish colour. This was distinct from the yellow colour of the senescing leaves.



Figure 6.2. The locations of juvenile and mature leaves on an epiphytic *F. deltoidea* var. *angustifolia* tree. The yellow arrow indicates a juvenile leaf, and the red arrow a mature leaf.

No.	Sex	Diameter	Height	No.	Sex	Diameter	Height
		(cm)	(m)			(cm)	(m)
1	Female	1.7	2.02	21	Female	0.6	1.45
2	Female	2.1	1.93	22	Male	1.7	2.07
3	Male	1.2	2.27	23	Female	1.0	4.17
4	Female	2.1	2.58	24	Male	0.8	3.65
5	Female	0.9	2.88	25	Female	0.5	1.60
6	Male	0.8	2.30	26	Male	1.0	2.77
7	Male	2.0	1.46	27	Female	1.8	3.45
8	Male	1.6	2.20	28	Male	1.0	1.12
9	Female	0.7	1.62	29	Female	0.7	0.99
10	Male	1.5	2.44	30	Male	1.3	1.68
11	Female	0.7	2.57	31	Male	1.4	1.33
12	Male	1.0	1.22	32	Male	1.2	0.70
13	Male	1.7	1.35	33	Female	0.8	1.10
14	Female	1.0	0.11	34	Male	1.0	0.73
15	Female	0.2	1.20	35	Male	0.9	2.47
16	Female	1.0	0.43	36	Female	1.0	2.32
17	Female	1.3	2.08	37	Male	2.1	1.85
18	Male	0.8	0.29	38	Female	1.2	1.68
19	Male	0.8	1.56	39	Male	0.5	0.48
20	Male	0.7	2.96	1			

Table 6.1 Sexes, diameter of the thickest stem and the height of the basal parts of trees of *F. deltoidea* var. *angustifolia* in Banting oil palm plantation.

	Consistent mixture of juvenile and mature leaves	Intermittent mixture of juvenile and mature leaves	Only mature leaves
Male	9	7	5
Female	10	5	3

Table 6.2. Types of leaf on the 39 accessible var. *angustifolia* fig trees.

Table 6.3 Juvenile leaves on var. *angustifolia* in Banting plantation as a proportion of total leaves based on 34 observations at approximately 14 day intervals.

Tree no.	Sex	Juvenile leav	ves (%)	Tree no.	Sex	Juvenile leav	/es (%)
		(Mean ± SD)	Range			Mean ± SD	Range
1	Female	8.09 ± 2.47	5-10	21	Female	23.68 ± 5.68	10-30
2	Female	3.97 ± 2.05	0-5	22	Male	0	0
3	Male	2.21 ± 2.52	0-5	23	Female	1.03 ± 3.85	0-20
4	Female	0	0	24	Male	0	0
5	Female	7.5 ± 2.54	5-10	25	Female	0	0
6	Male	27.94 ± 4.10	20-30	26	Male	0.88 ± 2.60	10-20
7	Male	5.00 ± 0.00	5	27	Female	0	0
8	Male	3.82 2.15	0-5	28	Male	30.97 ± 5.95	20-60
9	Female	9.18 ± 4.35	5-20	29	Female	13.87 ± 11.88	5-30
10	Male	0	0	30	Male	4.41 ± 2.39	0-10
11	Female	7.36 ± 2.53	5-10	31	Male	2.42 ± 2.54	0-5
12	Male	12.10 ± 4.96	5-20	32	Male	3.71 ± 2.22	0-5
13	Male	2.35 ± 2.53	0-5	33	Female	26.29 ± 8.75	10-40
14	Female	8.68 ± 5.81	5-30	34	Male	27.10 ± 10.39	10-60
15	Female	2.79 ± 3.73	0-15	35	Male	0	0
16	Female	8.38 ± 2.37	5-10	36	Female	9.85 ± 6.90	0-30
17	Female	20.00 ± 6.63	10-30	37	Male	7.79 ± 2.51	5-10
18	Male	9.19 ± 1.87	5-10	38	Female	0.74 ± 1.80	0-5
19	Male	3.87 ± 2.13	0-5	39	Male	22.06 ± 4.10	20-30
20	Male	0	0				

During the 15 months of observation individual plants often had a mixture of different leaf states at the same time (combinations of bare stems, young, mature and senescing leaves). In January 2017 the farmers at Banting sprayed the ground vegetation in the plantation. The main purpose of the spraying was to eliminate weeds and the other plants around the bases of the palms. The farmers did not deliberately want to spray the fig trees but some were damaged, especially those growing low down on the trunks. The effects of the sprays are shown clearly by the number of male plants that had one or more bare stems (Figure 6.3).

Among the male trees, 19 had new leaves present every time and the remaining tree (D10) had only one time (25.05.2016) when new leaves were not present. Mature leaves were present on each male tree throughout the period. Senescent leaves were present on at least some male trees throughout the year, but there were more male trees with senescent leaves in the period around January 2017, when the spraying took place, and fewer plants with senescing leaves at the beginning and end of the survey period (Figure 6.4).

Leaf production on the female trees was similar to that of the males. Young and mature leaves were present on all 18 female trees throughout the period of observations. The numbers of female plants with bare stems increased after the spraying activities on January 2017 (Figure 6.5). The numbers of senescing leaves kept going up and down throughout the year, with no clear pattern (Figure 6.6).



Figure 6.3 Changes in the frequency of male trees having one or more bare stems. (A total of 20 male trees were observed each time). Spaying activities occurred in January 2016.



Figure 6.4 Changes in the frequency of male trees having senescent leaves. (A total of 20 male trees were observed each time).



Figure 6.5 Changes in the frequency of female trees having bare stems. (A total of 18 female trees were observed each time).



Figure 6.6 Changes in the frequency of female trees having senescent leaves. (A total of 18 female trees were observed each time).

6.4.2 Fruiting phenology

6.4.2.1 Male plants

Figs were present all the time on almost all the male trees. The single male tree that did not bear figs all the time (D39) had figs present on about half of the sample dates. This plant was small, with about 30% juvenile leaves. Crop sample sizes (the number of figs present on a tree) on male trees ranged from less than ten to over 2000. Most crops on male trees numbered between 200-500 and 500-1000 figs (Figure 6.7).

No more than two trees had less than 10 figs or more than 2000 figs at any one time. Numbers of figs on the trees increased from September 2016 until November 2016 and from May 2017 until July 2017, when a few trees were cut off by the farmers. There was no evidence that larger plants produced larger crops because there was no significant relationship between the thickest stem diameter of a plant (a measure of plant size) and crop size categories (Spearman rank correlation, $r_s = 0.39$, n = 21, P > 0.05). Crop sizes were larger on male plants that originated higher on the trunks of the oil palms (Spearman rank correlation, $r_s = 0.54$, n = 21, P < 0.05).

The production of new figs by male trees was all year around and all the male trees were producing new figs from mid-July to mid Sept-16 (Figure 6.8 and Figure 6.9). The development of the figs on most male trees was asynchronous. The D and E phase figs were routinely still on the trees when young A phase figs appeared, so emerging fig wasps could often enter figs on the same trees.

Only a few male individuals produced synchronised crops with only one or two phases present at any one time. The release of pollinators (as shown by the presence of D+E phase figs) occurred throughout the year with a peak in February and March 2017. The sudden decrease in the numbers of figs in all developmental phases during the last three observations was due to the cutting of plants by the farmers. Seven male trees were affected. Tree D22 was the only plant that had DE phase present on every visit (Figure 6.10). This was the largest male plant with a crop size of more than 2000 figs all the time. DE phase figs were only recorded on single occasions on trees D28, D34 and D39. They had only small numbers of figs throughout. The peak of fig wasp emergence was in February 2017 when 16 of the 21 male trees had DE phase figs. This was about six months after the peak in production of new figs in August 2016.



Figure 6.7 Changes over time in crop sizes on 21 male trees.



Figure 6.8 Changes over time in the fig developmental phases on 21 male fig trees.

	Ma	ıy-	Jur	า-	Ju	II-	Au	g-	Se	p-	00	ct-	N	ov-	De	ec-	Ja	n-	Fe	b-	Ма	r-	Ар	-n	Ма	y-	Ju	ın-1	7	Ju	I I -	Au	g-
	16		16		16	5	16		16		16		16	3	16		17		17		17		17		17					17	'	17	
D3																																	
D6																																	
D7																																	
D8																																	
D10																																	
D12																																	
D13																																	
D18																																	
D19																																	
D20																																	
D22																																	
D24																																	
D26																																	
D28																																	
D30																																	
D31																																	
D32																																	
D34																																	
D35																																	
D37																																	
D39																																	

Figure 6.9 The presence of young AB phase figs on male trees of *F. deltoidea* var. *angustifolia* in Banting oil palm plantation in each month between May 2016 and August 2017. (Grey bars = Figs were present, white bars = figs were absent and black bars = when the trees were cut off).

	Ma	y-	Ju	n-	Ju	Ļ,	Au	ıg-	Se	ep-	00	ct-	N	ov-	De	-O	Ja	n-	Fe	b-	Ma	ar-	Ap	-v	Ma	y-	Ju	ın-1	7	Ju	d-	Au	g-
	16		16		16)	16		16		16		16	5	16		17		17		17		17		17					17		17	
D3																																	
D6																																	
D7																																	
D8																																	
D10																																	
D12																																	
D13																																	
D18																																	
D19																																	
D20																																	
D22																																	
D24																																	
D26																																	
D28																																	
D30																																	
D31																																	
D32																																	
D34																																	
D35																																	
D37																																	
D39																																	

Figure 6.10 The presence of DE phase figs on male trees of *F. deltoidea* var. *angustifolia* in Banting oil palm plantation in each month between May 2016 and August 2017. (Grey bars = Figs were present, white bars = figs were absent and black bars = when the trees were cut off).

6.4.2.2 Female plants

The development of figs on female trees was also asynchronous. The highest frequency of crop sample size categories on female trees were 10-100 and 100-500 figs, so smaller than on the male plants (Figure 6.11). Few trees had less than 10 figs present at any one time. The trees that had 500-1000 figs had this number consistently, whereas trees with larger crops were consistent in the beginning of the study and then their crop sizes increased in March 2017 and April 2017. Almost all the 18 female fig trees produced new figs throughout the year (Figure 6.12). The numbers of plants with C phase figs showed ups and downs with the lowest numbers in December 2016 when only 8 trees showed signs having had any of their figs pollinated. The period between July and September showed the highest number of trees bearing pollinated figs in both years, with 15 trees having pollinated figs. Different to the male plants, a significant positive correlation was found between crop size and the thickest diameter of the stems (Spearman rank correlation, $r_s = 0.56$, n = 18, P < 0.05). As in the male plants, females established higher on the oil palms had larger crops (Spearman rank correlation, $r_s = 0.59$, n = 18, P < 0.05).

Half the female trees had AB young figs present all the time while the rest had new figs more than 50% of the time (Figure 6.13 and Figure 6.14). The absence of new figs on trees D14 and D15, which were located close to the ground was due to spraying activities by the farmers. As in male trees, some female trees in the last three observation periods were removed by the farmers. Tree D25 produced very low numbers of figs (less than 50) whereas tree D27 consistently had very large crops of more than 2000 figs. Three female trees that consistently had ripe figs were D1, D2 and D4 (Figure 6.14). These trees had seeds available for dispersal all the time. The highest frequency of trees with seeds available to be dispersed was in December 2016.



Figure 6.11 The crop sizes of figs on female trees.

210



Figure 6.12 Frequency of the fig developmental phases on 18 female fig trees.

	Ma	ay-	Ju	n-	Ju	۱ ۱	Au	ıg-	Se	-qe	00	ct-	N	ov-	De	ec-	Ja	in-	Fe	eb-	Ma	ar-	Ap	or-	Ma	ay-	Ju	un-	Ju	l l	Au	ıg-
	16		16)	-1	6	16		16		16) I	16	5	16		1/	′ 	17	, 	17		17		17		1.	/	-1	1	17	
D1																																
D2																																
D4																																
D5																																
D9																																
D11																																
D14																																
D15																																
D16																																
D17																																
D21																																
D23																																
D25																																
D27																																
D29																																
D33																																
D36																																
D38																																

Figure 6.13 The presence of young AB phase figs on female trees of *F. deltoidea* var. *angustifolia* in Banting oil palm plantation in each month between May 2016 and August 2017. (Grey bars = figs were present, white bars = figs were absent and black bars = when the trees were cut off).

	Ma	ay-	Ju	n-	Ju	ıl-	Αι	ıg-	Se	эр	0	ct-	N	ov-	De	-O	Ja	ın-	Fe	eb-	Ma	ar-	Ap	or-	Ма	ıy-	Ju	un-	Ju	ıl-	Au	ıg-
	16	1	16	;	16	6	16	i	-1	6	16) 	16	6	16		17	7	17	,	17		17	,	17		17	7	17	7	17	,
D1																																
D2																																
D4																																
D5																																
D9																																
D11																																
D14																																
D15																																
D16																																
D17																																
D21																																
D23																																
D25																																
D27																																
D29																																
D33																																
D36																																
D38																																

Figure 6.14 The presence of mature phase figs on female trees of *F. deltoidea* var. *angustifolia* in Banting oil palm plantation in each month between May 2016 and August 2017. (Grey bars = figs were present, white bars = figs were absent and black bars = when the trees were cut off).

6.4.3 Fig development

One hundred and twenty four randomly selected A phase male figs on 21 trees were marked and observed until they were no longer on the trees. Ninety one of these figs were entered by pollinators and developed normally, whereas the remaining figs eventually aborted. The A phase figs were entered as early as the fourth week after being marked. The fastest maturation from A phase until E phase was 17 weeks while the slowest was 36 weeks (**Error! Reference source not found.**). The average duration from A phase until the emergence of pollinators was 28.36 ± 4.58 weeks (Mean \pm SD). DE phase male figs mostly stayed on the trees for two weeks before falling. Figs that had not been entered began to show signs of the changes associated with future abortion after 16 weeks. Un-entered figs stayed on the trees for up to 50 weeks before being aborted and falling from the trees.

After the figs had been exposed for 16 weeks, all the figs had either been pollinated or where showing indications that they were going to abort (Figure 6.15). Some figs showed external signs of being pollinated rapidly, within four weeks (Figure 6.16) and after 36 weeks all the pollinated figs were starting to release pollinator offspring (Figure 6.17). The first figs releasing the next generation of pollinators were on week 18, suggesting that the generation time was very short (only about 18 weeks) (Figure 6.17). No figs remained receptive after week 16 (by that day they were either pollinated or starting to abort). No C phase figs remained after week 36, which was 18 weeks later, again suggesting that the generation time was less than five months. Some figs showed signs that they were going to abort by week 6 (Figure 6.18), but presumably they had already been waiting to be pollinated by one or more days when the bags were removed. The aborting figs remained longer on the trees than those that were pollinated (Figure 6.19).



Summary of the frequency of developmental phases of figs on male trees. 124 unpollinated A phase figs were marked in week zero. Grey bars = AB phase, white bars = C phase, black bars = DE phase, hashed bars = aborting figs and spotted bars = absent figs.



Figure 6.15. The frequency of AB phases of figs on male trees. 124 un-pollinated A phase figs were marked in week zero.



Figure 6.16. The frequency of C phases of figs on male trees. 124 un-pollinated A phase figs were marked in week zero.



Figure 6.17. The frequency of DE phases of figs on male trees. 124 un-pollinated A phase figs were marked in week zero.



Figure 6.18. The frequency of aborting phases of figs on male trees. 124 unpollinated A phase figs were marked in week zero.

The 185 randomly-selected A phase female figs on 18 trees showed signs of being pollinated within as soon as 6 weeks (**Error! Reference source not found.**). Thirty nine of these figs were entered by pollinators and developed normally, whereas the remaining figs eventually aborted. The complete cycle from A phase to E phase varied from 16 to 34 weeks with a mean of 26.21 ± 5.89 weeks. Un-pollinated female figs were only detectable visually after 17 weeks and could last on the trees for 46 weeks before falling.

After the figs had been exposed for 16 weeks, all the figs had either been pollinated or where showing indications that they were going to abort (Figure 6.15). Some figs showed external signs of being pollinated rapidly, within sixr weeks (Figure 6.16) and after 36 weeks all the pollinated figs were starting to release pollinator offspring (Figure 6.17). The first figs ripen were on week 14 and ready to be consumed (Figure 6.17). No figs remained receptive after week 16 (by that day they were either pollinated or starting to abort). No C phase figs remained after week 36, which was 18 weeks later, again suggesting that the full life cycle of female fig was less than five months. Some figs showed signs that they were going to abort by week 8 (Figure 6.18), but presumably they had already been waiting to be pollinated by one or more days when the bags were removed. The aborting figs remained longer on the trees than those that were pollinated (Figure 6.19).



Summary of the frequency of developmental phases of figs on female trees. 185 unpollinated A phase figs were marked in week zero. Grey bars = AB phase, white bars = C phase, black bars = DE phase, hashed bars = aborting figs and spotted bars = absent figs.



The frequency of AB phases of figs on female trees. 185 un-pollinated A phase figs were marked in week zero.

219



The frequency of C phases of figs on female trees. 185 un-pollinated A phase figs were marked in week zero.



The frequency of DE phases of figs on female trees. 185 un-pollinated A phase figs were marked in week zero



The frequency of aborting phases of figs on female trees. 185 un-pollinated A phase figs were marked in week zero

The average development times from pollination to wasp emergence or female fig ripening (the lengths of C phases) were 20 weeks in the male figs and 18 weeks in the female figs (**Error! Not a valid bookmark self-reference.**). The durations of earlier fig developmental phases on male and female trees were similar, but DE phase male figs fell from the trees more quickly than E phase female figs. (**Error! Not a valid bookmark self-reference.**). Despite this, the figs were present for slightly longer on the male trees (**Error! Not a valid bookmark self-reference.**).

Figs that were not entered by pollinators stayed on the plants for about six months (Table 6.5). Un-entered male and female figs took the same length of time before showing signs that they were aborting, and once they started to abort were retained by the trees for the same length of time, though the times that aborting figs stayed on the trees was very variable Table 6.5).

Table 6.4. The numbers of weeks (means with range in brackets) of different phases during development of marked figs. Pairwise comparisons between sexes used Mann-Whitney U-tests.

	Ν	AB	С	DE	Total
	(figs)				
Male	91	5.46 (2-10)	20.91 (12-28)	2.04 (2-4)	28.36 (17-36)

Female	39	5.51 (4-8)	18.64 (10-30)	2.20 (2-4)	26.21 (16-34)
M-W U		1700.0	2207.5	1631.5	1378.0
P value		0.69	0.08	0.05	0.04

Table 6.5. The numbers of weeks (means with range in brackets) that unpollinated figs remained as AB phase in appearance and stayed on the trees after being recorded as starting to abort. Pairwise comparisons between sexes used Mann-Whitney U-tests.

	N (figs)	AB	Aborting	Time to abortion
Male	33	10.67 (6-16)	16.73 (2-38)	27.39 (8-52)
Female	146	10.69 (6-16)	13.56 (2-34)	24.25 (8-48)
M-W U		2379.5	2026.0	2086.5
P value		0.91	0.15	0.23

6.4.4 Pollinator limitation

Across the 15 month period of study there was considerable variation in the maximum numbers of figs recorded on different trees, but the range of maximum crop sizes on male and female trees was similar (Figure 6.15).

Aborting figs were present on most of the trees most of the time (Figure 6.8 and Figure 6.12), but the chance of pollinator entry was lower for figs on female trees. All the male trees had at least some of their figs pollinated, but the continuous absence of DE figs showed that female trees D25 and D27 did not have any pollinated figs throughout the whole sampling period (Figure 6.14). Few male trees had more un-entered than entered figs on any of the sampling dates, whereas un-pollinated figs routinely outnumbered pollinated figs on female trees (Table 6.6).

The proportion of figs on each tree that had failed to be entered and were starting to abort was calculated as the mean percentage of aborting figs divided by percentage the figs that had developed to the stage where entry by pollinators was possible (C phase + DE phased + aborting figs). The proportion of un-entered figs on male trees ranged from 0.05 to 0.57 (Figure 6.16). D28 had the highest rate of non-entry among

the male trees. This tree produced few figs (the mean crop size was less than 50) and only had DE (entered) figs once throughout the 15 months of observations. A high proportion of un-entered figs were present on every female tree throughout the sampling period (Figure 6.17), ranging from half to all of the figs. Female tree D25 never showed any sign of pollination even though it produced large crops of more than 2000 figs. There was a significant difference in the proportion of un-entered figs on male and female trees (ANOVA, F = 141.20, df = 1, 37, P < 0.001).



Figure 6.15 The maximum crop sizes on 21male and 18 female trees during 15 months of observations. Solid bars = male trees and open bars = female trees.

Sampling date		Male trees	3	F	emale tre	es
	P>NP	NP>P	NP=P	P>NP	NP>P	NP=P
25.05.2016	19	2	0	2	15	1
02.06.2016	19	2	0	3	13	2
13.06.2016	16	4	1	4	11	3
29.06.2016	16	4	1	4	11	3
14.07.2016	16	4	1	6	9	3
26.07.2016	18	2	1	10	6	2
09.08.2016	20	1	0	8	6	4
23.08.2016	17	3	1	7	9	2
06.09.2016	18	2	1	5	11	2
19.09.2016	19	2	0	2	13	3
04.10.2016	17	3	1	2	12	4
19.10.2016	17	3	1	2	12	4
07.11.2016	19	1	1	4	13	1
16.11.2016	18	3	0	3	13	2
28.11.2016	20	1	0	3	14	1
14.12.2016	18	3	0	2	15	1
28.12.2016	17	4	0	0	17	1
11.01.2017	19	1	1	0	18	0
25.01.2017	20	1	0	0	18	0
07.02.2017	20	1	0	0	18	0
21.02.2017	20	1	0	0	18	0
07.03.2017	20	0	1	0	18	0
23.03.2017	17	3	1	0	16	2
05.04.2017	17	3	1	0	14	4
20.04.2017	19	1	1	1	15	2
02.05.2017	17	2	1	2	13	3
15.05.2017	20	0	0	4	13	1
02.06.2017	19	1	0	1	16	1
13.06.2017	19	1	0	0	18	0
21.06.2017	19	1	0	0	18	0
03.07.2017	19	1	0	3	15	0

Table 6.6. The relative numbers of pollinated and non-pollinated figs on male and female fig trees (P = pollinated, NP = non-pollinated).

26.07.2017	13	0	0	4	12	0
09.08.2017	13	0	0	3	12	1
21.08.2017	13	0	0	3	13	0



Figure 6.16 The proportion of figs on 21 male trees that were aborting during the 15 months of observation (young AB figs were excluded when calculating proportions).



Figure 6.17 The proportion of figs on 18 female trees that were aborting during the 15 months of observation (young AB figs were excluded when calculating proportions).

Evidence given in Chapter 5 indicates that the pollinators of var. *angustifolia* are reluctant to disperse far from their natal figs and that dispersal distances away from natal trees are limited. This suggests that female plants that are further from male trees are likely to suffer greater pollinator limitation and that entry rates on male trees will be independent of the number of other male trees around them.

All the male var. *angustifolia* trees in the Banting plantation had at least some of their figs entered by pollinators. There was a significant positive correlation between the estimated numbers of figs releasing fig wasps on a given tree over the 15 months sampling period and the percentage of its figs that were entered by pollinators over that period (Spearman correlation, $r_s = 0.45$, df = 20, P < 0.05). Almost all the female trees had mature male plants growing within 10 metres (measured from the base of the trunks of their host oil palms) (Figure 6.18). There was a regular increase in numbers of mature male trees only within increasing distances from the focal female trees (Figure 6.21).

A mean of about 16% pollination was recorded if there was a male tree within 5 m (11 female trees) and 84% if the female trees were more isolated (7 female trees). Having a mature male plant within 5 m did not increase pollination rates (Figure 6.22). There was only one female tree with the nearest mature male than 30 m away from it. The percentage pollination of this tree was 12%. However, any relationship with distance could be obscured by the size of a female tree. The very small female tree D25 was never pollinated, yet the nearest male tree was less than 5m away. Male trees were also releasing highly variable numbers of pollinators. The likelihood of a female fig being pollinated was not related to distance to the nearest mature male tree (Pearson correlation, r = -0.11, df = 17, P > 0.05). However, pollination rates did increase in relation to the number of mature male trees within 50 m of a focal female tree (Figure 6.23) (Pearson correlation, r = 0.52, df = 17, P < 0.05).



Figure 6.18 The relationship between the percentage of entered figs in 18 female trees and the distance to the nearest adjacent mature male tree.



Figure 6.21 The cumulative frequencies of mature male trees only within varying distances from 18 focal female trees.



Figure 6.22 The relationship of pollination rates with the distance of the nearest mature male trees from focal female trees. (n = 18 female trees).



Figure 6.23 The relationship between pollination rates on 18 female trees and the number of mature male trees within a distance of 30 m.
6.5 Discussion

The phenology of 39 *F. deltoidea* trees growing as epiphytes in an oil palm plantation was observed over a period of 15 months to record within-tree and within-population patterns of leafing and fruiting. The phenology of this species is typical of a species living in an aseasonal environment (Harrison, 2000; Corlett, 1993), with production of new leaves and figs occuring throughout the year. This is in contrast to the leaf and fruiting phenologies of fig trees growing in more seasonal environments, and of most monoecious species (Krishnan et al., 2014). Lin et al. (2015) showed that tropical climatic factors (temperature and rainfall) did not shape the phenology of *Ficus benguetensis* Merr. in Northern Taiwan even though male and female trees showed different phenologies. A similar limited impact of climatic factors is seen with the phenology of the *F. deltoidea* in the oil palm plantation.

6.5.1 Leaf phenology

Most of the mature individuals had some elongate juvenile leaves as well as the smaller more rounded leaves that are seen as typical of older plants. New leaves and senescing leaves were present throughout the year. Suleman et al., (2011) found rather similar leafing phenology in the dioecious *Ficus montana* Burm. f. when the plants were kept under a controlled laboratory environment. However, in some years special events such as drought might stimulate more synchronised leaf senescence in *F. deltoidea*, as in *Ficus fulva* Reinw. ex Blume in Lambir Hills, Sarawak (Harrison et al., 2000). Stress caused by accidental spraying had this effect in the plantation and is an example of how control activities are likely to reduce biodiversity (Suzanti et al., 2016).

6.5.2 Fig phenology

In nursery pollination mutualisms, synchronisation between the development times of fruit and the pollinators is critical to ensure the survival of both parties (Sakai, 2002). At the population level the mutualism between dioecious fig trees and fig wasps is maintained by compatible fruiting phenology of the two plant sexes to allow seed production at the same time as maintaining pollinator populations. If there is a particularly favourable season for fig seed germination and establishment then female trees can concentrate seed production at this appropriate time.

The availability of receptive male figs at the times when pollinators emerge is critical to ensure the survival of fig wasp populations (Zhao et al., 2014). Estimating the size of var. *angustifolia* plants is difficult because of the shrubby growth form that can be compact or loose. The measure of tree size we used showed that larger female plants had more figs, but no relationship was confirmed in male plants. In both sexes the plants that had established higher up the trunks of the oil palms produced more figs. This may be because they experience higher light levels or other environmental conditions are more favourable further from the ground. Crop sizes of male and female trees did not differ significantly. A study of a dioecious fig tree under controlled conditions recorded the same crop size results (Suleman et al., 2011), whereas another dioecious species in Sarawak had males that produced more figs than the female trees (Harrison et al., 2000).

An asynchronous fruiting phenology usually occurs in *Ficus* species that occupy wet and aseasonal areas where the availability of water and ability for photosynthesis are consistent throughout the year (Patel, 1996). Asynchronous fruiting patterns, at the population and single tree levels, allow the production of the fig wasps and seeds all year and allow fig wasp populations to be maintained without the need for females to disperse long distances to find figs at a suitable stage for entry and oviposition (Patel et al., 1993; Patel, 1996; Hossaert-McKey and Bronstein, 2001). Having small numbers of mature female figs present all year-round can benefit local resident populations of frugivores (Patel, 1996) and may benefit the plants because their seed is being dispersed continuously (Mackay et al., 2018). Because only a small number of figs are mature at any one time it means that birds such as hornbills which disperse widely in search of large crops will not be attracted and this is likely to mean that long distance seed dispersal is rare. A disadvantage of crop asynchrony is that it provides a larger window for non-pollinator fig wasps to oviposit and they should become more abundant (Harrison and Yamamura, 2003). However, F. deltoidea in Peninsular Malaysia supports no known species of non-pollinators so this is not a problem for the plant and its pollinator.

Among dioecious *Ficus* species where receptive male and female figs are receptive at the same time the sexes typically emit similar attractant volatiles to ensure the visitation to female trees and ensure seed development (Hossaert-McKey et al., 2016). This is the case in *F. deltoidea*, where receptive figs are present on male and female trees almost continuously. When male and female figs are receptive at different times

the selection for mutual mimicry is relaxed and the volatiles released by receptive male and female figs can be different (Chen et al., 2009). This suggests that the receptive figs of male and female *F. deltoidea* are likely to emit similar volatiles, but experimental results suggest this is not the case (Chapter 5).

In monoecious fig trees the development of seeds and pollinator offspring is in the same figs and their development times are similar. In most dioecious fig trees, the development of figs is faster on male trees (Harrison et al., 2000; Patel and McKey, 1998; Suleman et al., 2011). Male fig trees also tend to have figs with more female flowers and produce figs more frequently than female fig trees (Mizuki et al., 2005). This may be because female fig trees need to allocate more resources to produce seeds than male trees provide to support pollinator offspring (Leigh and Nicotra, 2003). Short development times benefit the fig wasps as there is less time to be attacked by parasitoid fig wasps (Harrison and Yamamura, 2003), but this is not a factor in *F. deltoidea*. Assessing the exact timing of fig receptivity is really difficult without using fig wasp choice experiments (Harrison et al., 2000) but fig development after pollinator entry appears to be about the same in female and male var. *angustifolia*.

The period of receptivity varies among different plant species and reflects a balance between the benefits of a greater possibility pollinator visitation and the greater maintenance cost of the flowers (Hossaert-McKey et al., 1994; Rathcke, 2003). Non-pollinated figs that remain receptive for longer give extra time for the pollinators to enter them, but older figs may produce less seed and fig wasp offspring (Liu et al., 2013). A study of *Ficus montana* showed that in this species the ostiole can also remain open for at least three days after it is first entered (Suleman et al., 2011). Like *F. montana*, pollinator foundresses of *F. deltoidea* can each enter several figs to lay their eggs and pollinate (Chapter 7), so an extended period of receptivity after entry of the first pollinator may also be present in this species.

6.5.3 Pollination limitation

Many figs of *F. deltoidea* failed to be entered by figs wasps and were eventually aborted. Figs on female plants were particularly likely to fail to be pollinated, despite each foundress being capable of pollinating up to six figs (Chapter 7). This routine failure of pollination in female figs might be due to the fig wasps preferring to enter male figs (Chapter) or might occur due to the poor dispersal ability of the fig wasps

(Chapter 5). Successful pollen flow in flowering plants involves (i) both maternal and paternal plants producing receptive female flowers and source of pollen at the same time, (ii) a distance component between pollen donor and receptive flower locations; and (iii) a pollinator that is able to disperse the pollen across the distances required (Jordano, 2017).

In this study, all three aspects appear to be optimum but still the number of pollinated figs was very low. Insect mediated pollination depends on the insect's behaviour and the spacing of the plants (García and Chacoff, 2007). The nature of fig wasps, which are small, short lived, delicate and weak flyers might be the reason spatial limitation might lead to pollinator limitation of isolated fig trees, but the pollinators of some fig trees can travel over very long distances between trees (Compton, 2002; Ahmed et al., 2009). Although the pollinators of some monoecious *Ficus* are able to fly over hundreds of kilometres (Harrison and Rasplus, 2006; Ahmed et al., 2009), more restricted dispersion seems to be typical of pollinators of dioecious species, including var. *angustifolia*. The observed reduction in pollination rates as local densities of male trees declined in the oil palm plantation are in agreement with the results of the experiments described in Chapter 5 and suggest that the *Blastophaga* sp. has very limited dispersal.

Due to continuous production of figs by both sexes of *F. deltoidea*, the temporal overlap within trees was high with receptive figs and DE phase figs often present at the same time. This permits entry of the pollinators into figs on their natal trees (Patel, 1996) and reduces the energetic costs and dangers involved when they need to fly to figs on other trees (Chapter 7). In particular, entering figs on natal male trees ensures that the figs are suitable for oviposition, whereas dispersal to other trees can lead to entry into female figs, where they cannot reproduce, and cannot fly to other trees because their wings are lost during entry through the ostiole. Selection to avoid female trees is less prominent in seasonal environments where pollinators may be released at times when very few receptive male figs are available (Patel, 1996). In var. *angustifolia* the production of figs was all year-round, generating strong selection for pollinators to differentiate between male and female figs (Patel, 1996; Hossaert-McKey et al., 2016; Chapter 5).

6.6 Conclusions

In response to the stability of its Malaysian climate *F. deltoidea* produces new leaves and figs throughout the year. Plants with some elongate juvenile leaves can produce figs. The asynchrony of fig production within and between trees permits continuous production of fig wasps and seeds by male and female plants, which should allow fig wasp populations to cycle on the same plant. Despite the high density of plants within the oil palm plantation, many female figs failed to be pollinated and this suggests that its pollinator usually disperses very short distances from its natal figs.

Chapter 7

The pollination biology of *Ficus deltoidea* var. *angustifolia*, an epiphytic fig tree with very few flowers in its seed-producing figs

7.1 Introduction

Inflorescence design influences the relationship between plants and their pollinators and in particular the behaviour of the pollinators. Selection acts on inflorescence traits (and floral traits) to increase reproductive outcomes (Iwata et al., 2012). The number of flowers, inflorescence architecture, the number of flowers opening at one time (also known as display size) and the separation of sexual roles in time and space within an inflorescence are among the floral traits that influence pollinator behaviour (Harder et al., 2004). These floral traits evolve to maximize pollinator visitation rates, pollen removal and pollen transfer. Inflorescence design is also influenced by post-pollination factors such as diaspore size and its relationship to likelihood of dispersal.

7.1.1 Fig tree pollination

Fig trees (*Ficus*) are unique in having an urn-shaped inflorescence with its inner surface lined by tiny male and uni-ovulate female flowers. The inflorescences are called figs (syconia). As well as developing into seeds, these flowers provide development sites for the plants' fig wasp pollinators (Suleman et al., 2011). All *Ficus* species are pollinated by either a single or several species of fig wasps (Family Agaonidae, Hymenoptera) that are very rarely associated with other *Ficus* species (Cornille et al., 2012). Although most fig wasps are host specific, hybrids can occur if compatible fig trees are pollinated by fig wasps that developed in the figs of other species (Ghana et al., 2017). The specificity of the relationship is maintained by physical structure of the figs (van Noort and Compton, 1996) and differences in the volatile cues emitted from figs (Hossaert-McKey et al., 2016).

The access of pollinator foundresses to the stigmas of the flowers is controlled by a narrow ostiole, through which pollinators must crawl if seeds are to be produced. In *Ficus*, floral traits such as the number of male and female flowers in each inflorescence are easily altered to influence and respond to pollinator behaviour (Cook et al., 2004). Figs vary greatly in the numbers of flowers they contain, but inflorescence design in *Ficus* is relatively conservative and has remained unchanged for tens of millions of years. Figs vary in size and location, but this variety appears to be related to the different vertebrates that are responsible for seed dispersal, rather than differences in pollination biology (Compton et al., 2010).

One of the more variable aspects of inflorescence design in *Ficus* species is the relative numbers of male and female flowers that are present. Pollination of some species is active, with specific behaviours and pollen-carrying structures present in the fig wasps. This type of pollination is usually associated with a small number of male flowers in the figs (Kjellberg et al., 2001). Pollination can also be passive, where no specific pollen deposition or collection behaviour occurs. Figs with passive pollinators have higher ratios of anthers to female flowers and produce more pollen per fig (Kjellberg et al., 2001). To facilitate passive collection by their pollinators, the mature anthers in these figs burst open spontaneously (Jousselin et al., 2003).

A floral trait that controls pollinator behaviour in *Ficus* involves segregation of the sexual roles in their inflorescences. There are two types of breeding system associated with *Ficus* species. The figs of Monoecious species each have both male reproductive function (producing pollen and the fig wasp's offspring to transport the pollen) and female function (produce seed) in a single fig while in dioecious species, the male and female reproductive functions are separated on different trees (Nefdt and Compton, 1996). In dioecious figs, the figs on male trees support only pollinator offspring, while in female figs only seeds are produced (Raja et al., 2008a). Oviposition is prevented in female figs by having flowers with longer styles than in male figs with the surface of the stigmas difficult to penetrate by pollinator ovipositors (Tarachai et al., 2008). This floral trait ensures that only pollination occurs in the female figs. Pollination also occurs in the male figs, but they are not capable of developing seeds.

7.1.2 Re-emergence of foundresses

Foundress fig wasp females enter the fig cavity through the ostiole on their way to lay their eggs in the female flowers. From one to tens of foundresses may enter a single fig. When passing through the ostiole they lose their wings and parts of their antennae (Kjellberg et al., 2005b). It used to be thought that a foundress will always be trapped in the first fig they enter and die inside. Some re-emergence of fig wasps from figs of *Ficus carica* L. was nonetheless suggested by Grandi (1920). He found a lot of dissected figs had foundresses with their heads pointing out of the ostioles and he also found developing figs with no foundresses inside. Later discoveries have showed that females of some species of fig wasps regularly re-emerge and successfully enter and oviposit into subsequent figs (Gibernau et al., 1996; Suleman et al., 2013b).

Fig wasps that have entered a monoecious fig or a male dioecious fig can then walk to another fig and lay more eggs. However, if the fig wasp had first entered a female fig, re-emergence will still mean they are unable to reproduce as they can only walk to other female figs on the same tree. They still continue to pollinate flowers inside female figs, so re-emergence can benefit the host plant by allowing several figs to be pollinated by a single foundress fig wasp (Tarachai et al., 2008). Because such females can never reproduce, natural selection can only shape the behaviour of females that have entered male figs, so re-emergence from male figs needs to be of benefit to foundresses if the same behaviour will be selected for among females that entered female figs (Raja et al., 2008a).

Rates of re-emergence vary between species (Gibernau et al., 1996) and may be more common in pollinators of dioecious than monoecious fig species (Moore et al., 2003). The factors selecting for re-emergence behaviour will vary between species and reflect their population dynamics and environmental factors (Gibernau et al., 1996). Re-emergence is said to happen due to the limited number of oviposition sites in a single fig (Moore et al., 2003) and that females that entering more than one fig can produce more offspring (Gibernau et al., 1996). Only one offspring can develop in a single flower (Jousselin et al., 2001) and if the combined eggs loads of the foundresses that enter a fig are more that the total number of flowers the oviposition site limitation is inevitable. This means that if the egg load of a single female pollinator is larger than the number of female flowers present then re-emergence would always be favoured.

7.1.3 Pollinator sex ratio variation

Organisms typically produce equal numbers of male and female offspring (Fisher, 1958). However, Hamilton (1967) stated that in some situations where brothers and sisters mate with each other a female-biased sex ratio is optimal because this results in a greater number of grandchildren than if equal numbers of sons and daughters were produced. Hamilton (1967) proposed that organisms in isolated sub-populations (in a colony with limited dispersal where sib-mating is common) could adjust their offspring to a female-biased sex ratio because of this inbreeding and Local Mate Competition (LMC). In *Ficus*, where LMC conditions are typical inside the figs, a female-biased offspring sex ratio is expected to be produced and has frequently been observed (Martel et al., 2016). As foundress numbers entering a single fig increases a less female-biased offspring sex ratio becomes optimal and is often observed (Peng et al., 2014). In some species, at least, this results from an oviposition strategy where mostly male eggs are laid first, and where total clutch sizes decline with the increased competition for oviposition sites as more foundresses enter a fig (Raja et al. 2008).

The sex ratio of their offspring also often varies according to the number of foundresses that entered a fig. Hymenopterans are able to control the sex of their offspring, because male offspring result from a lack of fertilization and are haploid, whereas fertilized eggs produce diploid females. The ability to control the sex ratio by a foundress allows natural selection to influence female behaviour to produce sex ratios adjusted to local conditions (Kinoshita et al., 2002). Moore et al. (2005) and Raja et al. (2008b) showed that clutch size variation allows foundresses of one species to adjust their sex ratios because they lay most of their male eggs at the start of an oviposition sequence. The more eggs that are laid, the more female-biased the sex ratio becomes and more eggs are laid in figs where fewer foundresses are present because of competition for oviposition sites. Foundresses of the parasitoid wasp Nasonia vitripennis assess potential LMC by the presence of other females (Shuker and West, 2004) and numbers of foundresses can also directly influence offspring sex ratio in some fig wasps (Greeff and Newman, 2011). Because only female fig wasps carry pollen the sex ratio influences the proportion of fig wasp offspring that are valuable to their natal plant.

7.1.4 Ficus deltoidea reproduction

Ficus deltoidea is exceptional among Ficus species in that the figs produced by female plants contain only small numbers of flowers, sometimes just a single flower (Corner,

1969). Each pollinator foundress carries multiple pollen grains, and given that seed production in *Ficus* is often pollinator limited, a situation where each foundress can only generate a very limited number of seeds is likely to be sub-optimal and inefficient use of the foundresses. Given the apparent selective advantage for *F. deltoidea* of having figs that contain small numbers of large seeds (Corner, 1969), a feature that may be associated with the unusual, true epiphytic biology of many varieties of *F. deltoidea*, having female figs that allow foundresses to re-emerge from receptive figs and go on to pollinate further figs, would clearly be advantageous.

Foundresses that re-emerge have lost their wings (Suleman et al., 2013b). This prevents them from flying off in search of figs on male trees, but also means that they must walk between figs. This clearly limits their mobility and further requires that the plant produces figs that grow close together, and are at similar developmental stages. To be effective, this suite of characters necessarily requires foundresses to be willing to re-emerge from the first fig they enter, despite the risks that they will be making themselves more vulnerable to predation.

Every foundress that enters a female fig has no chance of reproducing. Because selection can only operate via male figs, where fig wasps do reproduce (Raja et al., 2008b), plants of this sex need to offer conditions that provide a selective advantage for those foundresses that enter more than one fig. Specifically, they need to produce more offspring than foundresses that opt to lay all their eggs in a single fig. This vicariant selection (Grafen and Godfray, 1991) might take the form of male figs having only small numbers of flowers, in combination with an ostiole structure that facilitates easy emergence from the figs.

7.2 Objectives

The questions addressed by the experiments described in this chapter were 1) How many foundress fig wasps enter a single fig of *F. deltoidea* var. *angustifolia*? 2) Does the number of foundresses vary between male and female figs? 3) How often do foundresses re-emerge from figs and does this differ between male and female figs? 4) How many figs can a single foundress enter? 5) How many eggs do foundresses lay in their first and subsequent figs? What is the relationship between egg loads and the number of female flowers in male figs? 6) How many flowers do they pollinate in

female figs? And 6) Do their offspring sex ratios vary between the first and subsequent figs that they enter?

7.3 Methods

7.3.1 Study varieties

The experiments were made using epiphytic *Ficus deltoidea* var. *angustifolia* in oil palm plantations and its pollinator *Blastophaga* sp. This variety has asynchronous leafing and fig phenology patterns on individual plants of both sexes and figs are produced more or less continuously throughout the year (Chapter 6). This results in typically several receptive figs being present on each tree at any one time. The anatomy of the figs is described in Chapter 4. Each male fig contains about 143 female flowers and each female fig contains only 4 female flowers. The figs of this variety have a long pedicel.

The only known pollinator for *Ficus deltoidea* was *Blastophaga quadrupes* which was probably reared from var. *lutescens* in 1993 collected in Java and Sumatra (Wiebes, 1993). Whether this pollinator species pollinates some other varieties of *F. deltoidea* is unknown, but it is likely that pollinator of var. *angustifolia* is restricted to this one variety (Chapter 5). The anatomy of the wasps and the large numbers of male flowers in male figs suggest that pollination by this species is passive (Chapter 4).

7.3.2 Study sites and experimental methods

The study was located in Banting oil palm plantation. Details of this site were provided in Chapter 2. To get an indication of the number of female fig wasps that can enter one particular fig, and how many re-emerge 200 male and 200 female figs were taken randomly from 21 male and 10 female trees during their early C phase (Galil and Eisikowitch, 1968). How the developmental phase of the figs was determined can be found in Chapter 2. Records of wings and bodies inside the figs provide only a minimum estimate of how many foundresses entered each fig because foundresses that lacked wings on entry could not be accounted for. The presence of wings indicates that the fig was the first one to be entered by one or more females. During early C phase, the wings can be seen either at the entrance to the ostiole or in the inner ostiole using a stereomicroscope.

The presence of a female fig wasp body also can be seen clearly in this phase as they have not decayed. If a developing fig did not contain any wings or foundress inside, then it indicates that a foundress that had previously entered another fig was responsible for pollination or galling the ovules. When a fig wasp body was present, the counts recorded their head position. If the body were found in the fig cavity, or pointing outwards through the ostiole, it shows they were potentially able to lay some eggs or pollinate the ovules. However, if the head were positioned in the ostiole facing towards the cavity, it means they died before reaching the female flowers.

To determine how many figs one foundress can enter, groups of 4 to 15 adjacent A phase figs were selected. These pre-receptive figs were confined in a fine mesh bags to prevent any entry of pollinators. One to two weeks later, mature male figs were collected just before fig wasp release and placed in mesh-covered containers to let the foundresses emerge naturally. After the foundresses emerged, they were brought to the plantation and a single foundress were placed in the bags using a fine brush (Figure 7.1). After the foundress was placed in the bag, it was closed again to prevent any entry from other pollinators. A total of 30 bags on 6 male trees and 38 bags on 8 female trees were included in this experiment.

Six week later, the bags were opened to record how many figs were in C phase, and count the numbers of fig wasp offspring and the numbers of seeds. The first fig entered by each foundress was found by looking for the presence of the wings under a stereoscopic microscope. The number of galls were counted in the male figs and the number of seeds were counted in the female figs. These counts allowed the size of clutches laid by the single foundress in their first and subsequent figs to be compared. However, the precise sequence of entry by the wingless foundress into the subsequent figs could not be determined.

7.3.3 Statistical methods

The number of foundresses remaining inside the figs, the differences in the head position of the foundress inside the figs and the frequency of figs entered by a single foundress were analysed using Fisher's Exact Test as some of the expected outcomes

were less than 5. For differences in the number of winged and wingless foundress in a fig, Chi square test were used. Pearson correlations were examined between empty visited fig percentage and the mean number of dead foundresses within male and female trees.

The total brood size for foundresses that entered different numbers of figs, total brood size of first entered and subsequent male figs, differences in total brood size between experimental and wild figs, and total seed production in the first and subsequent female figs entered were analysed using generalized linear models (GLM) with Poisson error distribution. If overdispersion occurred, quasipoisson errors were used for those count data.

The offspring sex ratios in emerged and non-emerged foundresses were analysed using generalized linear models (GLM) with a gaussian error. The offspring sex ratios in first and subsequently entered figs and comparisons of sex ratios between wild and experimental figs were analysed using generalized linear models (GLM) with a quasibinomial error distribution and logit link. Spearman rank test correlation were used when examined the relationship between brood size and sex ratio in natural and experimental figs. The tests were all performed in R studio (1.0.153) and SPSS Statistics 20.

7.4 Results

7.4.1 The number of fig wasps entering a fig

The minimum number of fig wasps entering a fig in male and female trees can be estimated from the number of remaining fig wasp bodies in the figs. Based on these, the range of fig wasps entering the male trees was from 1 to 6 while for female trees the range was from 1 to 4 fig wasps (Figure 7.2). These values do not include pollinators trapped while attempting to enter the figs. The plant sexes showed rather similar patterns, with empty developing figs without any foundress bodies being the most common and the frequency of figs with more foundresses present declining (Figure 7.2). Fisher's Exact test nonetheless found a significant difference between male and female trees in the number of remaining foundresses inside the figs (df = 6, P < 0.05). Developing figs without any foundresses inside were frequently observed in both sexes. In male trees 81 out of 200 of the figs did not have the foundresses'

body inside the fig while in the female figs, the numbers without foundresses' bodies were 110 out of 200. The number of visited figs without a foundress body varied from 14.3% to 66.7% in male fig trees (Table 7.1) and from 25.0% to 100% in female trees (Table 7.2).



Figure 7.1 The process of inserting a single foundress in mesh bags containing receptive figs.



Figure 7.2. The number of foundresses remaining inside figs of var. *angustifolia* that had seeds or galls developing. Solid bars = male trees, open bars = female trees.

Table 7.1 The numbers of naturally-visited male figs with foundress bodies either in the fig cavity or with the body stuck head-outwards in the ostiole. Means do not include figs with no foundress corpses.

Tree	n figs	Figs with foundresses (%)	Figs without foundresses (%)	No. of foundress corpses per fig		
				Mean	SD	Range
1	21	80.95	19.05	1.76	1.30	1-6
2	11	45.45	54.55	1.73	1.10	1-4
3	5	80.00	20.00	1.40	0.89	1-3
4	6	83.33	16.67	1.67	0.82	1-3
5	5	40.00	60.00	1.00	0.00	1
6	5	60.00	40.00	1.20	0.45	1-2
7	5	40.00	60.00	1.00	0.00	1
8	17	64.71	35.29	1.29	0.69	1-3
9	5	20.00	80.00	1.00	0.00	1-3
10	6	66.67	33.33	1.00	0.00	1
11	7	57.14	42.86	1.00	0.00	1
12	13	69.23	30.77	1.38	0.65	1-3
13	7	71.43	28.57	2.00	0.82	1-3
14	5	40.00	60.00	1.00	0.00	1
15	5	40.00	60.00	1.00	0.00	1
16	9	33.33	66.67	1.11	0.33	1-2
17	21	57.14	42.86	1.24	0.44	1-2
18	10	50.00	50.00	1.60	1.07	1-4
19	9	66.67	33.33	1.22	0.44	1-4
20	7	85.71	14.29	1.57	0.77	1-3
21	21	76.19	23.81	1.10	0.44	1-3
Total	200	62.00	38.00	1.57	0.49	1-6

Table 7.2 The numbers of naturally-visited female figs with foundress bodies either in the fig cavity or with the body stuck head-outwards in the ostiole. Means do not include figs with no foundress corpses.

Tree	n figs	Figs with foundresses (%)	Figs without foundresses (%)	No. of foundress corpses per fig		
				Mean	SD	Range
1	53	50.94	49.06	1.04	0.19	1-3
2	8	75.00	25.00	1.63	0.92	1-2
3	13	69.23	30.77	1.00	0.00	1
4	45	44.44	55.56	1.13	0.40	1-3
5	33	60.61	39.39	1.33	0.69	1-3
6	13	30.77	69.23	1.23	0.44	1-2
7	10	30.00	70.00	1.50	0.97	1-4
8	12	66.67	33.33	1.33	0.49	1-2
9	8	37.50	62.50	1.00	0.00	1
10	5	0.00	100.00	0.00	0.00	0
Total	200	50.00	50.00	1.30	0.41	0-3

From a total of 324 foundress bodies found in the 400 dissected figs, 286 bodies were situated with heads towards the outside, 28 were present inside the fig cavity and only 10 had died while attempting entry through the ostiole (Figure 7.3). More foundress bodies were found facing the ostiole in the male trees while more foundress body were found in the cavity in the female figs. There was a significant difference in the location of the foundress in the figs between male and female trees using Fishers Exact Test (df = 2, P < 0.001).

The wings counts in developing figs also give an estimate of the number of figs that were entered by winged foundresses (and possibly also additional foundresses without wings) and those entered only by females that already had lost their wings. The likelihood of having wings at the ostiole was quite similar in male and female figs (Table 7.3) and did not differ significantly between figs on male and female trees (Chi square, $\chi^2 = 0.28$, df = 1, P > 0.05).

In male figs the percentage of wing-free visited figs on a tree and the mean number of dead foundresses in figs where foundresses were present was negatively correlated (Pearson correlation, r = -0.55, n = 18, df = 17, P < 0.001) (Figure 7.4). The relationship was also negatively correlated in female figs (Figure 7.5, but not significant (Pearson correlation, r = -0.37, n = 10, df = 9, P > 0.05). This suggests that crops where more of the figs were entered only by foundresses that had already entered other figs had fewer foundresses that remained in the figs. Variation in foundress densities on different crops may be an explanation.



Figure 7.3 The positions of foundresses in the figs. HTO = head towards ostiole, ITC = body in the cavity, HTC = Head towards cavity. Solid bars = male trees, open bars = female trees.

Sex	n	n	No. of	No. of	Percentage	Frequency	Frequency	Percentage
	(trees)	(figs)	figs	figs	with wings	of 1	of >1	of >1
			with	with		winged	winged	winged
			wings	wings		foundress	foundress	
			present	absent		(Range)	(Range)	
						120	16	
Male	21	200	136	64	62.35	(2-17)	(0-4)	6.24
						103	28	
Female	10	200	131	69	63.96	(3-43)	(0-19)	11.46

Table 7.3. Wings counts in figs from male and female trees.



Figure 7.4 The relationship between the percentage of developing male figs with no foundress present and the mean number of foundresses in the figs where one or more foundresses were present. Each data point indicates one crop of male figs.



Figure 7.5 The relationship between the percentage of developing female figs with no foundress present and the mean number of foundresses in the figs where one or more foundresses were present. Each data point indicates one crop of female figs.

7.4.2 Emergence rates from experimental figs

In the bags where a single foundress was introduced, 50 of the 68 foundresses (73.53%) entered more than one fig with a maximum of four male figs and six female figs entered (Figure 7.6). The likelihood of re-emergence from the first fig was not significantly higher from female figs (79.0%) compared to male trees (66.7%) (Chi square, $\chi^2 = 1.30$, df = 1, P > 0.05), and the difference in the number of figs entered was not significant (Chi square, $\chi^2 = 4.48$, df = 3, P > 0.05).

7.4.3 Brood size in re-emergence and non-emergence foundresses

There was a large increase in the total brood sizes achieved among foundresses that re-emerged and entered additional figs. The brood size of non re-emerging females ranged from 42 to 137 while in the re-emerging females the brood size ranged from 26-236 (Table 7.4). There was a significant difference in the brood size of emerging and non re-emerging foundresses (GLM, $\chi^2 = 211.84$, df = 1, P < 0.001). Females that only oviposited in one fig had a mean brood size of 80.7, whereas foundresses with offspring in three figs averaged more than twice this number of offspring (198.2) (Figure 7.7). The number of offspring did not increase for foundresses that entered four rather than three figs (Figure 7.7). There was a significant difference in the total brood size depending on the number of figs they entered (GLM, $\chi^2 = 348.35$, df = 3, P < 0.001).

7.4.4 Brood sizes in the first and subsequent figs entered

The increased offspring numbers among females that re-emerged may be a reflection of their vigour, rather than having the opportunity to access more oviposition sites. Brood sizes were higher in the first figs entered by foundresses compared to the subsequent figs they entered (Figure 7.8) (GLM, $\chi^2 = 243.27$, df = 1, P < 0.001). However, there was also a significant difference in the brood sizes in the first fig entered between re-emerging and non re-emerging foundresses (GLM, $\chi^2 = 0.22$, df = 1, P < 0.001) where non re-emerging foundress laid fewer eggs than re-emerging foundresses.A comparison of brood sizes between the experimental and randomly taken naturally-pollinated figs (Figure 7.9) suggested that they were similar (means of 60.53 and 59.94 respectively) (GLM, $\chi^2 = 0.16$, df = 1, P > 0.05).

Foundress	No. of figs entered	Re-emerged	Female offspring	Male offspring	Total brood size
1	1	No	77	9	86
2	1	No	80	14	94
3	1	No	110	27	137
4	1	No	40	6	46
5	1	No	57	22	79
6	1	No	37	5	42
7	2	Yes	101	60	161
8	2	Yes	88	41	129
9	2	Yes	106	22	128
10	2	Yes	77	38	115
11	2	Yes	20	6	26
12	3	Yes	184	33	217
13	3	Yes	113	64	177
14	3	Yes	184	52	236
15	3	Yes	140	23	163
16	4	Yes	182	26	208
17	4	Yes	99	63	162
18	4	Yes	147	68	215

 Table 7.4 The differences in re-emergence and non-emergence foundress



Figure 7.6 The number of figs entered by single foundresses. Solid bars = figs on male trees, open bars = figs on female trees.



Figure 7.7. Total brood sizes for single *Blastophaga* sp. foundresses that entered different numbers of figs on male trees.



Figure 7.8. The numbers of pollinator offspring in the first figs entered by foundresses and the subsequent figs they entered.



Figure 7.9 Pollinator brood sizes in the experimental figs and naturally-pollinated figs.

7.4.5 Pollination by foundresses that entered one or more female figs

In figs on female trees, pollinator foundresses also routinely re-emerged and pollinated flowers in up to six figs. In the first figs that the foundresses entered (where wings were present), all the flowers were successfully pollinated. This number decreased slightly in subsequent figs (Figure 7.10). However, each fig contained very few flowers to be pollinated and a GLM (with Poisson error), showed there was no significant difference in the number of flowers pollinated by the foundresses in the first and the subsequent figs ($\chi^2 = 0.38$, df = 1, P > 0.05). In figs of both sexes, those entered by only wingless foundresses were usually located very close to the first fig the pollinator had entered (Figure 7.11).



Figure 7.10. The number of seeds produced in the first figs entered and the subsequent figs.



Figure 7.11. The developing male figs pollinated by a single foundress.

7.4.6 Variation in offspring sex ratios in figs entered by a single foundress

The sex ratio of *Blastophaga* sp. offspring was usually strongly female biased (Table 7.4 and Table 7.5). The offspring sex ratio (proportion of male offspring) was higher in the foundresses that re-emerged than those that did not re-emerge (Figure 7.12) (GLM, $\chi^2 = 199.32$, df = 1, P < 0.001). This was strongly influenced by a small number of subsequent-entry figs where most of the offspring were male (Table 7.5).

The sex ratio also appeared to be higher in the subsequent figs entered by the foundresses than in first figs (Figure 7.13 but this reflected the small number of foundresses that had mainly male offspring in subsequent figs and there was no significant difference (GLM, $\chi^2 = 1.79$, df = 1, P > 0.05). There was also no significant relationship between offspring sex ratio and brood size (Spearman correlation, $r_s = 0.01$, df = 30, P > 0.05, Figure 7.14, Figure 7.15). From the twelve re-emerged foundresses, only four foundresses showed a significant difference between the offspring sexes in the first and subsequent figs and three of these had more male than female offspring in subsequent figs (Table 7.6).

A comparison of sex ratios between the experimental and naturally-pollinated figs showed that the experimental figs contained a slightly higher proportion of male offspring (0.25 and 0.18 males respectively, Figure 7.16), but the difference was again not significant (GLM, $\chi^2 = 3.70$, df = 1, P > 0.05). As in the experimental figs, offspring sex ratios in naturally-pollinated figs were not correlated with the number of offspring present (Spearman correlation, $r_s = 0.02$, df = 99, P > 0.05, Figure 7.17). Unfertilised foundresses produce only male offspring. Six of the broods in the naturally-pollinated figs consisted of more male offspring than female offspring (Figure 7.17), but only one fig contained only male offspring. This fig may have been entered by one un-mated foundress.

Table 7.4 The offspring sex ratio ratios (number of male offspring/total offspring) in the first fig entered by a *Blastophaga* sp. foundress.

Fig no.	Female	Male	Total offspring	Sex ratio
1	77	9	86	0.10
2	80	14	94	0.15
3	110	27	137	0.20
4	40	6	46	0.13
5	57	22	79	0.28
6	37	5	42	0.12
7	35	27	62	0.44
8	62	29	91	0.32
9	58	11	69	0.16
10	61	14	75	0.19
11	18	6	24	0.25
12	106	23	129	0.18
13	90	30	120	0.25
14	114	37	151	0.25
15	61	12	73	0.16
16	58	5	63	0.08
17	61	7	68	0.10
18	53	16	69	0.23
Total	1178	300	1478	0.20

Fig no.	Female	Male	Total offspring	Sex ratio
1	66	33	99	0.33
2	26	12	38	0.32
3	48	11	59	0.19
4	16	24	40	0.60
5	2	0	2	0.00
6	50	6	56	0.11
7	18	5	23	0.22
8	44	8	52	0.15
9	48	8	56	0.14
10	44	8	52	0.15
11	5	44	49	0.90
12	21	13	34	0.38
13	28	4	32	0.13
14	5	29	34	0.85
15	26	7	33	0.21
16	31	3	34	0.09
17	34	2	36	0.06
18	29	6	35	0.17
19	2	27	29	0.93
20	46	11	57	0.19
21	52	11	63	0.17
22	23	7	30	0.23
Total	664	279	943	0.30

Table 7.5 The offspring sex ratio (number of male offspring/total offspring) in the subsequent figs entered by a *Blastophaga* sp. foundress.



Figure 7.12 *Blastophaga* sp. offspring sex ratios (number of male offspring/total offspring) in relation to emerging and not emerging behaviour.



Figure 7.13 *Blastophaga* sp. offspring sex ratios (number of male offspring/total offspring) in relation to oviposition sequence.

Foundress	Fir	st	Subsequ	ent	χ^2	P value
no.	Female	Male	Female	Male		
1	35	27	66	33	0.44	0.51
2	62	29	26	12	0.001	1.09
3	58	11	48	11	0.16	0.69
4	61	14	16	24	20.14	< 0.001
5	18	6	2	0	0.65	0.42
6	106	23	78	10	1.70	0.19
7	90	30	23	34	20.1	<0.001
8	114	37	70	15	1.49	0.22
9	61	12	79	11	0.59	0.44
10	58	5	124	21	1.72	0.19
11	61	7	86	61	20.93	<0.001
12	53	16	46	47	12.47	<0.001
Total	777	217	664	279	15.28	<0.001

Table 7.6 Offspring sex ratios in first and subsequently entered figs



Figure 7.14 The relationship between the sex ratio and brood size in experimental figs entered by a single foundress. Each point indicates the contents of one fig. One foundress generated 1- 4 data points.



Figure 7.15 The relationship between the total offspring sex ratio and total brood size in experimental figs entered by a single foundress. Each point indicates the total brood size for one foundress, and includes broods spread across two or more figs.



Figure 7.16 Pollinator offspring sex ratios in experimental and naturally-pollinated figs.



Figure 7.17 The relationship between the brood size and sex ratio in naturallypollinated figs. Each point indicates the contents of one fig. Several foundresses could have contributed offspring to each fig.

7.5 Discussion

The results presented in this chapter have clarified several aspects of the pollination biology of var. *angustifolia* and the biology of its pollinator *Blastophaga* sp. The figs of this species are small, and its female figs have very few flowers. Despite the small size of the figs they can be entered by multiple pollinators, in part because foundress females routinely enter more than one fig. This behaviour is present on both male and female plants and although figs on female plants are more likely to fail to be entered by pollinators (Chapter 6) female figs entered by pollinators have almost all their flowers pollinated. Foundresses that lay eggs in more than one fig produce more offspring, so re-emergence from figs is likely to be to their advantage. The sex ratios of pollinator offspring are sometimes different in the first and subsequent figs they enter where the male ratio was higher in the subsequent figs.

7.5.1 Multiple foundresses

The number of foundresses that enter a fig varies greatly between species and might be expected to be higher in species with bigger figs, because they have more flowers to be pollinated and offer more sites for pollinator oviposition (Tayou, 1991; Patel et al., 1995; Chen et al., 2001; Yang et al., 2012). In var. *angustifolia* the number of flowers in female figs is very small and a single foundress would normally be sufficient to pollinate all of them, but large numbers of dead foundresses were recorded in some naturally-entered var. *angustifolia* figs.

7.5.2 Walking rates

According to Grafen and Godfray (1991) once a female fig wasp enters the fig cavity, they never leave and re-emergence is not possible. This view is now out-dated and many pollinators are known to be able to re-emerge from figs (Herre et al., 2008). The occurrence of figs showing the production of galls and seeds without any dead foundress inside confirms this (Pereira and Prado, 2006). Even though female wasps lose their wings in the first fig, they still able to leave the first fig and enter other figs by walking (Suleman et al., 2013a). A study of *Kradibia tentacularis* (Grandi) showed the fig wasps were able to re-emerge and oviposit in up to four receptive *Ficus montana* Burm. f. figs and that they tend to enter the first receptive fig they encounter (Suleman et al., 2013a). Most of the foundresses re-emerge from female and male figs and go to the other figs to lay further eggs (Tarachai et al., 2008). Not being able

to re-emerge would prevent these foundresses from laying all their eggs (Raja et al., 2014).

Plant traits control whether foundress re-emergence is possible or not. In dioecious fig trees, allowing the foundress to re-emerge from male figs allows higher production of the next fig wasp generation (Moore et al., 2003). However, allowing re-emergence may result in the foundresses producing more male progeny and these do not give direct benefits to the plants (Anstett et al., 1998). In *F. deltoidea*, re-emergence gives benefits to both sexes of the plant. In the male trees, higher numbers of offspring are produced and some of these may carry pollen to female trees, and in female trees, re-emergence causes higher seed production. In *F. montana*, re-emergence rates increased with foundress density and this led to a more equal distribution of eggs/pollen across the figs (Moore et al., 2003).

The rates of emergence in this study were not significantly higher in female figs trees than male fig trees. Moore (2001) also found that rates of emergence were not dependent on plant sex. When foundresses enter female figs, they cannot lay eggs due to the long styles of the female flowers. This does not seem to make them more likely to re-emerge. The limitation of oviposition sites is one reason why fig wasps may re-emerge (Raja et al., 2008a) A foundress also wants to avoid losing all its offspring if something happens to one fig (fig abortion, parasitism and pathogens) so spreading the risk by laying in several figs is an advantage (Gibernau et al. 1996; Suleman et al., 2013b).

7.5.3 Brood size comparisons

Blastophaga sp., foundresses that re-emerged had more offspring than those that remained inside the first fig they entered. The total egg loads of fig wasps that enter a fig might be greater or smaller than the number of female flowers available for oviposition in monoecious or male figs (Nefdt and Compton, 1996). In *F. montana*, having more eggs than the number of female flowers in one fig promotes regular reemergence (Ghana et al., 2012). The number of female flowers available for oviposition is highly variable in var. *angustifolia* male figs between 94 and 259 female flowers with a mean of 162.22 (Chapter 4) but most male figs of var. *angustifolia* have fewer female flowers than the egg loads of a single pollinator. Single foundresses were able to produce up to about 200 offspring if they were able to enter three or four figs

to lay their eggs. This provides a selective advantage to re-emerge from male figs, especially as the figs may also have eggs laid by other foundresses so fewer sites will be available for oviposition. Female trees benefit because foundresses have the same behaviour in female figs (Raja et al., 2008b). Weaker females may be un-able to re-emerge and these individuals laid fewer eggs in the first figs they entered (Raja et al., 2014).

7.5.4 Seed production

In var. *angustifolia* the very small number of female flowers in female figs guaranteed extensive pollination even if entered by a single fig wasp. All the first figs entered were 100% pollinated. However, the second and subsequent figs entered figs did show a slightly lower number of seed produced. A study of female *F. montana* figs, where many more flowers are present than in var. *angustifolia*, showed there were slightly more overall seeds produced when foundresses re-emerged rather than staying in a single fig, with the first and second figs having equal numbers of seeds (Moore et al., 2003).

7.5.5 Sex ratios

Because brood sizes in the first figs entered were larger than in subsequent figs, there was an expectation that brood sizes in the first figs would be more female biased than in subsequent figs (Kjellberg et al., 2005a), if this species employs the male eggs first oviposition strategy as documented in *Kradibia tentacularis* (Suleman et al., 2013b). This was found not to be the case as most females had similar offspring sex ratios in the first and subsequent figs they entered. Having a higher proportion of female offspring is better from the plant's perspective as more pollen can be dispersed. Some subsequently-entered figs had more male than female offspring. This was not present in any of the first figs that were entered. If these foundresses had shared a fig with another foundress that had laid mainly fertilised (female) eggs, then having many male offspring could be beneficial (Suleman et al., 2013b).
7.6 Conclusions

The foundress *Blastophaga* sp. routinely re-emerged from both sexes of the figs they entered. Re-emergence from male figs resulted in more overall offspring, though the first figs entered contained more offspring than subsequent figs. The effect of re-emergence from female figs was that more seeds were pollinated by each female. Re-emergence is favoured by the relatively small number of oviposition sites in male figs, which is often less than the egg loads of a single foundress.

Chapter 8

General discussion and Conclusions

8.1 Epiphytes in oil palm plantations

Oil palms housed a high number of epiphytes on their trunks. The epiphytes can partially mitigate the adverse effects caused by the conversion of natural habitat to oil palms, which generates a high degree of biodiversity loss (Foster et al. 2001). This is because epiphytes can provide habitats and food for other organisms such as insects (Suzanti et al., 2016). In Peninsular Malaysia, oil palms are often grown in areas that were once rubber plantations. The extent of epiphytes on both trees is strongly dependent on management intensity, but the architecture of the oil palm trunk provides very suitable host conditions for epiphytes (Sofiyanti, 2013).

Bryophytes, pteridophytes, lichens and angiosperms were all abundant epiphytes on the oil palms surveyed near Kuala Lumpur. Among the angiosperm epiphytes were *Ficus* species (Moraceae) that can exist as true epiphytes or hemiepiphytes. The epiphytic habit during juvenile stages is said to be an adaptation to avoid deep shade in the forest understory (Hao et al., 2013). Hemiepiphytic *Ficus* are strangler fig trees that later can give negative effects on the host plants by completion for light and even preventing growth by strangling. There are about 500 hemiepiphytic *Ficus* species that start life as epiphytes but become normal trees once their roots reach the ground (Hao et al., 2016). There are few truly epiphytic *Ficus* species that never have contact with the ground. Varieties of the evergreen shrub *F. deltoidea* can live as an epiphyte or as a tree that can reach up to seven meters tall (Zimisuhara et al., 2015). In this study, the three *F. deltoidea* complex epiphytes never had contact with the ground and can be said to be true epiphytes as described by (Corner, 1969). True epiphytes usually do not give any harm to the host plants (Zotz, 2013).

Ten *Ficus* species were identified in five different oil palm plantations in Peninsular Malaysia in this study. They included creepers, hemiepiphytes and epiphytes. Most of the species do not produce fruit until they reach considerable size, by which time they can kill the palm trees. Plantation managers are unlikely to ever let them grow sufficiently large. However, there are a few *Ficus* did produce fruit even when small. These figs (including figs of *F. deltoidea*) can provide food resources to frugivores that can disperse the seeds (Shanahan et al., 2001). As a result, the smaller *Ficus* species are more likely to contribute to biodiversity in oil palm plantations.

Ficus species in the different plantations were at different heights on the palm trunks, but different species did not occupy different heights on their hosts. The height of the trunks might be the factor that determines this. Different plantations have different ages and have different types of soils that can affect the height of the trunks. Older palm trees can cater for more epiphytes but management largely determines the abundance of the epiphytes and only some managers retain the epiphyte communities on the trunks without cutting them down. Many epiphytes including the fig trees appear to do no harm to palm oil production and management that encourages them should be encouraged (Prescott et al., 2015).

8.2 Ficus deltoidea

Ficus is a unique genus that has a closed inflorescence formed by an outer receptacle lined on the inside by tiny florets (a syconium). The closed inflorescence protects the floral structures from harsh environments and predators. Only specialised fig wasp pollinators (Agaonidae) can enter the figs. The pollinators make their way into the fig cavity through an ostiole and then gall, oviposit and pollinate the female flowers. *Ficus* and Agaonidae have been evolving together for a long time (van Noort, 2003). In this obligate mutualistic relationship the fig wasps provide pollen transfer and the fig wasps receive development sites for their offspring (Li et al., 2016). In dioecious fig trees such as *Ficus deltoidea* the nature of the relationship is modified. Female trees receive pollination services but they not only provide no reward, they prevent their pollinators from escaping to reproduce on other trees.

The group of fig trees comprising *Ficus deltoidea* and its relatives is one of the most interesting in the whole genus, displaying recent evolution, diversification and novel

growth forms in and around the Sunda Shelf. The group can therefore be a model for patterns of speciation and co-diversification with fig wasp pollinators, as well as providing biogeographical insights from this hyper-diverse region. The chemistry of the group has also received considerable attention in recent years, due to their role in traditional medicine and economic potential.

Corner (1969) considered that the *F. deltoidea* group had originated in Borneo and then spread relatively recently across Sundaland. This is the region extending from Peninsular Malaysia to Borneo and Bali where shallow seas have meant that the changes in sea levels experienced during the Pleistocene resulted in the total land area varying greatly. Islands formed and fused during ice ages and become isolated or disappeared beneath the sea during warmer inter-glacials, such as that being experienced today. As a result of these sea level changes, plants and animals in the region have been periodically restricted to refugia and then allowed to expand their distributions when larger areas of land (roughly twice that present today) became exposed during the ice ages (Lohman et al., 2011). The recent spread of *F. deltoidea* has allowed them to diversify but whether they have diversified sufficiently to be considered as separate species is uncertain. Currently *F. deltoidea* is regarded as one species, with different varieties, but some of these occur in sympatry, which suggests they are distinct biological species.

It is likely that all the varieties of *F. deltoidea* are passively pollinated by their associated *Blastophaga* fig wasps. The absence of pollen pockets and coxal combs in their pollinators indicate that they passively pollinate the flowers. Passively pollinated *Ficus* produce many male flowers because pollen transfer is inefficient as their specific pollinators have a very limited pollen load in their pocket (Kjellberg et al., 2001). The ratio of anthers to ovules was high in all the varieties of *F. deltoidea* investigated, but was higher in some than others. *F. deltoidea* also possesses two traits of passively pollinated species as described by (Kjellberg et al., 2001). They have anthers that dehisce and project into the fig cavity so that as the female pollinators emerge from their galls they can be covered with pollen during their way out (Figure 8.1).



Figure 8.1 Dehiscing anthers projecting into the fig cavity in an early E phase male fig of *F. deltoidea* var. *angustifolia*. (Red arrow points to a dehiscing anther).

Ficus species are described as typically having from 30 to 1600 seeds per fig (Lansky et al., 2008). However, the varieties of *F. deltoidea* investigated had as low as 3 seeds in var. *angustifolia* and even one seed in another variety. Apart from having small numbers of seeds, seed size in *F. deltoidea* is also extremely large compared to other *Ficus*. The small numbers of flowers in each inflorescence allows for each seed to be large without increasing the overall size of the figs that contain them. The larger seeds than is normal for fig trees is a presumed adaptation for an epiphytic life style, as it means that young seedlings have more resources available initially (Kraft et al., 2015b). Larger seed size also can prolong the respiration process during carbon limited times by supplying more nutrient reserves (Westoby et al., 2002).

Seed sizes are also correlated with leaf toughness, and are suggested to form a syndrome associated with slow development in resource-limited environments (Kraft et al., 2015b). Having larger seeds can increase the chance of survival under extreme environments (Li et al., 2016). This trait might produce larger seedlings to reach deeper in soil pockets to absorb more water or higher into the air to have a better photosynthetic process, and avoid carbon deficit by having more metabolic resources (Westoby et al., 1996). Slow growth is a featured trait of *F. deltoidea* described by Berg and Corner (2005). An epiphyte usually undergoes slow growth to tolerate water deficits in the upper environment above the ground (Hao et al., 2016). A study by (Kraft et al., 2015a) showed that well-defended leaves are usually associated with a plant that has larger seeds and slow growth. *F. deltoidea* has all the traits typical of epiphytes which are small leaves, slow growth and large seeds (Hao et al., 2016).

Seeds and seedlings of the hundreds of species of hemiepiphytic fig trees face similar conditions to those of epiphytic *F. deltoidea*, but unusually large seeds have not developed in these lineages (Hao et al., 2016). The large seeds of *F. deltoidea* may therefore have evolved under different selection pressures (and within the *F. deltoidea* complex, some varieties are not epiphytic, or only facultatively epiphytic). The large seed of *F. deltoidea* was said to be the ancestral condition and the small seeds of other *Ficus* a derived condition (Corner, 1969). However, this may be unlikely because monoecious fig trees have probably the original breeding system in the genus (Harrison and Yamamura, 2003). There is no evidence that pollinators were ever much larger in the past (Compton et al., 2010), so seed and pollinator gall sizes would need to be much different in shared figs. Figs containing large seeds and many pollinators would have had to be larger than those seen today.

Large red tepals are a feature of subsection *Frutescentiae*, which includes the *F. deltoidea* group and species belonging to Series Podosyceae (Berg and Corner, 2005). Varieties of *F. deltoidea* all possess large tepals, with four tepals surrounding each ovule (sometimes 3-5 tepals). The size and colour varies between sexes. The female figs of all varieties tended to have larger tepals than the male figs. The larger size in females might reflect that there are fewer flowers, so each tepal needs to be bigger, but the function of the tepals is unknown. Their prominent appearance and colour is hidden inside enclosed, dark figs, so they are unlikely to have a role in photosynthesis or be visible to pollinators.

The tepals in *F. deltoidea* figs were largest during receptivity, before shrinking post pollination. This suggests that they might have a role in pollinator attraction. Other possibilities include an anti-microbial function to deal with microbes brought in to the figs by the pollinators or to help position female pollinators above the flowers to aid pollen transfer on to the stigmas (Hong et al., 1998). *Ficus awkeotsang* Makino is called the jelly fig because of the large amounts of gel that is produced inside the figs. This species also has red tepals and biochemical studies have shown that they contain pectinesterases involved in gel formation (Komae and Misaki, 1993). The seeds of *F. deltoidea* var. *angustifolia* have a thin layer of gel around them that seems to inhibit mould formation (S.G. Compton, personal communication), which supports the idea of an antimicrobial function for the large tepals in this species.

In nursery pollination mutualisms and especially the fig wasp-fig tree mutualism, timing is everything. The emergence of short-lived adult fig wasps should be at the same time as the receptivity of other figs. In seasonal environments, some male dioecious fig trees release most pollinators at times of the year when the most female figs are receptive (Patel, 1996). In contrast, Malaysia is a country close to the equator with little seasonal variation in climate. *F. deltoidea* produced new figs and new leaves throughout the year and individual plants often had figs at all stages of development. This asynchronous development can avoid the local extinction of the fig wasp (van Noort, 2003) and can provide food resource to frugivores all the time (Shanahan et al., 2001).

Flowering phenology may have a strong effect on the extent of intersexual chemical mimicry because good quality mimicry is believed to be more important in species where receptive male and female figs are available at the same time and the fig wasps

can make a choice (Hossaert-McKey et al., 2016). However, this study showed that the fig wasps associated with var. *angustifolia* are more likely to enter figs on male plants. This suggests they were able to distinguish between the sexes of their host, and that inter-sexual mimicry is not very effective in this species.

Previously, it was believed that once a foundress fig wasp enters a fig she will always be trapped inside. More recent studies have showed many fig wasps are able to reemerge after pollination and oviposition (Moore et al., 2003) and the foundresses can enter other figs after walking between them (Suleman et al., 2013b). The tightness of the ostiole can be seen as being beyond the control of fig trees, with pollinator release an accidental consequence (Gibernau et al., 1996) or as an adaptation by the plants to allow foundresses to pollinate several figs rather than just one. There are very few flowers in female figs of F. deltoidea and it would not be efficient use of pollinators if they were trapped in the first fig they entered and could only pollinate one fig. This study showed that Blastophaga sp. routinely re-emerge from male and female figs of var. angustifolia after pollination and oviposition. The life span of adult female Blastophaga sp. was not measured in this study but based on personal observations, their active behaviour lasts only for a few hours. This limited time requires each foundress to choose a fig to enter quickly and means that only adjacent figs can be entered after they re-emerge. The figs of var. angustifolia are produced quite close together which makes it easier for foundresses to walk between them.

Suleman et al. (2013b) said that re-emergence is likely to be due to a lack of oviposition sites inside male figs together with the added benefit of spreading the risk of offspring deaths by having offspring located in more than one fig. Inflorescence design in var. *angustifolia* helps to stimulate foundress re-emergence because many of the male figs have fewer female flowers than the number of eggs in the female fig wasps. Females that re-emerged also produced more offspring. Similar results have been found in other pollinators that can re-emerge (Gibernau et al., 1996; Moore et al., 2003). The sex ratio is female biased in *F. deltoidea* and this event benefited the plants as more pollen can be dispersed. However, this ratio changed in the first and later figs that were entered. The size of the clutches might have affected the sex ratio of the pollinator offspring (Kjellberg et al., 2005a). Because male eggs in *Blastophaga* may be laid first, as in some other species (Raja et al., 2008b), a bigger clutch will produce a smaller proportion of male offspring.

8.3 Is *Ficus deltoidea* a complex of different biological species?

Species boundaries are maintained by limiting gene flow between taxa. In flowering plants pollinator behaviour can strongly influence gene flow, especially if the pollinators are host specific (Sedeek et al., 2014). Floral scents emitted by receptive flowers act as cues to influence pollinator behaviour (Souto-Vilaros et al., 2018) and in the nursery pollination mutualism between fig trees and fig wasps, long distance detection of suitable host figs is mediated by the scent of receptive figs (Hossaert-McKey et al., 2010). Apart from the scent cues, pre-pollination filters that reduce the chance of entry by non-adapted pollinators include non-volatile cuticular cues, ostiole size and shape, and floral arrangement within the figs (Borges, 2016). For the fig wasps to be able to reproduce the style lengths in male figs must also not be longer than the ovipositors of the foundress (Weiblen 2004; Kjellberg et al., 2005b). The foundress must also be able to gall the ovule to allow their offspring to develop (Ghana et al., 2015a). Divergence in volatile signals between figs could represent an initial isolating mechanism between individuals which is later reinforced by pollinator behaviour and morphological co-adaptation with their hosts (Soutos-Vilaros et al., 2018).

One of the outcomes of pollinator sharing is the production of hybrid plants and increasing numbers of *Ficus* hybrids are being recorded (Wang et al., 2016). Introgession between *F. auriculata* Lour. and *F. oligodon* Miq. in China is facilitated by similar volatiles emitted by their receptive figs (Wang et al., 2016). The attractant volatiles released by *F. deltoidea* varieties have not been compared. The varieties vary in the size of their figs and the numbers of flowers they contain, but the lengths of the styles in male figs are rather similar and it seems likely that pollinators from one variety could oviposit in the ovules of other varieties, if they enter the figs. Our behavioural experiments showed that the fig wasp (*Blastophaga* sp.) only enters their natal fig species (var. *angustifolia*) if that species is present. In the absence of the natal fig trees, small numbers of this pollinator will enter figs of var. *deltoidea* and can induce the gall formation, but if eggs had been laid in these figs, then the larvae that emerged failed to develop until maturity. Limited reciprocal experiments with the pollinators reared from var. *deltoidea* suggest that these were also unwilling to enter receptive figs of var. *angustifolia*.

These results can explain the field observations at Batu Pahat, where epiphytic var. *angustifolia* and var. *deltoidea* grow closely together and sometimes on the same host trees. However, no plants with intermediate features are present and gene flow between them seems to be absent. Two other different varieties (var. *kunstleri* and var. *trengganuensis*) were also planted in Batu Pahat outside of their natural ranges. No sign of pollination was observed, suggesting that these varieties are also avoided by the pollinators of var. *deltoidea* and var. *angustifolia*, but the unwillingness of the pollinators to travel far could also explain their absence. The two varieties var. *deltoidea* and var. *angustifolia* have different leaves and figs so other features as well as attractant volatiles may provide cues to their pollinators.

The two varieties remain distinct even when in sympatry and they have pollinators with behaviour that prevents gene flow. This suggests that they represent two different biological species. Some at least of the many other varieties of *F. deltoidea* probably are also likely to be distinct biological species. Molecular evidence supports this conclusion (S. Segar personal communication) and different species of pollinator fig wasps have been reared from different varieties of *F. deltoidea* in Brunei (F. Kjellberg, personal communication to S.G. Compton, 2016).

Most studies fail to find a sexual preference, but the experiments showed that *Blastophaga* sp. preferred to enter male figs. It also seems unwilling to leave its natal host unless the host has no receptive figs present. This clearly benefits individual pollinators because they are more likely to survive and reproduce, but is not so good for the plant.

8.4 Conclusions

To our knowledge, this is the first biological study of *Ficus* in oil palm plantations. In Peninsular Malaysia *F. deltoidea* was the only true epiphyte. There were three varieties of *F. deltoidea* recorded as epiphytes in this study (var. *angustifolia*, var. *deltoidea* and var. *trengganuensis*). There are morphological differences between the figs of these and other varieties. In Batu Pahat plantation, two of the varieties live sympatrically. The absence of hybrids between var. *angustifolia* and var. *deltoidea* in Batu Pahat plantation suggests that there is a barrier to gene flow between them. The limited dispersal of the pollinators away from natal trees may make gene flow less

likely and experiments showed that pollinator behaviour is normally preventing gene flow between these varieties. The fig trees are likely to produce volatiles that only attract their own specific fig wasps to pollinate them. The overall conclusion is that these two varieties of *F. deltoidea* are distinct species.

As a further conclusion, we might say that the specificity between fig trees and fig wasps in the many different *F. deltoidea* varieties is likely to be high. No evidence of gene flow or intermediate forms were found when two varieties were in sympatry and they remained distinct. The implication from this study is that all or most of the varieties might be different species. Previously all the varieties under *F. deltoidea* were defined based on morphological characters only. However, according to the biological species concept, they are said to be a same species if they interbreed with each other when living together. There are other examples of what were considered to be a single *Ficus* species being more than one. For example *Ficus insipida* Willd. in Mexico and *Ficus inspida* in Panama were described as the same species but they are now understood to be totally different species (Su et al., 2008). The major implication from this study is the taxonomists may have underestimated the number of *Ficus* species.

8.5 Future work

Future work should include study of the composition of the volatile organic compounds released by receptive figs of all varieties of *F. deltoidea* to find if they produce similar blends or not. Male and female plants need to be examined separately as the pollinators of var. *angustifolia* seem to prefer male figs, which suggests that their volatiles are not identical. Molecular population genetics can be used to see if gene flow is occurring between varieties and to establish a phylogeny. Detailed morphological and molecular examinations of the fig wasps should be conducted to determine whether they are different species or not. Previously, the classification of *F. deltoidea* varieties was based solely on plant morphology. These studies can give additional information to update the classification of this complex and explore their evolution.

References

- ABDULLA, M. A., AHMED, K. A. A., ABU-LUHOOM, F. M. & MUHANID, M. 2010. Role of *Ficus deltoidea* extract in the enhancement of wound healing in experimental rats. *Biomedical Research-India*, 21, 241-245.
- ADAM, Z., HAMID, M., ISMAIL, A. & KHAMIS, S. 2007. Effect of *Ficus deltoidea* Aqueous Extract on Blood Glucose Level in Nomal and Mild Diabetic Rats. *Malaysian Journal of Health Sciences*, 5.
- ADAM, Z., KHAMIS, S., ISMAIL, A. & HAMID, M. 2012. *Ficus deltoidea:* A Potential Alternative Medicine for Diabetes Mellitus. *Evidence-Based Complementary and Alternative Medicine*.
- AFFELD, K., SULLIVAN, J., WORNER, S. P. & DIDHAM, R. K. 2008. Can spatial variation in epiphyte diversity and community structure be predicted from sampling vascular epiphytes alone? *Journal of Biogeography*, 35, 2274-2288.
- AHMED, S., COMPTON, S. G., BUTLIN, R. K. & GILMARTIN, P. M. 2009. Wind-borne insects mediate directional pollen transfer between desert fig trees 160 kilometers apart. *Proceedings of the National Academy of Sciences of the United States of America*, 106, 20342-20347.
- ALDHEBIANI, A. Y. 2018. Species concept and speciation. Saudi Journal of Biological Sciences, 25, 437-440.
- ANSTETT, M. C., GIBERNAU, M. & HOSSAERT-MCKEY, M. 1998. Partial avoidance of female inflorescences of a dioecious fig by their mutualistic pollinating wasps. *Proceedings of the Royal Society B-Biological Sciences*, 265, 45-50.
- ARDITTI, J. & GHANI, A. K. A. 2013. Numerical and physical properties of orchid seeds and their biological implications. *New Phytologist*, 145, 367.
- ASCHAN, G. & PFANZ, H. 2006. Why Snowdrop (*Galanthus nivalis* L.) tepals have green marks? *Flora Morphology, Distribution, Functional Ecology of Plants,* 201, 623-632.
- ASHMAN, T. L. 2009. Sniffing out patterns of sexual dimorphism in floral scent. *Functional Ecology*, 23, 852-862.
- AWANG, N. A., HASAN, Z. S. & SHAFIE, M. S. 2013. Morphological study of *Ficus deltoidea* Jack in Malaysia. *J Agric Sci Technol*, 3, 144-150.
- AZHAR, B., PUAN, C. L., AZIZ, N., SAINUDDIN, M., ADILA, N., SAMSUDDIN, S., ASMAH, S., SYAFIQ, M., RAZAK, S. A., HAFIZUDDIN, A., HAWA, A. & JAMIAN, S. 2015. Effects of in situ habitat quality and landscape characteristics in the oil palm agricultural matrix on tropical understory birds, fruit bats and butterflies. *Biodiversity and Conservation*, 24, 3125-3144.
- BADRON, U. H., TALIP, N., MOHAMAD, A. L., AFFENDDI, A. E. A. & JUHARI, A. A.
 A. 2014. Studies on Leaf Venation in Selected Taxa of the Genus *Ficus* L. (Moraceae) in Peninsular Malaysia. *Tropical life sciences research*, 25, 111-125.
- BAIN, A., HARRISON, R. D. & SCHATZ, B. 2014. How to be an ant on figs. *Acta Oecologica-International Journal of Ecology*, 57, 97-108.
- BALCÁZAR-VARGAS, M. P., VAN ANDEL, T. R., WESTERS, P. & ZUIDEMA, P. A. 2015. What drives the vital rates of secondary hemiepiphytes? A first assessment for three species of Heteropsis (Araceae) in the Colombian Amazon. *Journal of Tropical Ecology*, 31, 251-265.

BEDINGER, P. A., BROZ, A. K., TOVAR-MENDEZ, A. & MCCLURE, B. 2017. Pollen-Pistil Interactions and Their Role in Mate Selection. *Plant Physiology*, 173, 79-90.

- BENZING, D. H. 1990. Vascular epiphytes: general biology and related biota, Cambridge, UK, Cambridge University Press.
- BENZING, D. H. 2004. Vascular epiphytes *In:* LOWMAN, M. D. & RINKER, H. B. (eds.) *Forest canopies.* Berlin: Springer Verlag
- BERG, C. C. 1989. Classification and distribution of Ficus. Experientia, 45, 605-611.
- BERG, C. C. & CORNER, E. J. H. 2005. *Flora Malesiana,* National Herbarium Nederland, Leiden.
- BERNAYS, E. A. & CHAPMAN, R. F. 1994. *Host-Plant Selection by Phytophagous Insects,* New York, Chapman & hall.
- BLEHER, B., POTGIETER, C. J., JOHNSON, N. & BÖHNING-GAESE, K. 2003. The Importance of Figs for Frugivores in a South African Coastal Forest. *Journal of Tropical Ecology*, 19, 375-386.
- BOELTER, C. R., DAMBROS, C. S., NASCIMENTO, H. E. M. & ZARTMAN, C. E. 2014. A tangled web in tropical tree-tops: effects of edaphic variation, neighbourhood phorophyte composition and bark characteristics on epiphytes in a central Amazonian forest. *Journal of Vegetation Science*, 25, 1090-1099.
- BORGES, R. M. 2016. On the Air: Broadcasting and Reception of Volatile Messages in Brood-Site Pollination Mutualisms. *In:* BLANDE, J. D. & GLINWOOD, R. (eds.) *Deciphering Chemical Language of Plant Communication.*

BORGES, R. M., BESSIERE, J. M. & HOSSAERT-MCKEY, M. 2008. The chemical ecology of seed dispersal in monoecious and dioecious figs. *Functional Ecology*, 22, 484-493.

- BORGES, R. M. & KJELLBERG, F. 2014. Special Issue: Plant-insect mutualisms and their parasites: figs and fig wasps.
- BOSCH, J., GONZALEZ, A. M. M., RODRIGO, A. & NAVARRO, D. 2009. Plantpollinator networks: adding the pollinator's perspective. *Ecology Letters*, 12, 409-419.
- BOUCEK, Z., WATSHAM, A. & WIEBES, J. T. 1981. The fig wasp fauna of the receptacles of *Ficus thoningii* (Hymenoptera: Chalcidoidea). *Tijdschrift voor Entomologie*, 124, 149-233.
- BRADSHAW, H. D. & SCHEMSKE, D. W. 2003. Allele substitution at a flower colour locus produces a pollinator shift in monkeyflowers. *Nature*, 426, 176-178.
- BRONSTEIN, J. L. 2001. The Costs of Mutualism. *American Zoologist*, Vol. 41, 825-839.
- BRONSTEIN, J. L. & MCKEY, D. 1989. The fig pollinator mutualism: A model system for comparative biology. *Experientia*, 45, 601-604.
- BRUCE, T. J. A. 2015. Interplay between insects and plants: dynamic and complex interactions that have coevolved over millions of years but act in milliseconds. *Journal of Experimental Botany*, 66, 455-465.
- BUNAWAN, H., AMIN, N. M., BUNAWAN, S. N., BAHARUM, S. N. & NOOR, N. M. 2014. Ficus deltoidea Jack: A Review on Its Phytochemical and Pharmacological Importance. Evidence-Based Complementary and Alternative Medicine.
- BURGER, H., DOTTERL, S. & AYASSE, M. 2010. Host-plant finding and recognition by visual and olfactory floral cues in an oligolectic bee. *Functional Ecology*, 24, 1234-1240.

- BURROWS, J. & BURROWS, S. 2003. *Figs of Southern & South-Central Africa,* Hatfield, Umdaus Press.
- CARDELLINA, J. H. 2012. Review of Figs. The Genus *Ficus*. *Journal of Natural Products*, 75, 2270-2270.
- CHAPMAN, C. A., STRUHSAKER, T. T. & LAMBERT, J. E. 2005. Thirty Years of Research in Kibale National Park, Uganda, Reveals a Complex Picture for Conservation. *International Journal of Primatology*, 26, 539-555.
- CHAVES, O. M., BICCA-MARQUES, J. C. & CHAPMAN, C. A. 2018. Quantity and quality of seed dispersal by a large arboreal frugivore in small and large Atlantic forest fragments. *PLOS ONE*, 13.
- CHEN, C., SONG, Q. S., PROFFIT, M., BESSIERE, J. M., LI, Z. B. & HOSSAERT-MCKEY, M. 2009. Private channel: a single unusual compound assures specific pollinator attraction in *Ficus semicordata*. *Functional Ecology*, 23, 941-950.
- CHEN, Y. R., WU, W. J. & CHOU, L. S. 2001. Regulation of Fig Wasps Entry and Egress: The Role of Ostiole of *Ficus microcarpa* L. *Formosan Entomol*, 21, 171-182.
- COMPTON, S. G. 1993. One way to be a fig. African Entomology, 1, 151-159.
- COMPTON, S. G. 2002. Sailing with the wind: dispersal by small flying insects. *In:* BULLOCK, J. M., KENWARD, R. E. & HAILS, R. S. (eds.) *Dispersal ecology* Cambridge, UK: Cambridge University Press.
- COMPTON, S. G., BALL, A. D., COLLINSON, M. E., HAYES, P., RASNITSYN, A. P. & ROSS, A. J. 2010. Ancient fig wasps indicate at least 34 Myr of stasis in their mutualism with fig trees. *Biology Letters*, 6, 838-842.
- COMPTON, S. G., GREHAN, K. & VAN NOORT, S. 2009. A fig crop pollinated by three or more species of agaonid fig wasps. *African Entomology*, 17, 215-222.
- COMPTON, S. G., HOLTON, K. C., RASHBROOK, V. K., VAN NOORT, S., VINCENT, S. & WARE, A. 1991. Studies of Ceratosolen galili, a non - pollinating agaonid fig wasp (Hymenoptera, Agaonidae).
- COMPTON, S. G. & MCLAREN, F. A. C. 1989. Respiratory adaptations in some male fig wasps. *Proceedings of the Koninklijke Nederlandse Akademie Van Wetenschappen Series C-Biological and Medical Sciences*, 92, 57-71.
- COMPTON, S. G. & MUSGRAVE, M. 1993. Host relationships of Ficus burtt-davyi when growing as a strangler fig. *S.Afr.J.Bot*, 59, 425-430.
- COMPTON, S. G., ROSS, S. J. & THORNTON, I. W. B. 1994. Pollinator limitation of fig tree reproduction on the island of Anak-Krakatau (Indonesia). *Biotropica*, 26, 180-186.
- COMPTON, S. G. & VAN NOORT, S. 1992. Southern African fig wasps (Hymenoptera, Chalcidoidea) - resource utilization and host relationships. *Proceedings of the Koninklijke Nederlandse Akademie Van Wetenschappen-Biological Chemical Geological Physical and Medical Sciences*, 95, 423-435.
- COOK, J. M., BEAN, D., POWER, S. A. & DIXON, D. J. 2004. Evolution of a complex coevolved trait: active pollination in a genus of fig wasps. *Journal of Evolutionary Biology*, 17, 238-246.
- COOK, J. M. & RASPLUS, J. Y. 2003. Mutualists with attitude: coevolving fig wasps and figs. *Trends in Ecology & Evolution*, 18, 241-248.
- COOK, J. M. & SEGAR, S. T. 2010. Speciation in fig wasps. *Ecological Entomology*, 35, 54-66.
- CORLETT, R. T. 1993. Sexual dimorphism in the reproductive phenology of *Ficus grossularioides* Burm. f. in Singapore. *Malayan Nature Journal*, 46, 149-155.

CORLETT, R. T. 2017. Frugivory and seed dispersal by vertebrates in tropical and subtropical Asia: An update. *Global Ecology and Conservation*, 11, 1-22.

- CORNER, E. J. H. 1960. Taxonomic notes on *Ficus* Linn., Asia and Australasia. III. Subgen. *Ficus* and section *Ficus*. *Garden's Bulletin Singapore* 17, 416-441.
- CORNER, E. J. H. 1965. Check-list of *Ficus* in Asia and Australia with keys to identification. *Gdns' Bull., Singapore,* 21, 1-186.
- CORNER, E. J. H. 1969. The complex of *Ficus deltoidea*; A recent invasion of the Sunda Shelf. *Philosophical Transactions of the Royal Society of London*, 256, 281-317.
- CORNER, E. J. H. 1997. *Wayside Trees of Malaya,* Kuala Lumpur, Malaysia, Malayan Nature Society.
- CORNILLE, A., UNDERHILL, J. G., CRUAUD, A., HOSSAERT-MCKEY, M., JOHNSON, S. D., TOLLEY, K. A., KJELLBERG, F., VAN NOORT, S. & PROFFIT, M. 2012. Floral volatiles, pollinator sharing and diversification in the fig-wasp mutualism: insights from *Ficus natalensis*, and its two wasp pollinators (South Africa). *Proceedings of the Royal Society B-Biological Sciences*, 279, 1731-1739.
- COYNE, J. A. & ORR, H. A. 2004. *Speciation*, Sunderland, Massachusetts: Sinauer Associates.
- DANIELS, J. D. & LAWTON, R. 1991. Habitat and Host Preferences of Ficus Crassiuscula, A Neotropical Strangling Fig of the Lower-Montane Rain Forest.
- DAYANG NORWANA, A. A. B., KUNJAPPAN, R., CHIN, M., SCHONEVELD, G., POTTER, L. & ANDRIANI, R. 2011. The local impacts of oil palm expansion in Malaysia: An assessment based on a case study in Sabah State. *CIFOR, Bogor, Indonesia kertas kerja* 78, 1-17.
- DE OLIVEIRA, O. S., SOUZA, L., CESAR MUNERATTO, J. & ALBIERO, A. 2010. Morphological and Anatomical Features of the Flowers and Fruits during the Development of *Chamissoa altissima* (Jacq.) Kunth (Amaranthaceae). *Brazilian Archives of Biology and Technology*, 53, 1425-1432.
- DENG, J. Y., FU, R. H., COMPTON, S. G., HU, D. M., ZHANG, L. S., YANG, F., CHEN, Y. & KJELLBERG, F. 2016. Extremely high proportions of male flowers and geographic variation in floral ratios within male figs of *Ficus tikoua* despite pollinators displaying active pollen collection. *Ecology and Evolution*, 6, 607-619.
- DESAKU. 2005. Mas cotek, *Ficus deltoidea* mistletoe fig (Moraceae). <u>http://www.bio-asli.com/Herba/e_Herba/e_mascotek.asp</u> [Online].
- DIEKMANN, M. 1996. Relationship between flowering phenology of perennial herbs and meteorological data in deciduous forests of Sweden. *Canadian Journal of Botany-Revue Canadienne De Botanique*, 74, 528-537.
- DING, Y., LIU, G. F., ZANG, R. G., ZHANG, J., LU, X. H. & HUANG, J. H. 2016. Distribution of vascular epiphytes along a tropical elevational gradient: disentangling abiotic and biotic determinants. *Scientific Reports*, 6.
- DISLICH, C., KEYEL, A. C., SALECKER, J., KISEL, Y., MEYER, K. M., CORRE, M. D., FAUST, H., BASTIAN, H., KNOHL, A., KREFT, H., MEIJIDE, A. & NURDIANSYA, F. 2015. Ecosystem functions of oil palm plantations a review. University of Goettingen.
- DUNN, D. W., YU, D. W., RIDLEY, J. & COOK, J. M. 2008. Longevity, early emergence and body size in a pollinating fig wasp implications for stability in a fig-pollinator mutualism. *Journal of Animal Ecology*, 77, 927-935.

- DUTHIE, A. B. & NASON, J. D. 2016. Plant connectivity underlies plant-pollinatorexploiter distributions in *Ficus petiolaris* and associated pollinating and nonpollinating fig wasps. *Oikos*, 125, 1597-1606.
- ELIAS, L. G., SILVA, D. B., SILVA, R., PENG, Y. Q., YANG, D. R., LOPES, N. P. & PEREIRA, R. A. S. 2018. A comparative venomic fingerprinting approach reveals that galling and non-galling fig wasp species have different venom profiles. *Plos One,* 13.
- ELLWOOD, M. D. F. & FOSTER, W. A. 2004. Doubling the estimate of invertebrate biomass in a rainforest canopy. *Nature*, 429, 549.
- ELZINGA, J. A., ATLAN, A., BIERE, A., GIGORD, L., WEIS, A. E. & BERNASCONI, G. 2007. Time after time: flowering phenology and biotic interactions. *Trends in Ecology & Evolution*, 22, 432-439.
- FAYLE, T. M., ELLWOOD, M. D. F., TURNER, E. C., SNADDON, J. L., YUSAH, K. M.& FOSTER, W. A. 2008. Bird's nest ferns: Islands of biodiversity in the rainforest canopy. *Antenna*, 32, 34-37.
- FENNER, M. 1998. The phenology of growth and reproduction in plants. *Perspectives in Plant Ecology, Evolution and Systematics*, 1, 78-91.
- FERRER, M. M., GOOD-AVILA, S. V., MONTANA, C., DOMINGUEZ, C. A. & EGUIARTE, L. E. 2009. Effect of variation in self-incompatibility on pollen limitation and inbreeding depression in *Flourensia cernua* (Asteraceae) scrubs of contrasting density. *Annals of Botany*, 103, 1077-1089.
- FISHER, R. A. 1958. The genetical theory of natural selection, New YorkDover.
- FOSTER, W. A., SNADDON, J. L., TURNER, E. C., FAYLE, T. M., COCKERILL, T. D., ELLWOOD, M. D. F., BROAD, G. R., CHUNG, A. Y. C., EGGLETON, P., KHEN, C. V. & YUSAH, K. M. 2011. Establishing the evidence base for maintaining biodiversity and ecosystem function in the oil palm landscapes of South East Asia. *Philosophical Transactions of the Royal Society B-Biological Sciences*, 366, 3277-3291.
- GALIL, J. & EISIKOWITCH, D. 1968. On the pollination ecology of *Ficus sycomorus* in east Africa. *Ecology*, 49, 259–269.
- GALIL, J. & NEEMAN, G. 1977. Pollen transfer and pollination in thecommon fig (*Ficus carica* L.). *New Phytol* 79, 163-171.
- GARCÍA, D. & CHACOFF, N. P. 2007. Scale-Dependent Effects of Habitat Fragmentation on Hawthorn Pollination, Frugivory, and Seed Predation. *Conservation Biology*, 21, 400-411.
- GARDENER, M. C. & GILLMAN, M. P. 2002. The taste of nectar a neglected area of pollination ecology. *Oikos*, 98, 552-557.
- GAUTIER-HION, A. & MICHALOUD, G. 1989. Are Figs Always Keystone Resources for Tropical Frugivorous Vertebrates? A Test in Gabon. *Ecology*, 70, 1826-1833.
- GHANA, S., SULEMAN, N. & COMPTON, S. G. 2012. Factors Influencing Realized Sex Ratios in Fig Wasps: Double Oviposition and Larval Mortalities. *Journal of Insect Behavior*, 25, 254-263.
- GHANA, S., SULEMAN, N. & COMPTON, S. G. 2015a. Ability to gall: the ultimate basis of host specificity in fig wasps? *Ecological Entomology*, 40, 280-291.
- GHANA, S., SULEMAN, N. & COMPTON, S. G. 2015b. A comparison of pollinator fig wasp development in figs of *Ficus montana* and its hybrids with *Ficus* asperifolia. Entomologia Experimentalis Et Applicata, 156, 225-237.

- GHANA, S., SULEMAN, N. & COMPTON, S. G. 2017. Style length variation in male and female figs: development, inheritance, and control of pollinator oviposition. *Entomologia Experimentalis Et Applicata*, 162, 41-50.
- GHAZOUL, J. & SHEIL, D. 2010. *Tropical rain forest ecology, diversity and conservation,* United Kingdom, Oxford Biology.
- GIBERNAU, M., HOSSAERT-MCKEY, M., ANSTETT, M. C. & KJELLBERG, F. 1996. Consequences of protecting flowers in a fig: a one-way trip for pollinators? *Journal of Biogeography*, 23, 425-432.
- GLOVER, B. J. 2011. Pollinator Attraction: The Importance of Looking Good and Smelling Nice. *Current Biology*, 21, 307-309.
- GOLDBLATT, P. & MANNING, J. C. 2000. The Long-Proboscid Fly Pollination System in Southern Africa. *Annals of the Missouri Botanical Garden*, 87, 146-170.
- GONZAGA, A. D. 2012. Fig-fig pollinator interaction in lowland tropical rainforests of Peninsular Malaysia. Doctor of Philosophy, Universiti Kebangsaan Malaysia.
- GOODMAN, S. M. & GANZHORN, J. U. 1997. Rarity of figs (*Ficus*) on Madagascar and its relationship to a depauperate frugivore community. *Rev. Eco*, 52, 321-329.
- GOSSNER, M. M., PASALIC, E., LANGE, M., LANGE, P., BOCH, S., HESSENMOLLER, D., MULLER, J., SOCHER, S. A., FISCHER, M., SCHULZE, E. D. & WEISSER, W. W. 2014. Differential Responses of Herbivores and Herbivory to Management in Temperate European Beech. *Plos One*, 9.
- GRAFEN, A. & GODFRAY, H. C. J. 1991. Vicarious selection explains some paradoxes in dioecious fig;pollinator systems. *Proceedings of the Royal Society of London. Series B: Biological Sciences*, 245, 73-76.
- GRANDI, G. 1920. Studio morfologico e biologico della *Blastophaga psenes* (L.). *Boll. Lab. Zool,* 14, 63-204.
- GREEFF, J. & NEWMAN, D. V. K. 2011. Testing models of facultative sex ratio adjustment in the pollinating fig wasp Platyscapa awekei.
- GREEFF, J. M. 2002. Mating system and sex ratios of a pollinating fig wasp with dispersing males. *Proceedings of the Royal Society B-Biological Sciences*, 269, 2317-2323.
- HAHNKE, R. L., MEIER-KOLTHOFF, J. P., GARCÍA-LÓPEZ, M., MUKHERJEE, S., HUNTEMANN, M., IVANOVA, N. N., WOYKE, T., KYRPIDES, N. C., KLENK, H.-P. & GÖKER, M. 2016. Genome-Based Taxonomic Classification of Bacteroidetes. *Frontiers in Microbiology*, 7.
- HAKIMAN, M. & MAZIAH, M. 2009. Non enzymatic and enzymatic antioxidant activities in aqueous extract of different *Ficus deltoidea* accessions. *Journal of Medicinal Plants Research*, 3, 120-131.
- HAMILTON, W. D. 1967. Extraordinary Sex Ratios. Science, 156, 477.
- HAO, G. Y., CAO, K. F. & GOLDSTEIN, G. 2016. Hemiepiphytic Trees: *Ficus* as a Model System for Understanding Hemiepiphytism. *Tree Physiology*, 6, 3-24.
- HAO, G. Y., SACK, L., WANG, A. Y., CAO, K. F. & GOLDSTEIN, G. 2010. Differentiation of leaf water flux and drought tolerance traits in hemiepiphytic and non-hemiepiphytic *Ficus* tree species. *Functional Ecology*, 24, 731-740.
- HAO, G. Y., WANG, A. Y., SACK, L., GOLDSTEIN, G. & CAO, K. F. 2013. Is hemiepiphytism an adaptation to high irradiance? Testing seedling responses to light levels and drought in hemiepiphytic and non-hemiepiphytic *Ficus*. *Physiologia Plantarum*, 148, 74-86.

HARDER, L. D., JORDAN, C. Y., GROSS, W. E. & ROUTLEY, M. B. 2004. Beyond floricentrism: The pollination function of inflorescences. *Plant Species Biology*, 19, 137-148.

- HARRISON, R. D. 2005. Figs and the diversity of tropical rainforests. *Bioscience*, 55, 1053-1064.
- HARRISON, R. D. & RASPLUS, J. 2006. Dispersal of fig pollinators in Asian tropical rain forests. *Journal of Tropical Ecology*, 22, 631-639.
- HARRISON, R. D. & YAMAMURA, N. 2003. A few more hypotheses for the evolution of dioecy in figs (*Ficus*, Moraceae). *Oikos*, 100, 628-635.
- HARRISON, R. D., YAMAMURA, N. & INOUE, T. 2000. Phenology of a common roadside fig in Sarawak. *Ecological Research*, 15, 47-61.
- HEMBRY, D. H. & ALTHOFF, D. M. 2016. Diversification and coevolution in brood pollination mutualisms: Windows into the role of biotic interactions in generating biological diversity. *American Journal of Botany*, 103, 1783-1792.
- HEMP, A. 2006. Continuum or zonation? Altitudinal gradients in the forest vegetation of Mt. Kilimanjaro. *Plant Ecology*, 184, 27-42.
- HERRE, E. A. 1989. Coevolution of reproductive characteristics in 12 species of New World figs and their pollinator wasps. *Experientia*, 45, 637-647.
- HERRE, E. A., JANDER, K. C. & MACHADO, C. A. 2008. Evolutionary Ecology of Figs and Their Associates: Recent Progress and Outstanding Puzzles. *Annual Review of Ecology Evolution and Systematics*, 39, 439-458.
- HERRE, E. A., MACHADO, C. A., BERMINGHAM, E., NASON, J. D., WINDSOR, D.
 M., MCCAFFERTY, S. S., VANHOUTEN, W. & BACHMANN, K. 1996.
 Molecular phylogenies of figs and their pollinator wasps. *Journal of Biogeography*, 23, 521-530.
- HILL, D. S. 1967. Figs (*Ficus* spp.) and fig-wasps (Chalcidoidea) *Journal of Natural History,* 1.
- HIROTA, S. K., NITTA, K., KIM, Y., KATO, A., KAWAKUBO, N., YASUMOTO, A. A. & YAHARA, T. 2012. Relative Role of Flower Color and Scent on Pollinator Attraction: Experimental Tests using F1 and F2 Hybrids of Daylily and Nightlily. *Plos One*, 7.
- HODGKISON, R., AYASSE, M., HÄBERLEIN, C., SCHULZ, S., ZUBAID, A., MUSTAPHA, W. A. W., KUNZ, T. H. & KALKO, E. K. V. 2013. Fruit bats and bat fruits: the evolution of fruit scent in relation to the foraging behaviour of bats in the New and Old World tropics. *Functional Ecology*, 27, 1075-1084.
- HOLBROOK, N. M. & PUTZ, F. E. 1996a. From epiphyte to tree: differences in leaf structure and leaf water relations associated with the transition in growth form in eight species of hemiepiphytes. *Plant, Cell & Environment,* 19, 631-642.
- HOLBROOK, N. M. & PUTZ, F. E. 1996b. Physiology of tropical vines and hemiepiphytes: plants that climb up and plants that climb down. *In:* MULKEY, S. S., CHAZDON, R. L. & SMITH, A. P. (eds.) *Tropical forest plant ecophysiology.* New York, USA: Chapman & Hall.
- HONG, S. P., RONSE DE CRAENE, L. & SMETS, E. 1998. Systematic significance of tepal surface morphology in tribes Persicarieae and Polygoneae (Polygonaceae). *Botanical Journal of The Linnean Society*, 127, 91-116.
- HOSSAERT-MCKEY, M. & BRONSTEIN, J. L. 2001. Self-pollination and its costs in a monoecious fig (*Ficus aurea*, Moraceae) in a highly seasonal subtropical environment. *American Journal of Botany*, 88, 685-692.

- HOSSAERT-MCKEY, M., GIBERNAU, M. & FREY, J. E. 1994. Chemosensory attraction of fig wasps to substances produced by receptive figs. *Entomologia Experimentalis Et Applicata*, 70, 185-191.
- HOSSAERT-MCKEY, M., PROFFIT, M., SOLER, C. C. L., CHEN, C., BESSIERE, J. M., SCHATZ, B. & BORGES, R. M. 2016. How to be a dioecious fig: Chemical mimicry between sexes matters only when both sexes flower synchronously. *Scientific Reports*, 6.
- HOSSAERT-MCKEY, M., SOLER, C., SCHATZ, B. & PROFFIT, M. 2010. Floral scents: their roles in nursery pollination mutualisms. *Chemoecology*, 20, 75-88.
- HOWELL, A. D. & ALARCON, R. 2007. Osmia bees (Hymenoptera : Megachilidae) can detect nectar-rewarding flowers using olfactory cues. *Animal Behaviour*, 74, 199-205.
- INOUYE, D. W. 2008. Effects of climate change on phenology, frost damage, and floral abundance of montane wildflowers. *Ecology*, 89, 353-362.
- IWATA, T., NAGASAKI, O., ISHII, H. S. & USHIMARU, A. 2012. Inflorescence architecture affects pollinator behaviour and mating success in *Spiranthes sinensis* (Orchidaceae). *New Phytologist*, 193, 196-203.
- JAUHARLINA, J., LINDQUIST, E. E., QUINNELL, R. J., ROBERTSON, H. G. & COMPTON, S. G. 2012. Fig wasps as vectors of mites and nematodes. *African Entomology*, 20, 101-110.
- JOHN, R. 1686. Historia Plantarum, London, Clark.
- JORDANO, P. 2017. What is long-distance dispersal? And a taxonomy of dispersal events. *Journal of Ecology*, 105, 75-84.
- JOUSSELIN, E., HOSSAERT-MCKEY, M., VERNET, D. & KJELLBERG, F. 2001. Egg deposition patterns of fig pollinating wasps: implications for studies on the stability of the mutualism. *Ecological Entomology*, 26, 602-608.
- JOUSSELIN, E., RASPLUS, J. Y. & KJELLBERG, F. 2003. Convergence and coevolution in a mutualism: Evidence from a molecular phylogeny of *Ficus*. *Evolution*, 57, 1255-1269.
- KAHNT, B., MONTGOMERY, G. A., MURRAY, E., KUHLMANN, M., PAUW, A., MICHEZ, D., PAXTON, R. J. & DANFORTH, B. N. 2017. Playing with extremes: Origins and evolution of exaggerated female forelegs in South African Rediviva bees. *Molecular Phylogenetics and Evolution*, 115, 95-105.
- KANNAN, R. & JAMES, D. A. 1999. Fruiting Phenology and the Conservation of the Great Pied Hornbill (*Buceros bicornis*) in the Western Ghats of Southern India. *Biotropica*, 31, 167-177.
- KAUFMANN, S., MCKEY, D., HOSSAERT-MCKEY, M. & HORVITZ, C. 1991. Adaptations for a two-phase seed dispersal system involving vertebrates and ants in a hemiepiphytic fig (Ficus Microcarpa: Moraceae).
- KERDELHUÉ, C., HOCHBERG, M. E. & RASPLUS, J.-Y. 1997. Active Pollination of *Ficus sur* by Two Sympatric Fig Wasp Species in West Africa. *Biotropica*, 29, 69-75.
- KERDELHUE, C. & RASPLUS, J. Y. 1996. The evolution of dioecy among *Ficus* (Moraceae): An alternative hypothesis involving non-pollinating fig wasp pressure on the fig-pollinator mutualism. *Oikos*, 77, 163-166.
- KERDELHUE, C., ROSSI, J. P. & RASPLUS, J. Y. 2000. Comparative community ecology studies on old world figs and fig wasps. *Ecology*, 81, 2832-2849.
- KERDKAEW, T. 2018. *Frugivory, seed dispersal and gene flow of riparian figs in Western Thailand.* Doctor of Philosophy, University of Leeds.

- KINOSHITA, M., KASUYA, E. & YAHARA, T. 2002. Effects of time-dependent competition for oviposition sites on clutch sizes and offspring sex ratios in fig wasp.
- KJELLBERG, F., BRONSTEIN, J. L., VAN GINKEL, G., GREEFF, J. M., MOORE, J. C., BOSSU-DUPRIEZ, N., CHEVOLOT, M. & MICHALOUD, G. 2005a. Clutch size: a major sex ratio determinant in fig pollinating wasps? *Comptes Rendus Biologies*, 328, 471-476.
- KJELLBERG, F., JOUSSELIN, E., BRONSTEIN, J. L., PATEL, A., YOKOYAMA, J. & RASPLUS, J. Y. 2001. Pollination mode in fig wasps: the predictive power of correlated traits. *Proceedings of the Royal Society B-Biological Sciences*, 268, 1113-1121.
- KJELLBERG, F., JOUSSELIN, E., HOSSAERT-MCKEY, M. & RASPLUS, J. Y. 2005b.
 Biology, ecology and evolution of fig pollinating wasp (Chalcidoidea:Agaonidae). *In:* RAMAN, A., SCHAEFER, C. W. & WITHERS, T.
 M. (eds.) *Biology, ecology and evolution of gall inducing arthropods.* PlymouthUnited Kingdom: Science Publishers Inc.
- KJELLBERG, F. & MAURICE, S. 1989. Seasonality in the reproductive phenology of *Ficus* its evolution and consequences. *Experientia*, 45, 653-660.
- KJELLBERG, F., SULEMAN, N., RAJA, S., TAYOU, A., HOSSAERT-MCKEY, M. & COMPTON, S. G. 2014. Some pollinators are more equal than others: Factors influencing pollen loads and seed set capacity of two actively and passively pollinating fig wasps. *Acta Oecologica-International Journal of Ecology*, 57, 73-79.
- KLAHRE, U., GURBA, A., HERMANN, K., SAXENHOFER, M., BOSSOLINI, E., GUERIN, P. M. & KUHLEMEIER, C. 2011. Pollinator Choice in *Petunia* Depends on Two Major Genetic Loci for Floral Scent Production. *Current Biology*, 21, 730-739.
- KOBMOO, N., HOSSAERT-MCKEY, M., RASPLUS, J. Y. & KJELLBERG, F. 2010. *Ficus racemosa* is pollinated by a single population of a single agaonid wasp species in continental South-East Asia. *Molecular Ecology*, 19, 2700-2712.
- KOH, L. P. 2008. Can oil palm plantations be made more hospitable for forest butterflies and birds? . *Journal of Applied Ecology*, 45.
- KOMAE, K. & MISAKI, A. 1993. "Ai-Geok", A unique gel-forming polysaccharide of Ficus awkeotsang: Structure, gelation mechanism and utilization of the oligosaccharides In: NISHINARI, K. & E. (eds.) Food Hydrocolloids Structures, Properties, and Functions. New York: Plenum Press.
- KORINE, C., KALKO, E. K. V. & HERRE, E. A. 2000. Fruit characteristics and factors affecting fruit removal in a Panamanian community of strangler figs. *Oecologia*, 123, 560-568.
- KRAFT, N. J. B., GODOY, O. & LEVINE, J. M. 2015a. Plant functional traits and the multidimensional nature of species coexistence. *Proceedings of the National Academy of Sciences of the United States of America*, 112, 797-802.
- KRAFT, T. S., WRIGHT, S. J., I, T., LUCAS, P. W., OUFIERO, C. E., NOOR, M. N. S., SUN, I. F. & DOMINY, N. J. 2015b. Seed size and the evolution of leaf defences. *Journal of Ecology*, 103, 1057-1068.
- KRAMER, G. 2011. *Palm tree susceptibility to hemi-epiphytic parasitism by ficus.* Master of Science, University of Florida.
- KREFT, H., KOSTER, N., KUPER, W., NIEDER, J. & BARTHLOTT, W. 2004. Diversity and biogeography of vascular epiphytes in Western Amazonia, Yasuni, Ecuador. *Journal of Biogeography*, 31, 1463-1476.

- KRISHNAN, A., PRAMANIK, G. K., REVADI, S. V., VENKATESWARAN, V. & BORGES, R. M. 2014. High Temperatures Result in Smaller Nurseries which Lower Reproduction of Pollinators and Parasites in a Brood Site Pollination Mutualism. *Plos One*, 9.
- KUSHAIRI, A., SINGH, R. & ONG-ABDULLAH, M. 2017. The oil palm industry in Malaysia: thriving with transformative technologies. *Journal of Oil Palm Research*, 29, 431-439.
- KUSUMI, J., AZUMA, H., TZENG, H. Y., CHOU, L. S., PENG, Y. Q., NAKAMURA, K.
 & SU, Z. H. 2012. Phylogenetic analyses suggest a hybrid origin of the figs (Moraceae: *Ficus*) that are endemic to the Ogasawara (Bonin) Islands, Japan. *Molecular Phylogenetics and Evolution*, 63, 168-179.
- KUWABARA, A., IKÉGĂMI, K., KOSHIBA, T. & NÁGATA, T. 2003. Effects of ethylene and abscisic acid upon heterophylly in Ludwigia arcuata (Onagraceae). *Planta*, 217, 880-887.
- KUWABARA, A. & NAGATA, T. 2006. Cellular basis of developmental plasticity observed in heterophyllous leaf formation of *Ludwigia arcuata* (Onagraceae). *Planta*, 224, 761-770.
- LACHAISE, D., CARIOU, M., DAVID, J., LEMEUNIER, F., TSACAS, L. & ASHBURNER, M. 1988. Historical Biogeography of the *Drosophila melanogaster* Species Subgroup.
- LAMAN, T. G. & WEIBLEN, G. 1998. Figs of Gunung Palung National Park (west Kalimantan, Indonesia). *Tropical Biodiversity*, *5*, 245-297.
- LAMBERT, F. 1991. The conservation of fig-eating birds in Malaysia. *Biological Conservation*, 58, 31-40.
- LANSKY, E. P. & PAAVILAINEN, H. M. 2011. *Figs: The Genus Ficus,* London CRC Press Boca Raton.
- LANSKY, E. P., PAAVILAINEN, H. M., PAWLUS, A. D. & NEWMAN, R. A. 2008. *Ficus* spp. (fig): Ethnobotany and potential as anticancer and anti-inflammatory agents. *Journal of Ethnopharmacology*, 119, 195-213.
- LARSSON, M. 2005. Higher pollinator effectiveness by specialist than generalist flower-visitors of unspecialized *Knauita arvensis* (Dipsacaceae). *Oecologia*, 146, 394-403.
- LEE, Y. N. 2011. Increase Malaysia Palm Oil Production efficiency. Doctor of Philosophy, University of Bergen.
- LEES, A. C., MOURA, N. G., DE ALMEIDA, A. S. & VIEIRA, I. C. G. 2015. Poor Prospects for Avian Biodiversity in Amazonian Oil Palm. *Plos One,* 10.
- LEIGH, A. & NICOTRA, A. B. 2003. Sexual dimorphism in reproductive allocation and water use efficiency in *Maireana pyramidata* (Chenopodiaceae), a dioecious, semi-arid shrub. *Australian Journal of Botany*, 51, 509-514.
- LI, Z. T., PENG, Y. Q., WEN, X. L. & JANDER, K. C. 2016. Selective resource allocation may promote a sex ratio in pollinator fig wasps more beneficial for the host tree. *Scientific Reports*, 6.
- LIN, S. Y., CHOU, L. S., DI GIUSTO, B. & BAIN, A. 2015. Sexual specialization in phenology in dioecious *Ficus benguetensis* and its consequences for the mutualism. *Botanical Studies*, 56.
- LINNAEUS, C. 1753. Species Plantarum, CreateSpace Independent Publishing Platform.
- LIU, C., YANG, D. R., COMPTON, S. G. & PENG, Y. Q. 2013. Larger Fig Wasps Are More Careful About Which Figs to Enter - With Good Reason. *Plos One,* 8.

- LIU, G. X., YANG, D. R., PENG, Y. Q. & COMPTON, S. G. 2015. Complementary fruiting phenologies facilitate sharing of one pollinator fig wasp by two fig trees. *Journal of Plant Ecology*, 8, 197-206.
- LOHMAN, D. J., DE BRUYN, M., TIMOTHY, P., VON RINTELEN, K., HALL, R., NG, P. K. L., SHIH, H.-T., CARVALHO, G. R. & VON RINTELEN, T. 2011. Biogeography of the Indo-Australian Archipelago. *Annual Review of Ecology, Evolution, and Systematics,* 42, 205-226.
- LOMASCOLO, S., LEVEY, D., KIMBALL, R., BOLKER, B. & ALBORN, H. T. 2010. Dispersers shape fruit diversity in Ficus (Moraceae).
- LUSKIN, M. S. & POTTS, M. D. 2011. Microclimate and habitat heterogeneity through the oil palm lifecycle. *Basic and Applied Ecology*, 12, 540-551.
- MACGOWAN, I. & COMPTON, S. G. 2018. A new species of *Silba* (Diptera; Lonchaeidae) associated with figs. *Zootaxa*, 4455, 196-200.
- MACHADO, C. A., JOUSSELIN, E., KJELLBERG, F., COMPTON, S. G. & HERRE, E. A. 2001. Phylogenetic relationships, historical biogeography and character evolution of fig-pollinating wasps. *Proceedings of the Royal Society B-Biological Sciences*, 268, 685-694.
- MACHADO, C. A., ROBBINS, N., GILBERT, M. T. P. & HERRE, E. A. 2005. Critical review of host specificity and its coevolutionary implications in the fig/fig-wasp mutualism. *Proceedings of the National Academy of Sciences of the United States of America*, 102, 6558-6565.
- MACKAY, K. D., GROSS, C. L. & ROSSETTO, M. 2018. Small populations of fig trees offer a keystone food resource and conservation benefits for declining insectivorous birds. *Global Ecology and Conservation*, 14.
- MALDONADO, M. B., LOMASCOLO, S. B. & VAZQUEZ, D. P. 2013. The importance of pollinator generalization and abundance for the reproductive success of a generalist plant. *Plos One*, 8.
- MALE, T. D. & ROBERTS, G. E. 2005. Host associations of the strangler fig Ficus watkinsiana in a subtropical Queensland rain forest. *Austral Ecology*, 30, 229-236.
- MARTEL, V., SHUKER, D. M., BOULTON, R. A., DAMIENS, D. & BOIVIN, G. 2016. Sex allocation and the evolution of insemination capacity under local mate competition. *Entomologia Experimentalis et Applicata*, 159, 230-242.
- MARTINSON, E. O., JANDER, K. C., PENG, Y. Q., CHEN, H. H., MACHADO, C. A., ARNOLD, A. E. & HERRE, E. A. 2014. Relative investment in egg load and poison sac in fig wasps: Implications for physiological mechanisms underlying seed and wasp production in figs. *Acta Oecologica-International Journal of Ecology*, 57, 58-66.
- MAT, N., ROSNI, N. A., AB RASHID, N. Z., HARON, N., NOR, Z. M., NUDIN, N. F. H., YUNUS, A. & ALI, A. M. 2012. Leaf Morphological Variations and Heterophylly in *Ficus deltoidea* Jack (Moraceae). *Sains Malaysiana*, 41, 527-538.
- MAYR, E. 1942. Systematics and the Origin of Species, Columbia University Press.
- MEHLTRETER, K., FLORES-PALACIOS, A. & GARCÍA-FRANCO, J. 2005. Host preferences of low-trunk vascular epiphytes in a cloud forest of Veracruz, Mexico. *Journal of Tropical Ecology*, 21, 651-660.
- MICHALOUD, G., MICHALOUD-PELLETIER, S., WIEBES, J. T. & BERG, C. C. 1985. The co-occurrence of two pollinating species of fig wasp and one species of fig. *Proceedings. Koninklijke Nederlandse Akademie van Wetenschappen,* 88, 107-109.

- MILET-PINHEIRO, P., AYASSE, M., DOBSON, H. E. M., SCHLINDWEIN, C., FRANCKE, W. & DOTTERL, S. 2013. The Chemical Basis of Host-Plant Recognition in a Specialized Bee Pollinator. *Journal of Chemical Ecology*, 39, 1347-1360.
- MISBAH, H., AZIZ, A. A. & AMINUDIN, N. 2013. Antidiabetic and antioxidant properties of *Ficus deltoidea* fruit extracts and fractions. *BMC Complementary and Alternative Medicine*, 13
- MIZUKI, I., ISHIDA, K. & KIKUZAWA, K. 2005. Sexual and vegetative reproduction in the aboveground part of a dioecious clonal plant, *Dioscorea japonica* (Dioscoreaceae). *Ecological Research*, 20, 387-393.
- MOE, A. M., ROSSI, D. R. & WEIBLEN, G. D. 2011. Pollinator sharing in dioecious figs (*Ficus*: Moraceae). *Biological Journal of the Linnean Society*, 103, 546-558.
- MOLBO, D., MACHADO, C. A., SEVENSTER, J. G., KELLER, L. & HERRE, E. A. 2003. Cryptic species of fig-pollinating wasps: Implications for the evolution of the fig-wasp mutualism, sex allocation, and precision of adaptation. *Proceedings of the National Academy of Sciences of the United States of America*, 100, 5867-5872.
- MONDRAGON, D. & CALVO-IRABIEN, L. M. 2006. Seed dispersal and germination of the epiphyte Tillandsia brachycaulos (Bromeliaceae) in a tropical dry forest, Mexico, SPIE.
- MOORE, J. C. 2001. The ecology and evolution of gynodioecious figs and their pollinators. Doctor of Philosophy, University of Leeds.
- MOORE, J. C., DUNN, A. M., COMPTON, S. G. & HATCHER, M. J. 2003. Foundress re-emergence and fig permeability in fig tree-wasp mutualisms. *Journal of Evolutionary Biology*, 16, 1186-1195.
- MOORE, J. C., ZAVODNA, M., COMPTON, S. G. & GILMARTIN, P. M. 2005. Sex ratio strategies and the evolution of cue use. *Proceedings. Biological sciences*, 272, 1287-1294.
- MORALES, C. L. & TRAVESET, A. 2008. Interspecific pollen transfer: Magnitude, prevalence and consequences for plant fitness. *Critical Reviews in Plant Sciences*, 27, 221-238.
- MUMFORD, A. 2008. Orangutan Population Survey of Bawan, Central Kalimantan, Borneo, using Rapid Assessment Nest Survey Bachelor of Science, University of Oxford.
- MUÑOZ, A. A., CHACÓN, P., PÉREZ, F., BARNERT, E. S. & ARMESTO, J. J. 2003. Diversity and host tree preferences of vascular epiphytes and vines in a temperate rainforest in southern Chile. *Australian Journal of Botany*, 51, 381-391.
- NADARAJAH, P. & NAWAWI, A. 1993. Mycorrhizal status of epiphytes in Malaysian oil palm plantations. *Mycorrhiza*, 4, 21-25.
- NADKARNI, N. M., MERWIN, C. K. & NIEDER, J. 2001. Forest Canopies, Plant Diversity. *Encyclopedia of Biodiversity*, 3, 27-40.
- NAGIAH, C. & AZMI, R. 2012. A review of smallholder oil palm production: challenges and opportunities for enhancing sustainability - a Malaysian perspective. *Journal of Oil Palm & The Environment*, 3, 114-120.
- NAKAYAMA, H., SINHA, N. R. & KIMURA, S. 2017. How do plants and phytohormones accomplish heterophylly, leaf phenotypic plasticity, in response to environmental cues. *Frontiers in Plant Science*, 8.

- NEFDT, R. J. C. & COMPTON, S. G. 1996. Regulation of seed and pollinator production in the fig fig wasp mutualism. *Journal of Animal Ecology*, 65, 170-182.
- NEPI, M., GRASSO, D. A. & MANCUSO, S. 2018. Nectar in Plant-Insect Mutualistic Relationships: From Food Reward to Partner Manipulation. *Frontiers in plant science*, 9, 1063-1063.
- NOVOTNY, V., MILLER, S. E., BAJE, L., BALAGAWI, S., BASSET, Y., CIZEK, L., CRAFT, K. J., DEM, F., DREW, R. A. I., HULCR, J., LEPS, J., LEWIS, O. T., POKON, R., STEWART, A. J. A., SAMUELSON, G. A. & WEIBLEN, G. D. 2010. Guild-specific patterns of species richness and host specialization in plantherbivore food webs from a tropical forest. *Journal of Animal Ecology*, 79, 1193-1203.
- NUR FATIHAH, H. N., NASHRIYAH, M., NOR ZAIMAH, A. R., KHAIRIL, M. & ALI, A. M. 2014. Leaf morphology and anatomy of 7 varieties of *Ficus deltoidea* (Moraceae). *Turkish Journal of Botany*, 38, 677-685.
- ØDEGAARD, F. 2000. The relative importance of trees versus lianas as hosts for phytophagous beetles (Coleoptera) in tropical forests. *Journal of Biogeography*, 27, 283-296.
- OETTLI, P., BEHERA, S. K. & YAMAGATA, T. 2018. Climate based predictability of oil palm tree yield in Malaysia. *Scientific Reports* 8, 22-71.
- PANSARIN, E. R. & MACIEL, A. A. 2017. Evolution of pollination systems involving edible trichomes in orchids. *Aob Plants*, 9.
- PARKER, A. J., WILLIAMS, N. M. & THOMSON, J. D. 2016. Specialist pollinators deplete pollen in the spring ephemeral wildflower *Claytonia virginica*. *Ecology and Evolution*, 6, 5169-5177.
- PARRISH, T. L., KOELEWIJN, H. P., VAN DIJK, P. J. & KRUIJT, M. 2003. Genetic evidence for natural hybridization between species of dioecious *Ficus* on island populations. *Biotropica*, 35, 333-343.
- PATEL, A. 1996. Variation in a mutualism: Phenology and the maintenance of gynodioecy in two Indian fig species. *Journal of Ecology*, 84, 667-680.
- PATEL, A. & HOSSAERT-MCKEY, M. 2000. Components of reproductive success tn two dioecious fig species, *Ficus exasperata* and *Ficus hispida*. *Ecology*, 81, 2850-2866.
- PATEL, A., HOSSAERTMCKEY, M. & MCKEY, D. 1993. *Ficus* pollinator research in India past, present and future. *Current Science*, 65, 243-253.
- PATEL, A. & MCKEY, D. 1998. Sexual specialization in two tropical dioecious figs. *Oecologia*, 115, 391-400.
- PEARCY, R. W., VALLADARES, F., WRIGHT, S. J. & DE PAULIS, E. L. 2004. A functional analysis of the crown architecture of tropical forest Psychotria species: do species vary in light capture efficiency and consequently in carbon gain and growth? *Oecologia*, 139, 163-177.
- PEH, K. & FONG, L. C. 2003. Seed dispersal agents of two *Ficus* species in a disturbed tropical forest. *Ornithological Science*, 2, 119-125.
- PENG, G., DENG, L., CUI, W., MING, T. & SHEN, W. 2009. Remote sensing monitoring of tobacco field based on phenological characteristics and time series image—A case study of Chengjiang County, Yunnan Province, China. *Chinese Geographical Science*, 19, 186-193.
- PENG, Y. Q., COMPTON, S. G. & YANG, D. R. 2010. The reproductive success of *Ficus altissima* and its pollinator in a strongly seasonal environment: Xishuangbanna, Southwestern China. *Plant Ecology*, 209, 227-236.

- PEREIRA, R. A. S. & PRADO, A. P. 2006. Effect of local mate competition on fig wasp sex ratios. *Brazilian Journal of Biology*, 66, 603-610.
- PETRUZZELLO, M. 2017. *Epiphyte* [Online]. Encyclopedia Britannica. Available: https://www.britannica.com/plant/epiphyte [Accessed 21.10.2018 2018].
- PIGGOTT, A. G. 1980. The fern flora of oilpalm plantations in West Malaysia. *Fern Gazette* 12, 93-102.
- PRESCOTT, G. W., EDWARDS, D. P. & FOSTER, W. A. 2015. Retaining biodiversity in intensive farmland: epiphyte removal in oil palm plantations does not affect yield. *Ecology and Evolution*, 5, 1944-1954.
- PROFFIT, M., CHEN, C., SOLER, C., BESSIERE, J. M., SCHATZ, B. & HOSSAERT-MCKEY, M. 2009. Can chemical signals, responsible for mutualistic partner encounter, promote the specific exploitation of nursery pollination mutualisms? The case of figs and fig wasps. *Entomologia Experimentalis Et Applicata*, 131, 46-57.
- PROFFIT, M. & JOHNSON, S. D. 2009. Specificity of the signal emitted by figs to attract their pollinating wasps: Comparison of volatile organic compounds emitted by receptive syconia of *Ficus sur* and *F. sycomorus* in Southern Africa. *South African Journal of Botany*, 75, 771-777.
- PROKURAT, S. 2013. Palm Oil- Strategic Source of Renewable Energy in Indonesia and Malaysia. *Journal of Modern Science* 18, 425-443.
- PUTZ, F. E. & HOLBROOK, N. M. 1986. Notes on the natural history of hemiepiphytes. *Selbyana*, 9, 61-69.
- PUTZ, F. E., ROMANO, G. B. & HOLBROOK, N. M. 1995. Comparative phenology of epiphytic and tree-phase strangler figs in a Venezuelan palm savanna. *Biotropica*, 27, 183-189.
- QUARESMA, A. C., PIEDADE, M. T. F., FEITOSA, Y. O., WITTMANN, F. & STEEGE, H. T. 2017. Composition, diversity and structure of vascular epiphytes in two contrasting Central Amazonian floodplain ecosystems. *Acta Botanica Brasilica*, 31, 686-697.
- QUICKE, D. L. J. 2012. We Know Too Little about Parasitoid Wasp Distributions to Draw Any Conclusions about Latitudinal Trends in Species Richness, Body Size and Biology. *Plos One*, 7.
- RAJA, S., SULEMAN, N. & COMPTON, S. G. 2008a. Why do fig wasps pollinate female figs? *Symbiosis*, 45, 25-28.
- RAJA, S., SULEMAN, N. & COMPTON, S. G. 2014. Foundress fig wasps are more likely to re-emerge from older figs. *Journal of Insect Behavior*, 27, 786-790.
- RAJA, S. B., SULEMAN, N. & COMPTON, S. G. 2008b. Why do fig wasps pollinate female figs? *Symbiosis*, 45, 25-28.
- RAJPAR, M. N. & ZAKARIA, M. 2010. Density and diversity of water birds and terrestrial birds at Paya Indah Wetland Reserve, Selangor Peninsular Malaysia. *Journal of Biological Sciences*, 10, 658-666.
- RAMIREZ, W. & MALAVASI, J. 1997. Fig wasps: mechanisms of pollen transfer in *Malvanthera* and *Pharmacosycea* figs (Moraceae). *Revista De Biologia Tropical*, 45, 1635-1640.
- RASPLUS, J. Y. 1996. The one-to-one species specificity of the Ficus-Agaoninae mutualism: How casual?
- RATHCKE, B. J. 2003. Floral longevity and reproductive assurance: seasonal patterns and an experimental test with *Kalmia latifolia* (Ericaceae). *American Journal of Botany*, 90, 1328-1332.

- REVERTÉ, S., RETANA, J., GÓMEZ, J. M. & BOSCH, J. 2016. Pollinators show flower colour preferences but flowers with similar colours do not attract similar pollinators. *Annals of Botany*, 118, 249-257.
- REZANEZHAD, F., PRICE, J. S., QUINTON, W. L., LENNARTZ, B., MILOJEVIC, T. & VAN CAPPELLEN, P. 2016. Structure of peat soils and implications for water storage, flow and solute transport: A review update for geochemists. *Chemical Geology*, 429, 75-84.

RIFFLE, R. L. 1998. The Tropical Look, Portland, OR, Timber Press, Inc.

- ROCKWOOD, L. L. 1985. Seed Weight as a Function of Life Form, Elevation and Life Zone in Neotropical Forests. *Biotropica*, 17, 32-39.
- RODRIGUEZ, L. J., BAIN, A., CHOU, L. S., CONCHOU, L., CRUAUD, A., GONZALES, R., HOSSAERT-MCKEY, M., RASPLUS, J. Y., TZENG, H. Y. & KJELLBERG, F. 2017. Diversification and spatial structuring in the mutualism between *Ficus septica* and its pollinating wasps in insular South East Asia. *Bmc Evolutionary Biology*, 17.
- ROLL, A., NIU, F., MEIJIDE, A., HARDANTO, A., HENDRAYANTO, KNOHL, A. & HOLSCHER, D. 2015. Transpiration in an oil palm landscape: effects of palm age. *Biogeosciences*, 12, 5619-5633.
- RUML, M. & VULIC, T. 2005. Importance of phenological observations and predictions in agriculture. *Journal of Agricultural Sciences*, 50, 217-225.
- SAKAI, S. 2002. A review of brood-site pollination mutualism: plants providing breeding sites for their pollinators. *Journal of Plant Research*, 115, 161-168.
- SALES, K., KERR, L. & GARDNER, J. 2016. Factors influencing epiphytic moss and lichen distribution within Killarney National Park. *Bioscience Horizons: The International Journal of Student Research*, 9, hzw008-hzw008.
- SCARPELLA, E., BARKOULAS, M. & TSIANTIS, M. 2010. Control of leaf and vein development by auxin. *Cold Spring Harbor Perspectives in Biology*, 2.
- SCHMIDT, G. & ZOTZ, G. 2001. Herbivory in the epiphyte, *Vriesea sanguinolenta* Cogn. & amp; Marchal (Bromeliaceae). *Journal of Tropical Ecology*, 16, 829-839.
- SEDEEK, K. E. M., SCOPECE, G., STAEDLER, Y. M., SCHONENBERGER, J., COZZOLINO, S., SCHIESTL, F. P. & SCHLUTER, P. M. 2014. Genic rather than genome-wide differences between sexually deceptive *Ophrys* orchids with different pollinators. *Molecular Ecology*, 23, 6192-6205.
- SHANAHAN, M. & COMPTON, S. G. 2001. Vertical stratification of figs and fig-eaters in a Bornean lowland rain forest: how is the canopy different? *Plant Ecology*, 153, 121-132.
- SHANAHAN, M., SO, S., COMPTON, S. G. & CORLETT, R. 2001. Fig-eating by vertebrate frugivores: a global review. *Biological Reviews*, 76, 529-572.
- SHI, Z. H., YANG, D. R. & PENG, Y. Q. 2006. The style-length of the female florets and their fate in two dioecious species of Xishuangbanna, China. *Trees-Structure and Function*, 20, 410-415.
- SHUKER, D. M. & WEST, S. A. 2004. Information constraints and the precision of adaptation: Sex ratio manipulation in wasps. *Proceedings of the National Academy of Sciences of the United States of America*, 101, 10363-10367.
- SIRISHA, N., SREENIVASULU, M., SANGEETA, K. & MADHUSUDHANA, C. C. 2010. Antioxidant Properties of *Ficus* Species – A Review. *International Journal* of *PharmTech Research*, 2, 2174-2182.

- SITI KHAIRIYAH, M. H., ELNA ELFIRA, S., MOHD HANYSYAM, N., NURDIANA, S., NORASHIRENE, M. J. & FAEZAH, P. 2012. Entomofaunal Diversity of Diptera at FELDA Besout 6 Oil Palm Plantation *IERI Procedia*, 5, 45-50.
- SOFIYANTI, N. 2013. The diversity of epiphytic fern on the oil palm tree (*Elaeis guineensis* Jacq,) in Pekan Baru, Riau. *Jurnal Biologi,* 17, 51-55.
- SOLER, C. C. L., PROFFIT, M., BESSIERE, J. M., HOSSAERT-MCKEY, M. & SCHATZ, B. 2012. Evidence for intersexual chemical mimicry in a dioecious plant. *Ecology Letters*, 15, 978-985.
- SOUTO-VILAROS, D., PROFFIT, M., BUATOIS, B., RINDOS, M., SISOL, M., KUYAIVA, T., ISUA, B., MICHALEK, J., DARWELL, C. T., HOSSAERT-MCKEY, M., WEIBLEN, G. D., NOVOTNY, V. & SEGAR, S. T. 2018. Pollination along an elevational gradient mediated both by floral scent and pollinator compatibility in the fig and fig-wasp mutualism. *Journal of Ecology*, 106, 2256-2273.
- SPENCER, H., WEIBLEN, G. & FLICK, B. 1996. Phenology of *Ficus variegata* in a seasonal wet tropical forest at Cape Tribulation, Australia. *Journal of Biogeography*, 23, 467-475.
- SREEKAR, R., LE, N. T. P. & HARRISON, R. D. 2010. Vertebrate Assemblage at a Fruiting Fig (*Ficus caulocarpa*) in Maliau Basin, Malaysia. *Tropical Conservation Science*, 3, 218-227.
- STARR, F., K., S. & LOOPE, L. 2003. Ficus deltoidea, Mistletoe Fig (Moraceae) United States Geological Survey-Biological Resources Division Haleakala Field Station; Maui, HI, USA.
- STEWARD, G. A. & BEVERIDGE, A. E. 2010. A review of New Zealand Kauri (Agathis australis (D. Don) Lindl.): its ecology, history, growth and potential for management for timber. *New Zealand Journal of Forestry Science* 40, 33-59.
- SU, Z. H., IINO, H., NAKAMURA, K., SERRATO, A. & OYAMA, K. 2008. Breakdown of the one-to-one rule in Mexican fig-wasp associations inferred by molecular phylogenetic analysis. *Symbiosis*, 45, 73-81.
- SULEMAN, N., RAJA, S. & COMPTON, S. G. 2011. A comparison of growth and reproduction, under laboratory conditions, of males and females of a dioecious fig tree. *Plant Systematics and Evolution*, 296, 245-253.
- SULEMAN, N., RAJA, S. & COMPTON, S. G. 2013a. Parasitism of a pollinator fig wasp: mortalities are higher in figs with more pollinators, but are not related to local densities of figs. *Ecological Entomology*, 38, 478-484.
- SULEMAN, N., RAJA, S., QUINNELL, R. J. & COMPTON, S. G. 2013b. Putting your eggs in several baskets: oviposition in a wasp that walks between several figs. *Entomologia Experimentalis Et Applicata*, 149, 85-93.
- SUN, T. P. 2010. Gibberellin-GID1-DELLA: A pivotal regulatory module for plant growth and development. *Plant Physiology*, 154, 567-570.
- SUZANTI, F., KUSWARDANI, R. A., RAHAYU, S. & SUSANTO, A. 2016. Contribution of epiphytes on the canopy insect population in oil palm plantations in North Sumatera. *Journal of Engineering and Applied Sciences*, 11, 6982-6998.
- TARACHAI, Y., COMPTON, S. G. & TRISONTHI, C. 2008. The benefits of pollination for a fig wasp. *Symbiosis*, 45, 29-32.
- TAYOU, A. 1991. *Etude du mutualisme Ficus-pollinisateur: sex ratio et transport du pollen.* Doctor of Philosophy, Universite de Montpellier II.
- TEONG, I. T., FELIX, N. L. L., MOHD, S. & SULAEMAN, A. 2016. Characterization of Soil Organic Matter in Peat Soil with Different Humification Levels using FTIR. *IOP Conference Series: Materials Science and Engineering*, 136, 012010.

- TEUSCHER, M., VORLAUFER, M., WOLLNI, M., BROSE, U., MULYANI, Y. & CLOUGH, Y. 2015. Trade-offs between bird diversity and abundance, yields and revenue in smallholder oil palm plantations in Sumatra, Indonesia. *Biological Conservation*, 186, 306-318.
- THOMAS, P. A. & PACKHAM, J. R. 2007. Ecology of woodlands and forest. Description, dynamics and diversity, Cambridge, University press.
- THORNTON, I. W. B., COMPTON, S. G. & WILSON, C. N. 1996. The role of animals in the colonization of the Krakatau Islands by fig trees (*Ficus* species). *Journal* of Biogeography, 23, 577-592.
- THOROGOOD, C., DALTON, N., IRVINE, A. & HISCOCK, S. 2018. The reproductive biology of two poorly known relatives of the fig (*Ficus*) and insights into the evolution of the fig syconium. *Nordic Journal of Botany*, 36.
- TILMAN, D., FARGIONE, J., WOLFF, B., D'ANTONIO, C., DOBSON, A., HOWARTH, R., SCHINDLER, D., SCHLESINGER, W. H., SIMBERLOFF, D. & SWACKHAMER, D. 2001. Forecasting agriculturally driven global environmental change. *Science*, 292, 281-284.
- TNAH, L. H., LEE, S. L., TAN, A. L., LEE, C. T., NG, K. K. S. & NG, C. H. 2016. Intraspecific classification of *Ficus deltoidea* Jack subsp deltoidea (Moraceae) in Peninsular Malaysia based on morphological and molecular variations. *Biochemical Systematics and Ecology*, 67, 119-128.
- TSUKAYA, H. 2005. Leaf shape: genetic controls and environmental factors. *Int. J. Dev. Biol*, 547-555.
- TURNER, E. C. & FOSTER, W. A. 2009. The impact of forest conversion to oil palm on arthropod abundance and biomass in Sabah, Malaysia. *Journal of Tropical Ecology*, 25, 23-30.
- ULIAN, T., MATTANA, E., PRITCHARD, H. W. & SKWIERINSKI, R. 2013. Seasonality effects on plant phenology and seed ecology in *Oritrophium peruvianum* (Asteraceae), a threatened tropical alpine species. *South African Journal of Botany*, 88, 278-285.
- VAN NOORT, S. 2003. Fig wasps and the pollination of figs. *In:* J., B. & S., B. (eds.) *Figs of southern & south-central Africa.* Umdaus Press, Hatfield.
- VAN NOORT, S. & COMPTON, S. G. 1996. Convergent evolution of agaonine and sycoecine (Agaonidae, Chalcidoidea) head shape in response to the constraints of host fig morphology. *Journal of Biogeography*, 23, 415-424.
- VENEKLAAS, E. J., ZAGT, R. J., VANLEERDAM, A., VANEK, R., BROEKHOVEN, A. J. & VANGENDEREN, M. 1990. Hydrological properties of the epiphyte mass of a montane tropical rain-forest, Colombia. *Vegetatio*, 89, 183-192.
- VIJAY, V., PIMM, S. L., JENKINS, C. N. & SMITH, S. J. 2016. The impacts of oil palm on recent deforestation and biodiversity loss. *PLOS ONE*, 11, e0159668.
- VISSER, M. E., CARO, S. P., VAN OERS, K., SCHAPER, S. V. & HELM, B. 2010. Phenology, seasonal timing and circannual rhythms: towards a unified framework. *Philosophical Transactions of the Royal Society Biological Sciences*, 365, 3113-3127.
- WACHI, N., KUSUMI, J., TZENG, H. Y. & SU, Z. H. 2016. Genome-wide sequence data suggest the possibility of pollinator sharing by host shift in dioecious figs (Moraceae, *Ficus*). *Molecular Ecology*, 25, 5732-5746.
- WAGNER, K., MENDIETA-LEIVA, G. & ZOTZ, G. 2015. Host specificity in vascular epiphytes: a review of methodology, empirical evidence and potential mechanisms. *Aob Plants*, 7.

- WANG, G., CANNON, C. H. & CHEN, J. 2016. Pollinator sharing and gene flow among closely related sympatric dioecious fig taxa. *Proceedings of the Royal Society B-Biological Sciences*, 283.
- WANG, G., COMPTON, S. G. & CHEN, J. 2013. The mechanism of pollinator specificity between two sympatric fig varieties: a combination of olfactory signals and contact cues. *Annals of Botany*, 111, 173-181.
- WANG, R., COMPTON, S. G., QUINNELL, R. J., PENG, Y. Q., BARWELL, L. & CHEN, Y. 2015. Insect responses to host plant provision beyond natural boundaries: latitudinal and altitudinal variation in a Chinese fig wasp community. *Ecology and Evolution*, 5, 3642-3656.
- WANG, R., MATTHEWS, A., RATCLIFFE, J., BARWELL, L. J., PENG, Y. Q., CHOU, L. S., YU, H., YANG, H. W. & COMPTON, S. G. 2014. First record of an apparently rare fig wasp feeding strategy: obligate seed predation. *Ecological Entomology*, 39, 492-500.
- WANG, R. W., B.F, S., Q, Z., L, S. & L, Z. 2011. Asymmetric interaction and indeterminate fitness correlation between cooperative partners in the fig & fig wasp mutualism. *Journal of The Royal Society Interface*, 8, 1487-1496.
- WARE, A. B. & COMPTON, S. G. 1992. Breakdown of pollinator specificity in an African fig tree. *Biotropica*, 24, 544-549.
- WARNER, K. A., RUDALL, P. J. & FROHLICH, M. W. 2009. Environmental control of sepalness and petalness in perianth organs of waterlilies: a new Mosaic Theory for the evolutionary origin of a differentiated perianth. *Journal of Experimental Botany*, 60, 3559-3574.
- WASER, N. & OLLERTON, J. 2006. *Plant–Pollinator Interactions: From Specialization to Generalization,* Chicago, Illinois., University Press Chicago.
- WEBER, J., KARCZEWSKA, A., DROZD, J., LICZNAR, M., LICZNAR, S., JAMROZ,
 E. & KOCOWICZ, A. 2007. Agricultural and ecological aspects of a sandy soil as affected by the application of municipal solid waste composts.
- WEI, Z. D., KOBMOO, N., CRUAUD, A. & KJELLBERG, F. 2014. Genetic structure and hybridization in the species group of *Ficus auriculata:* can closely related sympatric *Ficus* species retain their genetic identity while sharing pollinators? *Molecular Ecology*, 23, 3538-3550.
- WEIBLEN, G. D. 2002. How to be a fig wasp. Annu. Rev. Entomol, 47, 299-330.
- WEIBLEN, G. D. & BUSH, G. L. 2002. Speciation in fig pollinators and parasites. *Molecular Ecology* 11, 1573-1578.
- WESTOBY, M., FALSTER, D. S., MOLES, A. T., VESK, P. A. & WRIGHT, I. J. 2002. Plant Ecological Strategies: Some Leading Dimensions of Variation Between Species. *Annual Review of Ecology and Systematics*, 33, 125-159.
- WESTOBY, M., JURADO, E. & LEISHMAN, M. 1992. Comparative evolutionary ecology of seed size. *Trends in Ecology & Evolution*, 7, 368-372.
- WESTOBY, M., LEISHMAN, M., LORD, J., H, P. & SCHOEN, D. J. 1996. Comparative ecology of seed size and dispersal. *Philosophical Transactions: Biological Sciences*, 351, 1309-1318.
- WIEBES, J. T. 1979. Co-evolution of figs and their insect pollinators. *Annual Review* of Ecology and Systematics, 10, 1-12.
- WIEBES, J. T. 1982. Fig wasps (Hymenoptera). *In:* GRESSITT, J. L. (ed.) *Biogeography and ecology of New Guinea.* The Hague: W. Junk Publishers.
- WIEBES, J. T. 1993. Agaonidae (Hymenoptera Chacidoidea) and *Ficus* (Moraceae): fig wasps and their figs, (*Blastophaga*). *Proc. Kon. Ned. Akad. V. Wetensch*, 96, 347-367.

WIEBES, J. T. 1994. *The Indo-Australian Agaoninae (pollinators of figs),* Netherlands, Leiderdorp.

- WILLIAMS-LINERA, G. & LAWTON, R. O. 1995. The ecology of hemi epiphytes in Forest canopies. *In:* NADKARNI, N. M. & LOWMAN, M. D. (eds.) *Forest Canopies.* New York: Academic Press.
- WILLIAMS, C. B. & SILLETT, S. C. 2007. Epiphyte communities on redwood (*Sequoia sempervirens*) in northwestern California. *Bryologist,* 110, 420-452.
- WITTMAN, P. K. 2000. The animal community associated with canopy bromeliads of the lowland Peruvian Amazon forest.
- WYSE, S. V. & BURNS, B. R. 2011. Do host bark traits influence trunk epiphyte communities? *New Zealand Journal of Ecology*, 35, 296-301.
- YANG, D. R., PENG, Y. Q., SONG, Q. S., ZHANG, G. M., WANG, R. W., ZHAO, T. Z.
 & WANG, Q. Y. 2002. Pollination biology of *Ficus hispida* in the tropical rainforests of Xishuangbanna, China. *Acta Botanica Sinica*, 44, 519-526.
- YANG, L. Y., MACHADO, C. A., DANG, X. D., PENG, Y. Q., YANG, D. R., ZHANG, D. Y. & LIAO, W. J. 2015. The incidence and pattern of copollinator diversification in dioecious and monoecious figs. *Evolution*, 69, 294-304.
- YANG, P., LI, Z. B., PENG, Y. Q. & YANG, D. R. 2012. Exchange of hosts: can agaonid fig wasps reproduce successfully in the figs of non-host *Ficus*? *Naturwissenschaften*, 99, 199-205.
- YU, H., ZHAO, N., CHEN, Y. & HERRE, E. 2008. Male and female reproductive success in the dioecious fig, *Ficus hirta* Vahl. in Guangdong Province, China: Implications for the relative stability of dioecy and monoecy. *Symbiosis*, 45, 121-127.
- ZACHARIADES, C., SCHATZ, B. & COMPTON, S. G. 2010. Wasp emergence from the figs of *Ficus sur:* characteristics and predation by ants. *Tropical Zoology*, 23, 121-138.
- ZHANG, F. P., PENG, Y. Q., COMPTON, S. G. & YANG, D. R. 2009. Floral characteristics of *Ficus curtipes* and the oviposition behavior of its pollinator fig wasp. *Annals of the Entomological Society of America*, 102, 556-559.
- ZHANG, G. M., SONG, Q. S. & YANG, D. R. 2006. Phenology of *Ficus racemosa* in Xishuangbanna, Southwest China. *Biotropica*, 38, 334-341.
- ZHANG, Q., ONSTEIN, R. E., LITTLE, S. A. & SAUQUET, H. 2018. Estimating divergence times and ancestral breeding systems in *Ficus* and Moraceae. *Annals of Botany*, 123, 191-204.
- ZHAO, T. T., COMPTON, S. G., YANG, Y. J., WANG, R. & CHEN, Y. 2014. Phenological Adaptations in *Ficus tikoua* Exhibit Convergence with Unrelated Extra-Tropical Fig Trees. *PLOS ONE*, 9, 1-17.
- ZIMISUHARA, B., VALDIANI, A., SHAHARUDDIN, N. A., QAMARUZZAMAN, F. & MAZIAH, M. 2015. Structure and principal components analyses reveal an intervarietal fusion in Malaysian mistletoe fig (*Ficus deltoidea* Jack) populations. *International Journal of Molecular Sciences*, 16, 14369-14394.
- ZOTZ, G. 1998. Demography of the epiphytic orchid, *Dimerandra emarginata. Journal* of Tropical Ecology, 14, 725-741.
- ZOTZ, G. 2013. The systematic distribution of vascular epiphytes a critical update. Botanical Journal of the Linnean Society, 171, 453-481.
- ZOTZ, G. 2016. Plants on Plants The Biology of vascular epiphytes. *Springer*, 190, 265-272.

- ZOTZ, G. & VOLLRATH, B. 2003. The epiphyte vegetation of the palm *Socratea exorrhiza* correlations with tree size, tree age and bryophyte cover. *Journal of Tropical Ecology*, 19, 81-90.
- ZOTZ, G. & WINTER, K. 1994. A One-year study on carbon, water and nutrient relationships in a tropical c-3-cam hemi-epiphyte, *Clusia-uvitana pittier*. *New Phytologist*, 127, 45-60.